UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

PURSUANT TO SECTION 13 OR 15(d) OF THE **SECURITIES EXCHANGE ACT OF 1934**

Date of report (Date of earliest event reported): February 24, 2023

ADAMIS PHARMACEUTICALS CORPORATION

(Exact Name of Registrant as Specified in Charter)

Delaware (State or other jurisdiction of incorporation)

0-26372 (Commission File Number)

82-0429727 (IRS Employer Identification No.)

11682 El Camino Real, Suite 300 San Diego, CA

(Address of Principal Executive Offices)

(Zip Code)

Registrant's telephone number, including area code: (858) 997-2400

(Former name or Former Address, if Changed Since Last Report.)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):

Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

 \times Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Exchange Act:

| | | Name of each exchange on | |
|-------------------------|-------------------|--------------------------|--|
| Title of each class | Trading Symbol(s) | which registered | |
| Common Stock | ADMP | NASDAQ Capital Market | |

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company \Box

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

Item 1.01. Entry into a Material Definitive Agreement.

Merger Agreement

On February 24, 2023, Adamis Pharmaceuticals Corporation, a Delaware corporation ("Adamis" or the "Company"), entered into an Agreement and Plan of Merger and Reorganization (the "Merger Agreement") with DMK Pharmaceuticals Corporation ("DMK"), a New Jersey corporation, and Aardvark Merger Sub, Inc., a Delaware corporation and a wholly-owned subsidiary of Adamis ("Merger Sub"). Under the terms of and subject to the satisfaction of the conditions described in the Merger Agreement, including approval of certain matters relating to the transaction by the Company's stockholders and DMK's stockholders, DMK will merge with and into Merger Sub (the "Merger"), with Merger Sub surviving as a wholly-owned subsidiary of Adamis. DMK Pharmaceuticals is a clinical stage neuro-biotechnology company focused on the development and commercialization of products for the treatment of a variety of neuro-based disorders with significant unmet medical needs. The current DMK product candidates have been selected and developed from a proprietary portfolio of small molecule neuropeptide analogues and include product candidates for the treatment of opioid use disorder, chronic pain, Parkinson's disease, and bladder problems. The Merger is intended to qualify for federal income tax purposes as a tax-free reorganization under the provisions of Section 368(a) of the U.S. Internal Revenue Code of 1986, as amended.

At the effective time of the Merger (the "Effective Time"), each share of common stock of DMK (other than dissenting shares, if any) will generally be converted into the right to receive a number of shares of Adamis common stock ("Common Stock") equal to the exchange ratio as defined in the Merger Agreement and described below; however, if a DMK shareholder's receipt of such shares would result in the shareholder's beneficial ownership of Adamis Common Stock exceeding a certain percentage limit specified in the Merger Agreement, then the shareholder will, in lieu of receiving shares of Common Stock in excess of such beneficial ownership limit, receive shares of a new series of Adamis convertible preferred stock (the "Series E Preferred") that is generally convertible into the number of shares of Adamis Common Stock that the stockholder would have been entitled to receive in excess of such beneficial ownership limit, but that is subject to certain beneficial ownership limitations on conversion and voting rights (the shares of Adamis Common Stock and Series E Preferred that are issuable pursuant to the Merger Agreement, sometimes referred to as the "Merger Consideration Shares"). The number of shares of Adamis Common Stock that will be issuable (including upon conversion of shares of Series E Preferred issuable in the transaction without regard to beneficial ownership conversion limitations) to holders of outstanding shares of DMK common stock (other than holders, if any, of dissenting shares) will be determined by dividing \$27,000,000 by the average closing prices of the Adamis Common Stock for the five trading days ending one trading day before the Effective Time (adjusted to give effect to the reverse stock split of the Common Stock); and the exchange ratio will be determined by dividing such number of shares by the number of outstanding DMK shares of common stock immediately before the Effective Time. Notwithstanding the foregoing, the Merger Agreement also provides that if the foregoing calculation of the total number of shares of Adamis Common Stock issuable to DMK stockholders (including shares issuable upon conversion of the Series E Preferred without regarding to beneficial ownership conversion limitations) would result in the holders of Adamis Common Stock immediately before the Effective Time owning less than a majority of the total number of post-merger outstanding shares of Adamis Common Stock (including shares issuable upon exercise of assumed DMK stock options) after the Effective Time (the "Adamis Percentage Threshold"), then the number of Adamis shares issuable to the DMK stockholder pursuant to the Merger (and with respect to the Series E Preferred determined on an as-converted basis) will be a number such that the holders of Adamis Common Stock immediately before the Effective Time hold a number of shares of Adamis Common Stock immediately after the Effective Time equal to the Adamis Percentage Threshold.

As contemplated by the Merger Agreement, Adamis intends to hold a special meeting of its stockholders and seek the approval of its stockholders to, among other things, vote on certain proposals the approval of which is necessary in order to effect the transaction, including a proposal to (a) issue the Merger Consideration Shares issuable in connection with the Merger, pursuant to the rules of The Nasdaq Stock Market LLC ("Nasdaq") and (b) amend the Company's restated certificate of incorporation to effect a reverse stock split of the Adamis Common Stock, in a ratio to be set by the Adamis board of directors and, assuming the issuance of Merger Consideration Shares is approved and other closing conditions described in the Merger Agreement are satisfied, determined prior to the closing of the Merger. The reverse stock split is intended to provide sufficient shares in order to complete the transactions contemplated by the Merger Agreement as well as to increase the trading prices of the Company's common stock in order to satisfy the minimum bid price requirements for continued listing of the Common Stock on the Nasdaq Capital Market.

The Merger Agreement contains a number of customary representations, warranties, and covenants of both parties, including, among others, covenants relating to (1) taking all action necessary to allow the respective companies' stockholders to vote on proposals relating to the Merger, (2) non-solicitation of alternative acquisition proposals, (3) the conduct of their respective businesses during the period between the date of signing the Merger Agreement and the closing of the Merger, and (4) Adamis filing with the U.S. Securities and Exchange Commission (the "SEC") after the closing of the Merger a registration statement or prospectus supplement covering the resale of shares of Adamis Common Stock that will be issued or issuable in connection with the Merger (the "Registration Statement"). The Merger Agreement provides that Adamis will pay certain actual, out-of-pocket transaction expenses incurred by DMK in connection with the transaction.

The Merger Agreement may be terminated by either party under certain circumstances, including, among others: (i) if the Merger has not been completed by June 30, 2023; (ii) if a court or other governmental entity has issued a final and non-appealable order prohibiting the closing; (iii) if the Company's or DMK's stockholders fail to approve the proposals required to complete the transaction; (iv) upon a material uncured breach by the other party that would result in a failure of the conditions to the closing; or (v) upon the occurrence of certain other triggering events as defined in the Merger Agreement.

Following the closing of the Merger, the Company's executive officers are expected to include officers from the Company and DMK. Ebrahim (Eboo) Versi, M.D., Ph.D., the co-founder and chief executive officer of DMK, is expected to become chief executive officer of the Company; David J. Marguglio, the current chief executive officer and President of Adamis, is expected to continue as President of the Company; and David C. Benedicto is expected to continue as the chief financial officer of the Company.

The Merger Agreement provides that the board of directors of the Company after completion of the Merger will consist of two directors designated by DMK, and three directors designated by the Company which are expected to be three of the current Adamis independent directors. Immediately after the closing of the Merger, it is expected that the Board will be comprised of Howard C. Birndorf, Meera J. Desai, Ph.D. and Vickie Reed, who are currently serving as independent directors of the Company, as well as Dr. Versi and Jannine Versi, as the designees of DMK, with Dr. Versi serving as Chair of the Board.

The transaction was approved by the boards of directors of both companies. Subject to a number of potential uncertainties, including without limitation the preparation and filing of a preliminary and definitive proxy statement with the SEC, the approval of DMK's and Adamis' respective stockholders of proposals relating to the Merger transaction, and the satisfaction of other closing conditions, Adamis anticipates that the transaction will close during the second quarter of 2023.

Support Agreement

Concurrently with the execution of the Merger Agreement, the principal stockholder of DMK (solely in its capacity as a DMK stockholder), holding in excess of approximately 95% of the currently outstanding shares of DMK capital stock, has entered into a support agreement with Adamis, DMK, and Merger Sub to vote all of the shares of DMK capital stock held by the stockholder in favor of adoption of the Merger Agreement and against any alternative acquisition proposals (the "Support Agreement").

Other

The Merger Agreement and form of Support Agreement are attached to this report as Exhibits 2.1 and 10.1, respectively. The foregoing descriptions of these documents do not purport to be complete and are qualified in their entirety by reference to the full text of the Merger Agreement and the Support Agreement, which are incorporated herein by reference. The Merger Agreement has been attached as an exhibit to this Report to provide investors and securityholders with information regarding its terms. It is not intended to provide any other factual information about Adamis or DMK or to modify or supplement any factual disclosures about the Company in its public reports filed with the SEC. The assertions embodied in the representations and warranties contained in the Merger Agreement were made solely for the purpose of the Merger Agreement and solely for the benefit of the parties thereto in connection with the negotiated terms of the Merger Agreement, and are qualified by information in confidential disclosure schedules delivered by the parties in connection with the signing of the Merger Agreement. Moreover, certain representations and warranties contained in the Merger Agreement from what might be viewed as material to the Company's stockholders. Accordingly, the representations and warranties in the Merger Agreement should not be relied on by any persons as characterizations of the actual disclosure about the Company or DMK at the time they were made and should be considered in conjunction with the entirety of the factual disclosure about the Company in the Company or DMK at the time they were made and should be considered in conjunction with the entirety of the factual disclosure about the Company's public reports filed with the SEC. Information concerning the subject matter of the representations and warranties may change after the date of the Merger Agreement, which subsequent information may or may not be fully reflected in the Company's public disclosures. The Merger Agreement ishould not be read alone, but

Item 3.02 Unregistered Sales of Equity Securities

Pursuant to the transactions contemplated by the Merger Agreement, following the Effective Time of the Merger the Company will issue shares of Common Stock and Series E Preferred. The number and class of shares to be issued, the nature of the transaction and the nature and amount of consideration received by the Company, are described in Item 1.01 of this Form 8-K, which is incorporated by reference into this Item 3.02. The shares to be issued by the Company in the Merger will be issued in a private placement exempt from registration under Section 4(a)(2) of the Act and Rule 506(b) of Regulation D, as the offer and sale of such securities does not involve a "public offering" as provided in Section 4(a)(2) of the Securities Act of 1933, as amended, and other applicable requirements are or will be satisfied.

Item 5.02. Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers.

The disclosures set forth in Item 1.01 are incorporated by reference into this Item 5.02.

Change in Company Officers and Directors

The Merger Agreement contemplates that effective upon the closing of the Merger, David J. Marguglio and Richard C. Williams will resign from the Company's (and subsidiaries') board of directors (the "Board") and committees of the Board on which they respectively served, which resignation will not have been the result of any disagreements with Adamis relating to the Adamis' operations, policies or practices. The Merger Agreement also contemplates that effective upon the closing of the Merger, two directors designated by DMK will be appointed to the Board, that those directors will Dr. Versi and Jannine Versi, who currently are directors of DMK, and that Dr. Versi will become Chair of the Board. The Merger Agreement provides that prior to and effective upon the closing of the Merger, David J. Marguglio will resign as chief executive officer of Adamis and will continue as the President of Adamis, and Dr. Versi will be appointed as the chief executive officer of Adamis.

Ebrahim (Eboo) Versi, M.D., Ph.D.

Dr. Ebrahim "Eboo" Versi, age 70, has over 30 combined years of experience as a physician, practicing surgeon, academic, teacher, entrepreneur, investor, and senior executive. Dr. Versi has spent more than 20 years in the pharmaceutical and medical device industry, holding positions with both large and small pharmaceutical companies. During his career in the industry, Dr. Versi has been the inventor of several patents and the recipient of several NIH grants and has helped develop drugs and devices for a variety of indications. Dr. Versi founded DMK Pharmaceuticals in June 2016, serving as a member of the board of directors and became its chief executive officer in 2017. Dr. Versi founded Versi Group LLC in July 2005 and has served as its president since its inception. Versi Group LLC owns a portfolio of patents related to drug development and methods of delivery for various disease states including neuroscience, urology, and cardiology. From January 2021 to November 2022, Dr. Versi served as chairman of Dina Pharma Inc, a clinical-stage biopharmaceutical company. From April 2016 to December 2022, Dr. Versi served as chairman of Opus Therapeutics Inc., a company focused on developing drugs for urological disorders. Dr. Versi served as Consultant Chief Medical Officer from 2013 to 2014 for Taris Biomedical, Inc., a company developing a device for delivery of lidocaine to the bladder. From 2007 to 2009 Dr. Versi served as the Chief Medical Officer of Mt. Cook Pharma Inc., a company developing drugs for urology, analgesia, Parkinson's Disease, and depression. From 2009 to 2011 Dr. Versi served as Vice President, Head of Drug Safety and Medical Affairs at Auxilium Pharmaceuticals Inc., a public biopharmaceutical company. From 2006 to 2007, Dr. Versi served as Senior Vice President of development of Plethora Solutions Holdings PLC (AIM:PLC), a urology company with development drug products and devices. From 2004 to 2005, Dr. Versi served as senior vice president of Odyssey Pharmaceuticals, a subsidiary of Pliva d.d., now a Teva company, where he was developing and marketing prescription drugs for a variety of medical conditions. From 2003 to 2004, Dr. Versi served as vice president of medical affairs of Yamanouchi Pharma America (now Astellas Pharma US, Inc.), a major Japanese pharmaceutical company with a focus on urological products. From 1998 to 2003, Dr. Versi served as Director of Urology and Women's Health for Pharmacia (which merged with Pfizer), where he worked on raising awareness of bladder problems by changing the focus from urinary incontinence to overactive bladder (OAB). From 1997 to 1999, Dr. Versi served as consulting medical director for Eli Lilly and Company.

Dr. Versi received a BA, MA and DPhil (Ph.D.) in 1978 from Oxford University in the United Kingdom before obtaining his MB BChir (MD) degree from Cambridge University in 1980. Following medical school, Dr. Versi completed a residency and fellowship at Kings College Hospital and served as a Senior Registrar at the Royal London Hospital. He was then a Senior Lecturer and Consultant (Attending) at St. Thomas' & Guys Hospitals before moving to the U.S. to accept an academic position as Associate Professor at Harvard Medical School which he held from 1993 to 1999. At Harvard, he set up the first urogynecology program at the Brigham & Women's Hospital and served as its Chief of Urogynecology. Since leaving Harvard, Dr. Versi has planned and executed clinical programs for multiple products that are currently marketed. Dr. Versi is an internationally recognized opinion leader with more than 100 scientific publications.

Jannine Versi

Jannine Versi, age 37, has served as a director of DMK since October 2017. In 2019, Ms. Versi co-founded Elektra Health, a privately-held healthcare technology company focused on women's health, and menopause in particular. She currently serves as the chief operating officer of Elektra Health. Prior to Elektra Health, Ms. Versi was on the founding team and served as vice president of clinical programs and partnerships at Cityblock Heath, a healthcare technology company, from 2017 to 2019. From 2014 to 2016, Ms. Versi served as Chief of Staff to the Under Secretary of Commerce for International Trade in the U.S. Department of Commerce as part of the Obama administration. Ms. Versi worked at Google from 2009 to 2012, reporting to Google's global marketing leadership in California and in the United Kingdom. Ms. Versi graduated *magna cum laude* from the University of Pennsylvania in 2007 and received an M.B.A. from Harvard Business School in 2014. From 2007 to 2008, Ms. Versi was a Fulbright Scholar conducting economic research in India. In 2021, Ms. Versi was a recipient of Harvard Business School's Blavatnik Fellowship in Life Science Entrepreneurship.

Ms. Versi is the daughter of Dr. Versi. With that exception, there are no family relationships among any of the new persons expected to become directors or officers upon the closing of the Merger. Dr. Versi is a stockholder and option holder of DMK and will receive Merger Consideration Shares pursuant to the Merger, and DMK stock options held by Dr. Versi will be assumed by the Company pursuant to the Merger and the transactions contemplated by the Merger Agreement. Ms. Versi is the trustee of a member of Versi Group, LLC, a shareholder in DMK, and personally holds options to purchase shares of DMK common stock, which options will be assumed by the Company pursuant to the Merger and the transactions contemplated by the Merger Agreement.

Item 7.01 Regulation FD Disclosure.

On February 27, 2023, Adamis issued a press release announcing the signing of the Merger Agreement. A copy of the press release announcing the signing of the Merger Agreement is attached hereto as Exhibit 99.1 and is incorporated herein by reference. The press release and the information set forth therein shall not be deemed to be filed for purposes of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or otherwise be subject to the liabilities of that section, nor shall it be deemed to be incorporated by reference in any filing under the Securities Act of 1933, as amended, or the Exchange Act.

Forward-Looking Statements

This Current Report on Form 8-K and the press release attached hereto as Exhibit 99.1 contain forward-looking statements based upon Adamis' and DMK's current expectations. This communication contains "forward-looking statements" within the meaning of the Private Securities Litigation Reform Act of 1995. Forward-looking statements are identified by terminology such as "may," "should," "expects," "plans," "anticipates," "could," "intends," "target," "projects," "contemplates," "believes," "estimates," "predicts," "potential" or "continue" or the negative of these terms or other similar words. These statements are only predictions. Adamis and DMK have based these forward-looking statements largely on their then-current expectations and projections about future events, as well as the beliefs and assumptions of management. Forward-looking statements are subject to a number of risks and uncertainties, many of which involve factors or circumstances that are beyond each of Adamis' and DMK's control, and actual results could differ materially from those stated or implied in forward-looking statements due to a number of factors, including but not limited to: (i) risks associated with Adamis' and DMK's ability to obtain the stockholder approval required to consummate the proposed Merger and the timing of the closing of the proposed Merger, including the risks that a condition to closing would not be satisfied within the expected timeframe or at all or that the closing of the proposed Merger will not occur; (ii) the outcome of any current legal proceedings or future legal proceedings that may be instituted against either of the parties or others, including proceedings related to the Merger Agreement; (iii) the occurrence of any event, change or other circumstance or condition that could give rise to the termination of the Merger Agreement; (iv) unanticipated difficulties or expenditures relating to the proposed Merger, the response of business partners and competitors to the announcement of the proposed Merger, and/or potential difficulties in employee retention as a result of the announcement and pendency of the proposed Merger; (v) whether the combined business of DMK and Adamis will be successful; and (vi) those risks detailed in Adamis' most recent Annual Report on Form 10-K and subsequent reports filed with the SEC, as well as other documents that may be filed by Adamis from time to time with the SEC. Accordingly, you should not rely upon forward-looking statements as predictions of future events. Neither Adamis nor DMK can assure you that the events and circumstances reflected in the forward-looking statements will be achieved or occur, and actual results could differ materially from those projected in the forward-looking statements. The forward-looking statements made in this communication relate only to events as of the date on which the statements are made. Except as required by applicable law or regulation, Adamis and DMK undertake no obligation to update any forwardlooking statement to reflect events or circumstances after the date on which the statement is made or to reflect the occurrence of unanticipated events. Investors should not assume that any lack of update to a previously issued "forward-looking statement" constitutes a reaffirmation of that statement.

Additional Information about the Merger and Where to Find It

Adamis intends to file a proxy statement in connection with the proposed transaction. Investors and stockholders are urged to read this filing when it becomes available because it will contain important information about the transaction. This Current Report on Form 8-K does not constitute an offer of any securities for sale or the solicitation of any proxy. BEFORE MAKING ANY VOTING OR INVESTMENT DECISION, ADAMIS' STOCKHOLDERS ARE URGED TO READ THE PROXY STATEMENT REGARDING THE PROPOSED TRANSACTION CAREFULLY AND IN ITS ENTIRETY WHEN IT BECOMES AVAILABLE BECAUSE IT WILL CONTAIN IMPORTANT INFORMATION ABOUT THE PROPOSED TRANSACTION AND RELATED MATTERS. Investors and stockholders may obtain free copies of the proxy statement and other relevant documents (when they become available) and other documents filed with the Securities and Exchange Commission at the Securities and Exchange Commission's web site at: www.sec.gov. In addition, investors and stockholders may obtain free copies of the documents filed with the Securities and Exchange Commission by Adamis by contacting David C. Benedicto, Adamis' chief financial officer, at (858) 997-2400.

Participants in the Solicitation

Adamis and DMK, and their respective directors and executive officers, may be deemed to be participants in the solicitation of proxies from the companies' stockholders in connection with the proposed transaction. Information regarding the interests of directors and executive officers in the transaction will be included in the proxy statement to be filed by Adamis. Investors and security holders are urged to read the Company's proxy statement and the other relevant materials when they become available before making any voting or investment decision with respect to the proposed transaction. Additional information regarding directors and executive officers of Adamis is also included in the Company's annual report on Form 10-K for the year ended December 31, 2021, and, when it becomes available, its annual report on Form 10-K for the year ended December 31, 2022, filed with the Securities and Exchange Commission, which is available as described above.

No Offer or Solicitation

This Current Report on Form 8-K is not intended to and shall not constitute an offer to sell or the solicitation of an offer to buy any securities or the solicitation of any vote in any jurisdiction pursuant to the proposed transaction or otherwise, nor shall there be any sale of securities in any jurisdiction in contravention of applicable law. No offering of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the Securities Act of 1933, as amended.

Item 9.01. Financial Statements and Exhibits.

(c) Exhibits.

| Exhibit No. | Description |
|-------------|---|
| <u>2.1</u> | Agreement and Plan of Merger and Reorganization, dated as of February 24, 2023, by and among Adamis Pharmaceuticals, Inc., Adamis Merger Sub, Inc., and DMK Pharmaceuticals Corporation. * |
| <u>10.1</u> | Form of Support Agreement, dated February 24, 2023, by and among Adamis Pharmaceuticals, Inc., Aardvark Merger Sub, Inc., DMK Pharmaceuticals Corporation, and certain stockholders of DMK Pharmaceuticals Corporation. |
| <u>99.1</u> | Press release dated February 27, 2023. |

* Exhibits and/or schedules have been omitted pursuant to Item 601(a)(5) of Regulation S-K. The registrant hereby undertakes to furnish supplementally copies of any of the omitted exhibits and schedules upon request by the SEC; provided, however, that the registrant may request confidential treatment pursuant to Rule 24b-2 of the Exchange Act, as amended, for any exhibits or schedules so furnished.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the Registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

ADAMIS PHARMACEUTICALS CORPORATION

Dated: February 27, 2023

By:/s/ David J. MarguglioName:David J. MarguglioTitle:Chief Executive Officer

Exhibit 2.1

AGREEMENT AND PLAN OF MERGER AND REORGANIZATION

between

ADAMIS PHARMACEUTICALS CORPORATION,

a Delaware corporation,

AARDVARK MERGER SUB, INC.,

a Delaware corporation

and

DMK PHARMACEUTICALS CORPORATION,

a New Jersey corporation

Dated as of February 24, 2023

| | | Page |
|-------------|---|------|
| ARTICLE I | THE MERGER | 2 |
| 1.1 | Merger of DMK into Merger Sub | 2 |
| 1.2 | Effect of the Merger | 2 |
| 1.3 | Closing; Effective Time | 2 |
| 1.4 | Certificate of Incorporation and Bylaws; Directors and Officers | 3 |
| 1.5 | Reverse Split of Adamis Common Stock | 3 |
| 1.6 | Shares to be Issued; Effect on Capital Stock | 4 |
| 1.7 | Calculation of the Exchange Ratio | 6 |
| 1.8 | Dissenting Shares | 7 |
| 1.9 | No Further Transfer of DMK Capital Stock | 8 |
| 1.10 | Exchange of Certificates | 8 |
| 1.11 | Further Action | 10 |
| 1.12 | Tax Consequences; Withholding | 10 |
| 1.13 | Plan of Merger | 10 |
| ARTICLE II | REPRESENTATIONS AND WARRANTIES OF DMK | 10 |
| 2.1 | Organization and Qualification; Charter Documents | 11 |
| 2.2 | Subsidiaries | 11 |
| 2.3 | Authority; Vote Required | 11 |
| 2.4 | No Conflict | 12 |
| 2.5 | Non-Contravention; Consents | 12 |
| 2.6 | Governmental Authorizations | 13 |
| 2.7 | Capitalization | 13 |
| 2.8 | Title to Assets | 15 |
| 2.9 | DMK Financial Statements | 16 |
| 2.10 | Absence of Certain Changes | 16 |
| 2.11 | Interested Party Transactions | 17 |
| 2.12 | Intellectual Property | 17 |
| 2.13 | Taxes | 18 |
| 2.14 | Employee Benefit Plans | 20 |
| 2.15 | Employee Matters | 21 |
| 2.16 | Insurance | 21 |
| 2.17 | Compliance with Law | 21 |
| 2.18 | Environmental Matters | 22 |
| 2.19 | Legal Proceedings | 22 |
| 2.20 | Contracts; No Defaults | 22 |
| 2.21 | Labor Matters | 23 |
| 2.22 | Regulatory Compliance | 23 |
| 2.23 | Unlawful Payments | 24 |
| 2.24 | Financial Advisor | 24 |
| 2.25 | Takeover Statutes; No Rights Plan; Appraisal Rights | 24 |
| 2.26 | Takeover Statutes; No Rights Plan; Appraisal Rights | 24 |
| ARTICLE III | REPRESENTATIONS AND WARRANTIES OF ADAMIS AND MERGER SUB | 25 |
| 3.1 | Organization and Qualification; Charter Documents | 25 |
| | | |

TABLE OF CONTENTS (continued)

| 3.2 | Subsidiaries | <u>Page</u> 25 |
|------------|---|-------------------|
| 3.2 3.3 | Authority | 25 26 |
| 3.4 | No Conflict | 20 |
| 3.5 | Non-Contravention; Consents | 26 |
| 3.6 | Governmental Authorizations | 20 |
| 3.7 | Capitalization | 28 |
| 3.8 | Title to Assets | 28 |
| 3.9 | SEC Reports; Financial Statements; Listing and Maintenance Requirements | 28 |
| 3.10 | Absence of Certain Changes | 20 |
| 3.11 | Interested Party Transactions | 29 |
| 3.12 | Intellectual Property | 29 |
| 3.13 | Taxes | 30 |
| 3.14 | Employee Benefit Plans | 32 |
| 3.15 | Employee Matters | 33 |
| 3.16 | Insurance | 33 |
| 3.17 | Compliance with Law | 33 |
| 3.18 | Environmental Matters | 34 |
| 3.19 | Legal Proceedings | 34 |
| 3.20 | Contracts; No Defaults | 34 |
| 3.21 | Labor Matters | 35 |
| 3.22 | Regulatory Compliance | 35 |
| 3.23 | Unlawful Payments | 36 |
| 3.24 | Representations Complete | 36 |
| 3.25 | Financial Advisor | 36 |
| 3.26 | Disclaimer of Reliance | 36 |
| ARTICLE IV | CONDUCT BEFORE THE EFFECTIVE TIME | 37 |
| 4.1 | Access and Investigation | 37 |
| 4.2 | Operation of Adamis' Business | 37 |
| 4.3 | Operation of DMK's Business | 38 |
| 4.4 | Notification of Certain Matters | 38 |
| 4.5 | No Solicitation | 39 |
| ARTICLE V | ADDITIONAL AGREEMENTS | 41 |
| 5.1 | Proxy Statement | 41 |
| 5.2 | DMK Stockholder Approval | 42 |
| 5.3 | Adamis Stockholders Meeting; Change in the Adamis Board Recommendation; Adoption of Agreement by Adamis as Sole | |
| | Stockholder of Merger Sub | 42 |
| 5.4 | Registration Statement | 43 |
| 5.5 | Information Statement; DMK Written Consent | 44 |
| 5.6 | Adamis Stockholders Meeting | 45 |
| 5.7 | Regulatory Approvals | 46 |
| 5.8 | Indemnification of Officers and Directors | 46 |
| 5.9 | Additional Agreements | 47 |
| 5.10 | Disclosure; Public Announcements | 48 |
| 5.11 | Directors; Officers | 48 |
| 5.12 | Tax Matters | 48 |
| 5.13 | Adamis Amendments | 49 |
| | | |

TABLE OF CONTENTS (continued)

| | | Page |
|-------------|--|------|
| 5.14 | DMK's Auditors | 49 |
| 5.15 | Legends | 49 |
| 5.16 | Confidentiality | 50 |
| 5.17 | Section 16 Matters | 50 |
| 5.18 | Listing | 50 |
| 5.19 | Private Placement | 50 |
| 5.20 | Termination of Certain DMK Agreements | 51 |
| 5.21 | DMK Options | 51 |
| 5.22 | Allocation Certificate | 52 |
| 5.23 | Employee Benefit Matters | 52 |
| 5.24 | Takeover Statutes | 52 |
| 5.25 | Merger Sub | 53 |
| 5.26 | Supplement to Disclosure Schedules | 53 |
| 5.27 | Series C Preferred Certificate of Designation | 53 |
| ARTICLE VI | CONDITIONS PRECEDENT TO OBLIGATIONS OF EACH PARTY | 53 |
| 6.1 | Stockholder Approval | 53 |
| 6.2 | No Restraints | 54 |
| 6.3 | Governmental Authorization | 54 |
| 6.4 | No Governmental Proceedings Relating to Contemplated Transactions or Right to Operate Business | 54 |
| 6.5 | Series E Preferred Certificate of Designation | 54 |
| ARTICLE VII | ADDITIONAL CONDITIONS PRECEDENT TO OBLIGATIONS OF ADAMIS AND MERGER SUB | 54 |
| 7.1 | Accuracy of Representations | 54 |
| 7.2 | Performance of Covenants | 54 |
| 7.3 | No Material Adverse Effect | 55 |
| 7.4 | Agreements and Other Documents | 55 |
| 7.5 | Conversion of DMK Convertible Notes | 55 |
| 7.6 | FIRPTA Certificate | 55 |
| 7.7 | Third Party Consents | 55 |
| 7.8 | Securities Law Matters | 55 |
| 7.9 | Dissenting Shares | 56 |
| 7.10 | DMK License Agreements | 56 |
| | ADDITIONAL CONDITIONS PRECEDENT TO OBLIGATION OF DMK | 56 |
| 8.1 | Accuracy of Representations | 56 |
| 8.2 | Performance of Covenants | 56 |
| 8.3 | No Material Adverse Effect | 56 |
| 8.4 | Documents | 56 |
| 8.5 | Sarbanes-Oxley Certifications | 57 |
| 8.6 | Board of Directors | 57 |
| 8.7 | Officers | 57 |
| 8.8 | Adamis Reverse Stock Split Certificate of Amendment | 57 |
| 8.9 | SEC Reports | 57 |
| 8.10 | Nasdaq Listing | 57 |
| | TERMINATION | 57 |
| 9.1 | Termination | 57 |
| | | |

TABLE OF CONTENTS (continued)

| | | Page |
|-----------|--|------|
| 9.2 | Effect of Termination | 59 |
| 9.3 | Expenses | 59 |
| ARTICLE X | MISCELLANEOUS PROVISIONS | 59 |
| 10.1 | Non-Survival of Representations and Warranties | 59 |
| 10.2 | Amendment | 60 |
| 10.3 | Waiver | 60 |
| 10.4 | Entire Agreement; Counterparts; Exchanges by Facsimile | 60 |
| 10.5 | Applicable Law; Jurisdiction | 60 |
| 10.6 | Waiver of Jury Trial | 61 |
| 10.7 | Notices | 61 |
| 10.8 | Cooperation | 62 |
| 10.9 | Severability | 62 |
| 10.10 | Other Remedies; Specific Performance | 62 |
| 10.11 | Conflict Waiver; Attorney-Client Privilege | 62 |
| 10.12 | Construction | 65 |
| | | |

AGREEMENT AND PLAN OF MERGER AND REORGANIZATION

THIS AGREEMENT AND PLAN OF MERGER AND REORGANIZATION (the "*Agreement*") is dated as of February 24, 2023 (the "*Agreement Date*"), and is entered into by and among Adamis Pharmaceuticals Corporation, a Delaware corporation ("*Adamis*"), Aardvark Merger Sub, Inc., a Delaware corporation and a wholly-owned subsidiary of Adamis ("*Merger Sub*"), and DMK Pharmaceuticals Corporation, a New Jersey corporation ("*DMK*"). Certain capitalized terms used in this Agreement are defined in <u>Exhibit A</u> attached hereto.

BACKGROUND

A. The Board of Directors of Adamis (the "*Adamis Board*") and the Board of Directors of DMK (the "*DMK Board*") have each determined that it is in the best interests of their respective stockholders for DMK and Adamis to enter into a business combination transaction pursuant to which DMK will merge with and into Merger Sub (the "*Merger*"), with Merger Sub continuing after the Merger as the surviving corporation and a wholly owned subsidiary of Adamis.

B. Pursuant to the Merger, each outstanding share of common stock, \$0.001 par value per share, of DMK ("*DMK Common Stock*") will, in accordance with the provisions of this Agreement, be converted into the Merger Consideration, consisting of shares of Adamis common stock, \$0.0001 par value per share ("*Adamis Common Stock*") and, in some instances, shares of Adamis Series E (or other series designated by Adamis) Convertible Preferred Stock, \$0.0001 par value per share ("*Series E Preferred*"), established pursuant to the Certificate of Designation of Preferences, Rights and Limitations of Series E Convertible Preferred Stock (the "*Series E Preferred Certificate of Designation*"), substantially in the form of <u>Exhibit C</u> attached hereto, with such changes thereto as Adamis and DMK may mutually agree, all as described in this Agreement.

C. In connection with, and before the consummation of, the Merger, a reverse stock split of Adamis Common Stock shall be consummated, pursuant to which a certain number of outstanding shares of Adamis Common Stock (determined by the Reverse Stock Split Ratio) will be converted into one (1) share of Adamis Common Stock, as described further in Section 1.5 below.

D. The Board of Directors of Adamis (i) has approved and declared advisable this Agreement (including the plan of merger, as such term is used in Section 14A:10-1 of the New Jersey Business Corporation Act (the "*NJBCA*")) (the "*Plan of Merger*"), the Merger and the other transactions contemplated by this Agreement, (ii) has determined that the Merger is in the best interests of Adamis and its stockholders and has determined to recommend to the stockholders of Adamis the issuance of Adamis, in its capacity as the sole stockholder of Merger Sub, vote to adopt this Agreement and approve the Merger and such other actions as are contemplated by this Agreement; and the board of directors of Merger Sub has adopted, approved and declared advisable this Agreement (including the Plan of Merger), the Merger, and the other transactions contemplated by this Agreement, on the terms and subject to the conditions set forth in this Agreement.

E. The Board of Directors of DMK (i) has approved and declared advisable this Agreement (including the Plan of Merger), the Merger and the other transactions contemplated by this Agreement, and (ii) has determined that the Merger is in the best interests of DMK and its stockholders and has determined to recommend the approval of this Agreement (and the related matters contemplated by this Agreement) and the Merger to the stockholders of DMK.

F. The Parties intend, by executing this Agreement, to adopt a plan of reorganization within the meaning of Section 368 of the Internal Revenue Code of 1986, as amended.

G. As an inducement to Adamis to enter into this Agreement, concurrently herewith certain securityholders of DMK have entered into an agreement with Adamis, in substantially the form attached hereto as <u>Exhibit B</u>, or such other form as is reasonably satisfactory to Adamis (a "*DMK Support Agreement*"), pursuant to which each such person has agreed, among other things, to vote all shares of capital stock of DMK owned by such person to approve the Merger, this Agreement and the transactions contemplated hereby.

AGREEMENT

The Parties to this Agreement, intending to be legally bound, agree as follows:

ARTICLE I THE MERGER

1.1. <u>Merger of DMK into Merger Sub</u>. Upon the terms and subject to the conditions set forth in this Agreement and in accordance with the NJBCA, at the Effective Time DMK shall be merged with and into Merger Sub, and the separate existence of DMK shall cease. Merger Sub will continue as the surviving corporation following the Merger (the "*Surviving Corporation*") under the DGCL.

1.2. <u>Effect of the Merger</u>. The Merger shall have the effects set forth in this Agreement and the applicable provisions of the NJBCA. As a result of the Merger, Merger Sub will continue as a wholly-owned subsidiary of Adamis. Without limiting the generality of the foregoing, and subject thereto, at the Effective Time, all the property, rights, privileges, powers and franchises of DMK and Merger Sub shall vest in the Surviving Corporation, and all debts, liabilities and duties of DMK and Merger Sub shall become the debts, liabilities and duties of the Surviving Corporation.

1.3. Closing; Effective Time. Unless this Agreement is earlier terminated pursuant to the provisions of Section 9.1, and subject to the satisfaction or waiver of the conditions set forth in Articles VI, VII and VIII, the consummation of the Merger (the "*Closing*") shall take place remotely, on a date to be agreed by Adamis and DMK (the "*Closing Date*"), which shall be no later than the fifth Business Day after the satisfaction or waiver of the last to be satisfied or waived of the conditions set forth in Articles VI, VII and VIII (other than those conditions that by their nature are to be satisfied at the Closing, but subject to the satisfaction or waiver of such conditions), or at such other time, date and place as Adamis and DMK may mutually agree in writing. At the Closing, subject to the terms and conditions of this Agreement, the Parties hereto shall (a) cause the Merger to be consummated by executing and filing with the Secretary of State of Delaware and the Secretary of State of New Jersey a certificate of merger, executed and acknowledged in accordance with and containing such information as is required by the Delaware General Corporation Law (the "*DGCL*") and the NJBCA to effect the Merger (the "*Certificate of Merger*") and (b) on or after the Closing Date duly make all other filings and recordings required by the NJBCA and the DGCL in order to effectuate the Merger. The Merger shall become effective at such time as the Certificate of Merger has been duly filed with the Secretary of State of New Jersey or at such later time as is agreed to by the Parties hereto in writing and specified in the Certificate of Merger in accordance with the relevant provisions of the DGCL and the NJBCA (the time as of which the Merger becomes effective being referred to as the "*Effective Time*").

1.4. <u>Certificate of Incorporation and Bylaws; Directors and Officers</u>. At the Effective Time:

(a) <u>Surviving Corporation Certificate of Incorporation</u>. By virtue of the Merger, the certificate of incorporation of Merger Sub, as in effect immediately before the Effective Time, shall be the certificate of incorporation of the Surviving Corporation until thereafter amended as provided by the DGCL and such certificate of incorporation;

(b) <u>Adamis Certificate of Incorporation</u>. Adamis shall take all actions necessary so that the restated certificate of incorporation of Adamis, as amended (the "*Adamis Restated Certificate*"), shall remain in effect following the Effective Time, until such time as it may be thereafter amended as permitted by the DGCL; provided, however, that before the Effective Time, Adamis shall file an amendment to its Restated Certificate to effect the Reverse Stock Split;

(c) <u>Surviving Corporation Bylaws</u>. The Bylaws of the Surviving Corporation shall be identical to the bylaws of Merger Sub as in effect immediately prior to the Effective Time, until thereafter amended as provided by the DGCL and such bylaws;

(d) <u>Surviving Corporation Directors and Officers</u>. The Parties shall take the actions necessary so that the directors and officers of the Surviving Corporation, each to hold office in accordance with the certificate of incorporation and bylaws of the Surviving Corporation, shall be the directors and officers of Adamis as described in Section 5.11, or such other directors and officers as Adamis and DMK may mutually agree; and

(e) <u>Directors and Officers of Adamis</u>. The Parties shall take the actions necessary so that the directors and officers of Adamis, each to hold office in accordance with the Restated Certificate and Adamis Bylaws, shall be as set forth in Section 5.11.

1.5. <u>Reverse Split of Adamis Common Stock</u>.

(a) <u>Adamis Reverse Stock Split Certificate of Amendment</u>. Before the Effective Time, and subject to receipt of the requisite stockholder approval at the Adamis Stockholders Meeting, Adamis shall cause to be filed a certificate of amendment to its Restated Certificate in form and substance satisfactory to Adamis (the "*Adamis Reverse Stock Split Amendment*"), whereby without any further action on the part of Adamis, DMK or any stockholder of Adamis:

(i) a number of shares of Adamis Common Stock, based on the Reverse Stock Split Ratio, issued and outstanding immediately before the filing of the Adamis Reverse Stock Split Amendment shall automatically, without any action by the holder thereof, be reclassified and combined into one validly issued, fully paid and non-assessable share of Adamis Common Stock (sometimes referred to as "*New Adamis Common Stock*"), subject to the treatment of fractional share interests as described below (the "*Reverse Stock Split*"); and

(ii) a number of shares of Adamis Common Stock, based on the Reverse Stock Split Ratio, held as treasury stock or held or owned by Adamis immediately before the filing of the Adamis Reverse Stock Split Amendment, shall each be reclassified and combined into and become one share of Adamis Common Stock.

(iii) As used in this Agreement, references to "Adamis Common Stock" shall, after the Effective Time of the Reverse Stock Split, refer to post-Reverse Stock Split shares of New Adamis Common Stock.

(b) <u>No Fractional Shares</u>. No fractional shares of Adamis Common Stock shall be issued in connection with the Reverse Stock Split, and no certificates or scrip representing such fractional shares shall be issued. Any holder of Adamis Common Stock who would otherwise be entitled to receive a fraction of a share of Adamis Common Stock (after aggregating all fractional shares of Adamis Common Stock issuable to such holder) shall, in lieu of such fraction of a share and upon surrender of such holder's certificate representing such fractional shares of Adamis Common Stock, instead receive from Adamis an amount of cash (rounded to the nearest whole cent), without interest, equal to the product of (1) the closing sale price per share of the Adamis Common Stock as reported by The Nasdaq Capital Market (or other exchange or market on which the Adamis Common Stock is then traded) on the last trading day preceding the date of the effective date of the Reverse Stock Split by (2) the number of pre-Reverse Stock Split shares of Adamis Common Stock held by such holder that would otherwise have been exchanged for such fractional share interests.

(c) <u>Reverse Stock Split and the Exchange Ratio</u>. The Exchange Ratio set forth herein assumes the effectiveness of the Reverse Stock Split described above.

1.6. Shares to be Issued; Effect on Capital Stock. Subject to the terms and conditions of this Agreement, at the Effective Time, by virtue of the Merger and without any action on the part of Adamis, Merger Sub, DMK or any DMK Stockholder, the following shall occur:

Conversion of DMK Common Stock. Subject to the terms of Section 1.8 and Section 1.10(h) regarding Dissenting Shares, and (a)subject to the other provisions of this Section 1.6(a), each share of DMK Common Stock issued and outstanding immediately before the Effective Time (other than any shares of DMK Common Stock to be canceled pursuant to Section 1.6(b), if any, and excluding any Dissenting Shares, to the extent provided in Section 1.8), will be converted automatically into the right to receive: (i) that number of shares of New Adamis Common Stock equal to the Exchange Ratio, and (ii) any cash, without interest, to be paid in lieu of any fractional share of DMK Common Stock in accordance with Section 1.6(e). Notwithstanding the preceding sentence, in the event that the aggregate number of shares of Adamis Common Stock that would be issuable to a particular DMK Stockholder (a "Specified DMK Stockholder") pursuant to the preceding sentence would result in such Specified DMK Stockholder beneficially owning shares of Adamis Common Stock in excess (or having voting power in excess) of 9.99% of the shares of Adamis Common Stock outstanding immediately after the Effective Time (the "Adamis Common Stock Consideration Cap"), then in lieu of the issuance to the Specified DMK Stockholder of shares of Adamis Common Stock in excess of the Adamis Common Stock Consideration Cap (such number of shares of Adamis Common Stock in excess of the Adamis Common Stock Consideration Cap referred to as the "Excess Cap Shares"), such Specified DMK Stockholder shall instead receive a number of shares of Series E Preferred (the "Merger Consideration Preferred Shares"), which Merger Consideration Preferred Shares shall be convertible, subject to the beneficial ownership limitations and other provisions set forth in the Series E Preferred Certificate of Designation, into a number of shares of Adamis Common Stock (the "Series E Preferred Conversion Shares") equal to the Excess Cap Shares, with such Merger Consideration Preferred Shares entitled to such voting rights, including voting together with the Adamis Common Stock on an as-converted basis with the number of votes to which such Merger Consideration Preferred Shares are entitled calculated as set forth in the Series E Preferred Certificate of Designation. The aggregate number of shares of Adamis Common Stock and Merger Consideration Preferred Shares (including, upon conversion of shares of Series E Preferred, the shares of Adamis Common Stock issuable upon conversion of the Merger Consideration Preferred Shares) that are issuable to the stockholders of DMK as a result of the Merger will be referred to as the "Merger Consideration" or the "Merger Consideration Shares." For purposes of the foregoing determination, unless otherwise provided in the Series E Preferred Certificate of Designation, beneficial ownership shall be calculated in accordance with Section 13(d) of the Exchange Act and the rules and regulations promulgated thereunder.

(b) <u>Cancellation of Treasury and Adamis-Owned Shares</u>. Any shares of DMK Capital Stock held as treasury stock or held or owned by DMK, Adamis or any direct or indirect wholly-owned Subsidiary of DMK or of Adamis immediately before the Effective Time shall be canceled and shall cease to exist, and no consideration shall be delivered in exchange therefor.

(c) <u>DMK Restricted Stock</u>. If any shares of DMK Common Stock issued and outstanding immediately before the Effective Time are unvested or are subject to a repurchase option or the risk of forfeiture or under any applicable restricted stock purchase agreement, stock restriction agreement, cancellation agreement or other agreement with DMK (such shares, the "*DMK Restricted Stock*"), then subject to any acceleration of vesting or lapse of repurchase options or risk of forfeiture resulting from the transactions contemplated by this Agreement, the Merger Consideration Shares issued in exchange for such shares of DMK Restricted Stock pursuant to Section 1.6(a) will to the same extent be unvested and subject to the same repurchase option or risk of forfeiture, and the certificates or, if shares are represented in uncertificated or in book-entry form, then the comparable evidence of such shares, representing such Merger Consideration Shares shall accordingly be marked with appropriate legends to reflect such repurchase option or risk of forfeiture. DMK and Adamis shall take all action that may be necessary to ensure that, from and after the Effective Time, Adamis is entitled to exercise any such repurchase option or right of cancellation or other right set forth in any such restricted stock purchase agreement or other agreement that will continue after the Effective Time.

(d) <u>Capital Stock of Merger Sub</u>. Each share of common stock of Merger Sub issued and outstanding immediately before the Effective Time shall remain outstanding and shall represent one validly issued, fully paid and nonassessable share of common stock of the Surviving Corporation. Each stock certificate of Merger Sub evidencing ownership of any such shares shall, as of the Effective Time, evidence ownership of such shares of common stock of the Surviving Corporation.

(e) <u>No Fractional Shares</u>. No fractional shares of Adamis Common Stock shall be issued in connection with the Merger, and no certificates or scrip representing such fractional shares shall be issued. Each holder of shares of DMK Common Stock who would otherwise be entitled to receive a fraction of a share of Adamis Common Stock (after aggregating all fractional shares of Adamis Common Stock to be received by such holder) shall, in lieu of such fraction of a share and upon surrender of such holder's certificate representing shares of DMK Capital Stock (the "*DMK Stock Certificate*"), instead receive from Adamis an amount of cash (rounded to the nearest whole cent), without interest and subject to applicable Tax withholding, equal to (1) the Adamis Average Closing Price, multiplied by (2) such fraction of a share of New Adamis Common Stock that such holder would otherwise be entitled to receive.

(f) <u>Unregistered Shares</u>. The issuance of the Merger Consideration Shares of Adamis capital stock to be issued to the DMK Stockholders shall not be registered under the Securities Act of 1933, as amended (the "*Securities Act*"), in reliance upon the private placement exemption under Regulation D and/or Section 4(a)(2) of the Securities Act. As described further below, after the Closing, Adamis will endeavor to file a registration statement (or a prospectus supplement) covering the resale, from time to time, of the shares of Adamis Common Stock issued (or issuable) as Merger Consideration (including without limitation the Series E Preferred Conversion Shares), and in the case of a registration statement use reasonable efforts to cause such registration statement to become effective.

1.7. <u>Calculation of the Exchange Ratio</u>. For purposes of this Agreement, the "*Exchange Ratio*" shall be a number of shares of New Adamis Common Stock issuable in exchange for one (1) share of DMK Common Stock outstanding immediately before the Effective Time, with such number of shares of Adamis Common Stock determined as follows: subject to Section 1.6(a), (1) the DMK Valuation, divided by (2) the Adamis Average Closing Price (with the result rounded to two (2) decimal places or such other number of places as Adamis and DMK may agree), and then (3) divided by the number of DMK Outstanding Shares (with the result rounded to six (6) decimal places, or such other number of places as Adamis and DMK may agree), in which:

(a) "DMK Valuation" means \$27,000,000.00;

(b) "Adamis Average Closing Price" means the average, for the five (5) trading days ending one (1) trading day before the effective date of the Merger, of (i) for trading days commencing on the first trading day after the effective date of the Reverse Stock Split, the Closing Price of the Adamis Common Stock, and (ii) for trading days on or before the effective date of the Reverse Stock Split, the Closing Price multiplied by the Reverse Stock Split Ratio;

(c) "*Closing Price*" means, with respect to any particular trading day, the last reported sale price of the Adamis Common Stock at the 4:00 p.m., Eastern Time, end of regular trading hours on the Nasdaq Capital Market (or other principal exchange, market or quotation system on which the Adamis Common Stock is then-listed or traded) (and if closing sale prices are not reported on such other market or quotation system, then the average of the high bid and low asked prices for such trading days); and

(d) *"DMK Outstanding Shares"* means the total number of shares of DMK Capital Stock outstanding immediately prior to the Effective Time, assuming, without limitation or duplication, the conversion of any DMK Convertible Notes into DMK Common Stock;

provided, however, that notwithstanding the foregoing, if the calculation of the Exchange Ratio as provided above would result in the holders of Adamis Common Stock immediately before the Effective Time owning less than 50.1% of the aggregate of (i) the number of shares of Adamis Common Stock held by such stockholders immediately after the Effective Time, plus (ii) the number of shares of Adamis Common Stock issuable to the DMK Stockholders pursuant to the Exchange Ratio as calculated above (including shares issuable upon conversion of the Merger Consideration Preferred Shares determined without regard to beneficial ownership limitations), plus (iii) the number of shares of Adamis Common Stock that are issuable upon exercise of DMK Options assumed by Adamis pursuant to this Agreement (the "*Adamis Percentage Threshold*"), then the number of shares constituting the Merger Consideration (determined on an as-converted basis including shares issuable upon conversion of the Merger Consideration Preferred Shares) shall be a number such that the holders of Adamis Common Stock immediately before the Effective Time hold a number of shares of Adamis Common Stock immediately after the Effective Time equal to the Adamis Percentage Threshold.

1.8. Dissenting Shares. Notwithstanding any other provision of this Agreement to the contrary, only to the extent that dissenters' rights or appraisal rights pursuant to the NJBCA apply to the transactions contemplated by this Agreement and that DMK Stockholders have dissenters rights or appraisal rights under the NJBCA with respect to the Merger and the transactions contemplated by this Agreement, then any shares of DMK Capital Stock that are outstanding immediately prior to the Effective Time and which are held by stockholders who have exercised and perfected applicable dissenters' rights or appraisal rights for such shares of DMK Capital Stock in accordance with the NJBCA (such shares referred to as "Dissenting Shares"), shall not be converted into or represent a right to receive Adamis Common Stock pursuant to Section 1.6(a), but instead shall be converted in to the right to receive such consideration as may be determined to be due with respect to such Dissenting Shares pursuant to the NJBCA. If a holder of Dissenting Shares (a "Dissenting Stockholder") withdraws such holder's demand for such payment and appraisal or becomes ineligible for such payment and appraisal, then, as of the later of the Effective Time or the date of which such Dissenting Stockholder withdraws such demand or otherwise becomes ineligible for such payment and appraisal, such holder's Dissenting Shares will cease to be Dissenting Shares and will be converted into the right to receive shares of Adamis Common Stock attributable to such Dissenting Shares as determined in accordance with Section 1.6(a). DMK will give Adamis prompt notice of any demands received by DMK for appraisal of shares of DMK Capital Stock, withdrawals of such demands, and any other instruments that relate to such demands received by DMK. Adamis and DMK shall jointly participate in all negotiations and proceedings with respect to such demands except as limited by applicable Law. Neither Adamis nor DMK will, except with prior written consent of the other, make any payment with respect to, or settle or offer to settle, any such demands, unless and to the extent required to do so under applicable Law.

1.9. No Further Transfer of DMK Capital Stock. At the Effective Time, all shares of DMK Capital Stock outstanding immediately before the Effective Time shall automatically be exchanged, and all holders of DMK Capital Stock that were outstanding immediately before the Effective Time shall cease to have any rights as stockholders of DMK, except the right to receive the consideration described in Section 1.6(a) or Section 1.8, as applicable. No further transfer of any such shares of DMK Capital Stock shall be made on such stock transfer books after the Effective Time. Subject to Section 1.10(f), if, after the Effective Time, any shares of DMK Capital Stock are presented to DMK or Adamis (or the Exchange Agent), such shares of DMK Capital Stock shall be canceled and shall be exchanged as provided in Section 1.10.

1.10. Exchange of Certificates.

(a) <u>Exchange Agent</u>. The transfer agent for Adamis, or such other bank, trust company or transfer agent as Adamis may select and designate before the Effective Time (the "*Exchange Agent*"), will act as agent of Adamis for such purposes as Adamis may determine, which may include purposes of mailing and receiving transmittal letters and distributing the Merger Consideration to the holders of DMK Common Stock. In the alternative, Adamis may perform some actions not undertaken by the Exchange Agent.

(b) <u>Adamis to Provide Common Stock</u>. Promptly after the Effective Time, Adamis shall supply or cause to be supplied or made available to the Exchange Agent for exchange in accordance with this Section 1.10, through such procedures as Adamis may adopt, instructions regarding issuance of certificates (or evidence of issuance in uncertificated or book-entry form) evidencing the shares of Adamis Common Stock (or Merger Consideration Preferred Shares, as the case may be) issuable as the Merger Consideration (sometimes referred to as the "*Exchange Shares*").

Exchange Procedures. As promptly as practicable after the Effective Time, Adamis or the Exchange Agent will mail to each (c) holder of record of DMK Capital Stock whose shares would be converted into the right to receive shares of the Merger Consideration pursuant to Section 1.6(a): (i) a letter of transmittal in customary form mutually agreeable to DMK and Adamis; (ii) such other customary documents as may be required pursuant to such instructions; and (iii) instructions for use in effecting the surrender of DMK Capital Stock in exchange for certificates representing shares of Adamis constituting Merger Consideration (or evidence of shares in uncertificated or book-entry form). Upon surrender of DMK Capital Stock for cancellation to the Exchange Agent, together with such letter of transmittal and other documents, duly completed and validly executed in accordance with the instructions thereto, the holder of such DMK Capital Stock shall be entitled to receive in exchange therefor, (x) a certificate (or evidence of shares in uncertificated or book-entry form) representing the number of whole Exchange Shares into which the DMK Common Stock represented thereby shall have been converted into the right to receive as of the Effective Time, (y) any dividends or other distributions to which such holder is entitled pursuant to Section 1.10(d), and (z) cash in respect of any fractional shares as provided in Section 1.6(e), and the DMK Capital Stock so surrendered shall forthwith be canceled. Until so surrendered, each such outstanding share of DMK Capital Stock will be deemed from and after the Effective Time, for all corporate purposes other than the payment of dividends, to evidence the ownership of the number of full Exchange Shares into which such shares of DMK Capital Stock shall have been so converted and the right to receive cash in lieu of the issuance of any fractional shares. If any DMK Stock Certificate shall have been lost, stolen or destroyed, Adamis may, in its discretion and as a condition precedent to the issuance of any certificate (or evidence of shares in uncertificated or book-entry form) representing Merger Consideration, require the owner of such lost, stolen or destroyed DMK Stock Certificate to provide a reasonable affidavit as indemnity against any claim that may be made against the Exchange Agent, Adamis or the Surviving Corporation with respect to such DMK Stock Certificate.

(d) <u>Distributions With Respect to Unexchanged Shares</u>. No dividends or other distributions with respect to Adamis Common Stock with a record date after the Effective Time will be paid to the holder of any unsurrendered DMK Capital Stock with respect to the Merger Consideration consisting of Adamis Common Stock represented thereby until the holder of record of such DMK Capital Stock shall surrender such shares of DMK Capital Stock. Subject to applicable law, following surrender of any such DMK Capital Stock, there shall be delivered to the record holder of DMK Capital Stock a certificate representing whole shares of Adamis Common Stock or shares of Merger Consideration Preferred Shares, as applicable, issued in exchange therefor (or evidence of shares in uncertificated or book-entry form) (including any cash in respect of any fractional shares), without interest at the time of such surrender, and the amount of any such dividends or other distributions with a record date after the Effective Time theretofore payable (but for the provisions of this Section) with respect to such shares of Adamis Common Stock.

(e) <u>Transfers of Ownership</u>. If any certificate for Exchange Shares (or evidence of shares in uncertificated or book-entry form) is to be issued in a name other than that in which DMK Stock Certificate surrendered in exchange therefor is registered, it will be a condition of the issuance thereof that the DMK Capital Stock so surrendered will be properly endorsed and otherwise in proper form for transfer and that the person requesting such exchange will have paid to Adamis or any agent designated by it any transfer or other taxes required by reason of the issuance of a certificate for Exchange Shares in any name other than that of the registered holder of the DMK Capital Stock surrendered, or established to the satisfaction of Adamis or any agent designated by it that such tax has been paid or is not payable, and shall provide such written assurances regarding federal and state securities law compliance as Adamis may reasonably request, including, without limitation, an opinion of counsel to such DMK holder.

(f) <u>Termination of Exchange Shares</u>. Any Exchange Shares which remain undistributed to the stockholders of DMK twelve (12) months after the Effective Time shall be delivered to Adamis, upon demand, and any stockholders of DMK who have not previously complied with this Section shall thereafter look only to Adamis for payment of their claim for their portion of the Exchange Shares and any dividends or distributions with respect to the Exchange Shares.

(g) <u>No Liability</u>. Notwithstanding anything to the contrary in this Section, none of the Exchange Agent, Adamis, DMK or any party hereto shall be liable to any person for any amount properly paid to a public official pursuant to any applicable abandoned property, escheat or similar Legal Requirement.

(h) <u>Dissenting Shares</u>. The provisions of this Section shall also apply to Dissenting Shares that lose their status as such, except that the obligations of Adamis under this Section shall commence on the date of loss of such status and the holder of such shares shall be entitled to receive in exchange such shares to which such holder is entitled pursuant to Section 1.6.

1.11. Further Action. If, at any time after the Effective Time, any further action that is commercially reasonable and lawful is determined by Adamis and DMK to be necessary or appropriate to carry out the purposes of this Agreement or to vest Adamis with full right, title and possession of all shares of DMK Capital Stock, then the officers and directors of DMK and Adamis shall be fully authorized (in the name of DMK and/or Adamis or otherwise) to take such action.

1.12. Tax Consequences; Withholding. For U.S. federal (and applicable state and local) income tax purposes, the Merger is intended to constitute a reorganization within the meaning of Section 368(a) of the Code (the "U.S. Tax Treatment"). The Parties hereby (a) adopt this Agreement insofar as it relates to the Merger as a "plan of reorganization" within the meaning of Treasury Regulations Section 1.368-2(g), (b) agree to file and retain such information as shall be required under Treasury Regulations Section 1.368-3, and (c) agree to file all Tax Returns on a basis consistent with the U.S. Tax Treatment, unless otherwise required by a "determination" that is final within the meaning of Section 1313(a) of the Code. Notwithstanding the foregoing or anything else to the contrary contained in this Agreement, the Parties acknowledge and agree that, other than the representations set forth in Sections 2.13(i) and 3.13(i), no Party is making any representation or warranty as to the qualification of the Merger as a reorganization under Section 368(a) of the Code or as to the effect, if any, that any transaction consummated on, after or prior to the Effective Time has or may have on any such reorganization status. Each of the Parties acknowledges and agrees that each such Party (A) has had the opportunity to obtain independent legal and tax advice with respect to the transactions contemplated by this Agreement, and (B) is responsible for paying its own Taxes, including any adverse Tax consequences that may result if the Merger is determined not to qualify as a reorganization under Section 368(a) of the Code.

1.13. <u>Plan of Merger</u>. Article I and Article II hereof and, solely to the extent necessary under the NJBCA, the other provisions of this Agreement shall constitute a "plan of merger" for the purposes of the NJBCA, including Section 14A.10-1 thereof. In addition, the Parties agree to prepare, adopt and approve a mutually agreeable plan of merger for purposes of the DGCL and the NJBCA, consistent with the provisions of this Agreement.

ARTICLE II REPRESENTATIONS AND WARRANTIES OF DMK

DMK represents and warrants to Adamis that the statements contained in this Article II are true and correct as set forth herein and as qualified by the disclosure schedules separately delivered to Adamis concurrently herewith (collectively, the "*DMK Disclosure Schedule*"). The disclosures set forth in the DMK Disclosure Schedule shall be arranged in paragraphs corresponding to the numbered and lettered paragraphs contained in this Article II. The disclosures in any section or subsection of the DMK Disclosure Schedule shall qualify other sections and subsections in this Article II to the extent it is reasonably apparent from a reading of the disclosure that such disclosure is applicable to such other sections and subsections.

2.1. Organization and Qualification; Charter Documents.

(a) DMK is a corporation duly organized, validly existing, and in good standing under the laws of its jurisdiction of incorporation, with requisite corporate power and authority to conduct its business as now being conducted and to own or use its properties and assets. DMK is duly qualified to do business as a foreign corporation and is in good standing under the laws of each state or other jurisdiction in which either the ownership or use of the properties owned or used by it, or the nature of the activities conducted by it, requires such qualification, except where the failure to be so qualified or in good standing would not have a Material Adverse Effect on DMK.

(b) DMK has made available to Adamis accurate and complete copies of: (i) the certificate of incorporation, bylaws and other charter and organizational documents of DMK, including all amendments thereto; (ii) the stock records of DMK; and (iii) the minutes and other records of the meetings and other proceedings (including any actions taken by written consent or otherwise without a meeting) of the stockholders of DMK and the board of directors or any committee thereof of DMK. The stock records, minute books and other records of DMK are accurate, up-to-date and complete in all material respects.

2.2. <u>Subsidiaries</u>. DMK does not have any direct or indirect Subsidiaries. Except as set forth in the DMK Disclosure Schedule, DMK does not otherwise own any shares in the capital of or any interest in, or control, directly or indirectly, any corporation, partnership, limited liability company, association, joint venture or other business entity (each an "*Entity*"). The DMK Disclosure Schedule sets forth the name and share ownership of any parent entity that owns more than 50% of the outstanding shares of DMK and, for such entity, the names of the shareholders, members, managers, or other equity owners of such entity.

2.3. Authority; Vote Required.

(a) DMK has all requisite corporate power and authority to enter into this Agreement and the other agreements to which it is a party that this Agreement requires to be entered into in connection with the transactions contemplated hereby (collectively, the "Ancillary Agreements"), to perform its obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby. The execution and delivery of this Agreement and the Ancillary Agreements and the consummation of the transactions contemplated hereby and thereby have been duly authorized by all necessary corporate action on the part of DMK, subject only to the approval of this Agreement by the stockholders of DMK. The DMK Disclosure Schedule accurately sets forth (i) the names of the members of the board of directors of DMK, and (ii) the names and titles of the officers of DMK. The Board of Directors of DMK has unanimously approved this Agreement constitutes (and the Ancillary Agreements will constitute at the Closing) duly executed and delivered by DMK, and this Agreement constitutes (and the Ancillary Agreements will constitute at the Closing) the valid and binding obligation of DMK enforceable against DMK in accordance with their terms, except that such enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting or relating to creditors' rights generally, and is subject to general principles of equity.

(b) The affirmative vote (or written consent) of the holders of a majority of the shares of DMK Common Stock outstanding on the record date for the DMK Stockholders Meeting (or, if the DMK Stockholders act by written consent, then the record date for such action by written consent) and entitled to vote thereon (the "*Required DMK Stockholder Vote*"), is the only vote (or written consent) of the holders of any class or series of DMK Capital Stock necessary to adopt and approve this Agreement and approve the Contemplated Transactions.

2.4. <u>No Conflict</u>. The execution and delivery by DMK of this Agreement and the Ancillary Agreements to which DMK is a party, does not, and the consummation of the transactions contemplated hereby and thereby will not, (a) conflict with, or result in any material violation of, any provision of the DMK Charter (in its current form and as it may be amended immediately before the Effective Time) or the DMK Bylaws, (b) result in any material violation of or default under (with or without notice or lapse of time, or both), or give rise to a right of termination, cancellation or acceleration of any obligation or loss of any benefit under, any material mortgage, indenture, lease, contract, grant, funding arrangement, or other agreement or instrument, permit, concession, franchise or license of DMK, (c) subject to obtaining the approval of DMK's stockholders and, except as would not reasonably be expected to have a Material Adverse Effect on DMK, conflict with, or result in any violation of any judgment, order, decree, statute, law, ordinance, rule or regulation applicable to DMK or any of its properties or assets, or (d) conflict with, or result in a violation of any resolution adopted by DMK's stockholders, DMK's board of directors or any committee of DMK's board of directors.

2.5. <u>Non-Contravention; Consents</u>.

(a) Subject to compliance with obtaining the Required DMK Stockholder Vote and the filing of the Certificate of Merger with the Secretary of State of Delaware pursuant to the DGCL and the Secretary of State of New Jersey pursuant to the NJBCA, neither (x) the execution, delivery or performance of this Agreement by DMK, nor (y) the consummation of the Contemplated Transactions, will directly or indirectly (with or without notice or lapse of time):

(i) contravene, conflict with or result in a material violation of, or give any Governmental Entity or other Person the right to challenge the Contemplated Transactions or to exercise any material remedy or obtain any material relief under, any Law or any Order by which DMK, or any of the material assets owned or used by DMK, is subject;

(ii) contravene, conflict with or result in a material violation of any of the terms or requirements of, or give any Governmental Entity the right to revoke, withdraw, suspend, cancel, terminate or modify, any Governmental Authorization that is held by and material to DMK;

(iii) contravene, conflict with or result in a violation or breach of, or result in a default under, any provision of any DMK Material Contract, or give any Person the right to: (A) declare a default or exercise any remedy under any DMK Material Contract, (B) any material payment, rebate, chargeback, penalty or change in delivery schedule under any DMK Material Contract, (C) accelerate the maturity or performance of any DMK Material Contract, or (D) cancel, terminate or modify any term of any DMK Material Contract, except in each of the above cases as would not reasonably be expected to have a Material Adverse Effect on DMK; or

(iv) result in the imposition or creation of any material Encumbrance upon or with respect to any asset owned or used by DMK (except for Permitted Encumbrances) that is material to its business.

(b) Except for (i) the Required DMK Stockholder Vote, (ii) the filing of the Certificate of Merger with the Secretary of State of Delaware pursuant to the DGCL and the Secretary of State of the State of Delaware pursuant to the NJBCA, (iii) such filings as may be required under applicable securities laws, and (iv) such consents, waivers, approvals, orders, authorizations, registrations, declarations and filings which, if not obtained or made, would not have a Material Adverse Effect on DMK, DMK is not and will not be required to make any filing with or give any notice to, or to obtain any waiver or Consent from, any Governmental Entity or Person in connection with (x) the execution, delivery or performance of this Agreement, or (y) the consummation of the Contemplated Transactions.

2.6. <u>Governmental Authorizations</u>.

(a) DMK has obtained each material federal, state, county, local or foreign governmental consent, license, permit, grant, or other authorization of a Governmental Entity (i) pursuant to which DMK currently operates or holds any interest in any of its properties, or (ii) that is required for the operation of DMK's business or the holding of any such interest, and all of such authorizations are in full force and effect, except for such consents, licenses, permits, grants or other authorizations, which if not obtained would not have a Material Adverse Effect on DMK.

(b) To DMK's Knowledge, neither it nor any of its officers or employees are: (i) a party to any material action, suit, proceeding, investigation or lawsuit ("Action") involving DMK's activities in the healthcare industry; (ii) subject to any Actions or sanctions by any payor (including Medicare or Medicaid); and (iii) listed by a federal agency as excluded, disbarred, suspended, or otherwise ineligible to participate in federal programs, including Medicare or Medicaid, or listed on the General Service Administration list of parties excluded from Federal Procurement and Non-Procurement Programs.

2.7. <u>Capitalization</u>.

(a) The authorized capital stock of DMK consists of 60,000 shares of DMK Common Stock, \$0.001 par value, and no shares of DMK Preferred Stock, of which there were issued and outstanding, as of the date of this Agreement, 38,837 shares of DMK Common Stock. The names and record and beneficial owners of DMK Capital Stock, which constitute all of the outstanding capital stock of the Company, and the number of shares held by each owner, are set forth in Section 2.7 of the DMK Disclosure Schedule. With respect to any such owner that is an entity, the DMK Disclosure Schedule sets forth the shareholders, members or other equity owners of such entity as well as the board of directors, officers, or managers of such entity. Except as set forth in Section 2.7 of the DMK Disclosure Schedule, (i) each owner of DMK Capital Stock, and each holder of a DMK Convertible Note, is an "accredited investor" as defined in Rule 501 of Regulation D promulgated under the Securities Act. All of the outstanding shares of DMK Capital Stock (i) have been duly authorized and validly issued, and are fully paid and non-assessable, (ii) except for rights of first refusal, exchange, repurchase, forfeiture and/or cancellation rights in favor of DMK, are not subject to preemptive rights or rights of first refusal created by statute, the DMK Charter, the DMK Bylaws or any agreement to which DMK is a party or by which it is bound, and (iii) to the Knowledge of DMK, have been issued in compliance in all material respects with federal and state securities laws. There are no declared or unpaid dividends with respect to any shares of DMK Capital Stock. Except as set forth in the DMK Disclosure Schedule, there are no issued or outstanding DMK Options, convertible securities (whether debt or equity), or other rights of any kind entitling any person to purchase or acquire shares of DMK Capital Stock, and DMK has not adopted any stock option plan or similar employee benefit plan pursuant to which equity securities of DMK may be issued.

Except for DMK's 2016 Stock Plan, as amended (the "DMK Plan"), DMK does not have any stock option plan or any other (b) plan, program, agreement or arrangement providing for any equity-based compensation for any Person. As of the date of this Agreement, DMK has reserved 4,850 shares of DMK Common Stock for issuance under DMK Plan, none of which have been issued or are currently outstanding, 4,500 shares have been reserved for issuance upon exercise of DMK Options granted under DMK Plan, and 350 shares of DMK Common Stock remain available for future issuance pursuant to DMK Plan. The DMK Disclosure Schedule sets forth the following information with respect to each DMK Option outstanding as of the date of this Agreement: (i) the name of the optionee; (ii) the number of shares of DMK Common Stock subject to such DMK Option at the time of grant; (iii) the number of shares of DMK Common Stock subject to such DMK Option as of the date of this Agreement; (iv) the per share exercise price of such DMK Option; (v) the date on which such DMK Option was granted; (vi) the applicable vesting schedule, including any acceleration provisions, and the number of vested and unvested options as of the date of this Agreement; (vii) the expiration date of such DMK Option, and (viii) whether such DMK Option is intended to be an "incentive stock option" (as defined in the Code) or a non-qualified stock option. DMK has not granted or awarded, and as of the date of this Agreement there are not outstanding, any shares of restricted stock, any stock appreciation right or restricted stock unit, or any other derivative security or security exercisable or exchangeable for or convertible into shares of DMK Common Stock (a "DMK Other Award"). DMK has made available to Adamis an accurate and complete copy of DMK Plan, forms of all award agreements evidencing outstanding equity awards thereunder, any equity award agreements that differ in any material respect from such forms of award agreements and evidence of Board and stockholder approval of DMK Plan and any amendments thereto. Except as disclosed in the DMK Disclosure Schedule, no vesting of DMK Options will accelerate in connection with the closing of the Contemplated Transactions.

(c) The DMK Disclosure Schedule sets forth the following information with respect to each outstanding convertible promissory note or other convertible debt of DMK (collectively, the "DMK Convertible Notes"): (i) the name of each holder of a DMK Convertible Note; (ii) the outstanding balance, accrued interest as of the end of the month preceding the month of the Agreement Date, and interest rate applicable to each DMK Convertible Note; (iii) a brief description of the conversion terms applicable to each DMK Convertible Note; (iv) stated maturity date of each DMK Convertible Note; and (v) any other material terms of such DMK Convertible Note.

(d) Except for the outstanding DMK Options and DMK Convertible Notes set forth in the DMK Disclosure Schedule, there is no: (i) outstanding subscription, option, call, warrant or right (whether or not currently exercisable) to acquire any shares of the capital stock or other securities of DMK, (ii) outstanding security, instrument or obligation that is or may become convertible into or exchangeable for any shares of the capital stock or other securities of DMK, (iii) stockholder rights plan (or similar plan commonly referred to as a "poison pill") or Contract under which DMK is or may become obligated to sell or otherwise issue any shares of its capital stock or any other securities, or (iv) condition or circumstance that may give rise to or provide a basis for the assertion of a claim by any Person to the effect that such Person is entitled to acquire or receive any shares of capital stock or other securities of DMK. There are no outstanding or authorized stock appreciation, phantom stock, profit participation or other similar rights with respect to DMK. No consent of the holders of DMK Options is required in connection with the actions contemplated by this Agreement.

(e) All outstanding shares of DMK Common Stock, DMK Options, DMK Convertible Notes, and other securities of DMK have been issued and granted in material compliance with (i) all applicable securities laws and other applicable Law, and (ii) all requirements set forth in applicable Contracts.

(f) With respect to DMK Options, (i) each grant was duly authorized no later than the date on which the grant of such DMK Option was by its terms to be effective (the "*Grant Date*") by all necessary corporate action, (ii) each grant purported to be made under the DMK Plan was made in all material respects in accordance with the terms of DMK Plan, and (iii) the per share exercise price of each DMK Option was not less than the fair market value of a share of DMK Common Stock on the applicable Grant Date determined in a manner consistent with Section 409A of the Code, has not otherwise been subject to "modification" or "extension" within the meaning of Section 409A of the Code, and does not have any feature for the deferral of compensation other than the deferral of recognition of income until the later of exercise or disposition of such option or SAR.

2.8. <u>Title to Assets</u>.

(a) DMK owns, and has good, valid and marketable title to, all material assets purported to be owned by it, including all assets reflected on the Current Balance Sheet and all other material assets reflected in DMK's books and records as being owned by DMK. All of said assets are owned by DMK free and clear of any liens or other Encumbrances, except for (x) any lien for current taxes not yet due and payable, and (y) minor liens that have arisen in the ordinary course of business and that do not (in any case or in the aggregate) materially detract from the value of the assets subject thereto or materially impair the operations of DMK.

Except as disclosed on the DMK Disclosure Schedule, DMK does not own any real property and DMK is not party to any material lease for real property either as a lessee or lessor.

2.9. DMK Financial Statements.

(a) DMK has furnished to Adamis copies of (i) unaudited combined balance sheets of DMK as of December 31, 2021 and 2020, and the related combined statement of operations, statement of changes in stockholders' equity and statement of cash flows for the 12-months ended December 31, 2021 and 2020 (the "*DMK Annual Financial Statements*"), and (ii) unaudited combined balance sheets of DMK as of September 30, 2022, and the related combined statement of operations, statement of changes in stockholders' equity and statement of cash flows for the nine months ended September 30, 2022 (the "*DMK Interim Financial Statements*") (the financial statements in clauses (i) through (ii) above, together with the DMK Audited Financial Statements are and will be accurate and complete in all material respects, have been prepared in accordance with GAAP consistently applied and present fairly the financial position of DMK as of the dates thereof, and the results of its operations for the respective periods then ended, subject to normal and recurring year-end adjustments and the absence of notes and, in the case of the DMK Interim Financial Statements is referred to herein as the "*Current Balance Sheet*."

(b) DMK maintains a system of internal accounting controls designed to provide reasonable assurance that: (i) transactions are executed in accordance with management's general or specific authorizations; (ii) transactions are recorded as necessary to permit preparation of the financial statements of DMK in conformity with GAAP and to maintain accountability of DMK's assets; (iii) access to DMK's assets is permitted only in accordance with management's general or specific authorization; and (iv) the recorded accountability for DMK's assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any differences. DMK maintains internal control over financial reporting that provides reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with GAAP.

(c) The DMK Disclosure Schedule lists, and DMK has delivered to Adamis, accurate and complete copies of the documentation creating or governing, all securitization transactions and "off-balance sheet arrangements" (as defined in Item 303(c) of Regulation S-K under the Exchange Act) effected by DMK since January 1, 2020.

(d) Neither DMK nor its independent auditors have identified (i) any significant deficiency or material weakness in the design or operation of the system of internal accounting controls utilized by DMK, (ii) any fraud, whether or not material, that involves DMK, DMK's management or other employees who have a role in the preparation of financial statements or the internal accounting controls utilized by DMK, or (iii) any claim or allegation regarding any of the foregoing, in each case since January 1, 2020.

2.10. <u>Absence of Certain Changes</u>. Since September 30, 2022 (the "*Base Date*"), except as set forth in the DMK Disclosure Schedule, DMK has conducted its business only in the Ordinary Course of Business (except for matters relating to the execution and performance of this Agreement and the discussions, negotiations and transactions related thereto) and there has not occurred any change, event or condition (whether or not covered by insurance) that has resulted in, or would reasonably be expected to result in, a Material Adverse Effect on DMK.

2.11. <u>Interested Party Transactions</u>. DMK is not indebted to any director, officer or employee of DMK (except for amounts due as normal salaries and bonuses and in reimbursement of ordinary expenses and as set forth in the DMK disclosure schedule with regards to convertible notes), and no such person is indebted to DMK. DMK is not a party to any transaction involving over \$120,000 in which any director, officer or 5% stockholder of DMK (or a member of such person's immediate family) had a direct or indirect material interest, except where such person's interest arises solely from his or her ownership of DMK Capital Stock or convertible note. The DMK Disclosure Schedule describes any material transactions or relationships, since January 1, 2019, between, on one hand, DMK and, on the other hand, any (a) executive officer or director of DMK or any of its Subsidiaries or any of such executive officer's or director's immediate family members, (b) owner of more than five percent (5%) of the voting power of the outstanding DMK Capital Stock or (c) to the Knowledge of DMK, any "related person" (within the meaning of Item 404 of Regulation S-K under the Securities Act) of any such officer, director or owner (other than DMK) in the case of each of (a), (b) or (c) that is of the type that would be required to be disclosed under Item 404 of Regulation S-K under the Securities Act.

2.12. Intellectual Property.

(a) DMK owns or possesses the right to use the Intellectual Property utilized in connection with the conduct of its business and intended business, all of which is owned by or licensed to DMK (the "**DMK Patent and Proprietary Rights**"), except where the failure to own or possess such rights would not have a Material Adverse Effect on DMK. Such DMK Patent and Proprietary Rights are sufficient in all material respects for the conduct of DMK's business and proposed business. Concurrently with the execution hereof, DMK has provided to Adamis an accurate, true and complete listing of all DMK Registered IP and other DMK Patent and Proprietary Rights. Each Person who is or was an employee or contractor of DMK and who is or was involved in the creation or development of any material DMK Patent and Proprietary Rights purported to be owned by DMK has signed a valid, enforceable agreement containing a present assignment of such Intellectual Property to DMK and confidentiality provisions protecting trade secrets and confidential information of DMK. DMK has taken all reasonable measures to protect and maintain the confidentiality of the trade secrets included in the DMK Patent and Proprietary Rights.

(b) Since January 1, 2020, except as disclosed in the DMK Disclosure Schedule, DMK has not received any written notice of any asserted rights with respect to any of DMK Patent and Proprietary Rights which, if determined unfavorably with respect to the interests of DMK would have a Material Adverse Effect on DMK. DMK is not bound by, and no DMK Patent and Proprietary Rights are subject to, any Contract containing any covenant or other provision that in any material respect limits or restricts the ability of DMK to use, exploit, assert, defend, or enforce any DMK Patent and Proprietary Rights.

(c) To DMK's Knowledge, DMK has never infringed (directly, contributorily, by inducement, or otherwise), misappropriated, or otherwise violated or made unlawful use of any right to Intellectual Property of any other Person or engaged in unfair competition, which infringement, misappropriation, violation or use (if the subject of any unfavorable decision, ruling or finding), individually or in the aggregate, would result in a Material Adverse Effect on DMK. No material infringement, misappropriation, or similar claim or Legal Proceeding is pending or, to DMK's Knowledge, threatened against DMK, or any other Person who is or may be entitled to be indemnified, defended, held harmless, or reimbursed by DMK with respect to such claim or Legal Proceeding, which claim or Legal Proceeding (if the subject of any unfavorable decision, ruling or finding), individually or in the aggregate, would result in a Material Adverse Effect on DMK.

(d) To DMK's Knowledge, DMK has not engaged in patent or copyright misuse or any fraud or inequitable conduct in connection with any DMK Patent and Proprietary Rights, and no trademark or trade name owned, used, or applied for by DMK conflicts or interferes in any material respect with any trademark or trade name owned, used, or applied for by any other Person.

(e) DMK is in compliance in all material respects with the terms of all representations, warranties, covenants and other obligations contained in any funding grant, funding agreement, funding program or other funding arrangement with or from any Governmental Entity, non-governmental organization, foundation or other similar entity the proceeds of which were used, directly or indirectly, to develop or create, in whole or in part, any DMK Patent and Proprietary Rights, DMK Registered IP or other DMK Intellectual Property rights, and is in compliance in all material respect with the terms of all applicable Laws relating to any such grant, agreement, program or arrangement.

2.13. <u>Taxes</u>.

(a) DMK has prepared and timely filed all material Tax Returns relating to any and all Taxes concerning or attributable to DMK, and such Tax Returns are true and correct in all material respects and have been completed in accordance with applicable law in all material respects. DMK has delivered or made available to Adamis correct and complete copies of all federal income Tax Returns, examination reports, and statements of deficiencies assessed against or agreed to by DMK filed or received since January 1, 2020 or, if more recent, since inception.

(b) DMK (i) is not delinquent in any material respects in the payment of any Taxes due and owing by DMK, and (ii) has withheld and timely paid all Taxes required to have been withheld and paid with respect to any amounts paid or owing to any employee, independent contractor, creditor, stockholder, or other third party.

(c) There is no material Tax deficiency outstanding or assessed or, to DMK's Knowledge, proposed against DMK that is not reflected as a liability on the Current Balance Sheet, nor has DMK executed any agreements or waivers extending any statute of limitations on or extending the period for the assessment or collection of any Tax (other than extensions which have expired). No claim (other than claims that have been resolved) has ever been made by an authority in a jurisdiction where DMK does not file Tax Returns that it is or may be subject to taxation by that jurisdiction. There are no liens for Taxes (other than Taxes not yet due and payable) upon any of the assets of DMK.

(d) To the Knowledge of DMK, DMK has no material liabilities for unpaid Taxes that have not been accrued for or reserved on the Current Balance Sheet, whether asserted or unasserted, contingent or otherwise.

(e) DMK has not received from any Governmental Entity any (i) written notice indicating an intent to open an audit or other review, (ii) request for information related to Tax matters, or (iii) written notice of deficiency or proposed adjustment of or any amount of Tax proposed, asserted, or assessed by any Governmental Entity against DMK, in each case other than such as have been resolved before the date of this Agreement.

(f) DMK is not a party to any tax-sharing agreement or similar arrangement with any other party, and DMK has not assumed any obligation to pay any Tax obligations of, or with respect to any transaction relating to, any other person or agreed to indemnify any other person with respect to any Tax.

(g) DMK has not been a member of an affiliated group of corporations filing a consolidated federal income tax return other than a group of which DMK was the parent.

(h) DMK has not been at any time a United States Real Property Holding Corporation within the meaning of Section 897(c)(2) of the Code during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code.

(i) DMK is not aware of any facts, and has not knowingly taken or agreed to take or refrain from taking any action, in each case, that would reasonably be expected to prevent or impede the Merger from qualifying as a "reorganization" within the meaning of Section 368(a) of the Code.

(j) DMK has not filed a consent under Section 341(f) of the Code concerning collapsible corporations. DMK is not a party to any contract, agreement, plan or arrangement, including but not limited to the provisions of this Agreement, covering any employee or former employee of DMK that, individually or collectively, could give rise to the payment of (i) any "excess parachute payment" within the meaning of Section 280G of the Code (or any corresponding provisions of state, local or foreign Tax law) and (ii) any amount that will not be fully deductible as a result of Section 162(m) of the Code (or any corresponding provisions of state, local or foreign Tax law).

(k) DMK will not be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion there) ending after the Closing Date as a result of any: (A) change in method of accounting for taxable period ending on or prior to the Closing Date; (B) "closing agreement" as described in Section 7121 of the Code (or any corresponding or similar provision of state, local or foreign income Tax law) executed on or prior to the Closing Date; (C) intercompany transactions or any excess loss account described in Treasury Regulations under Section 1502 of the Code (or any corresponding or similar provisions of state, local or foreign income Tax law); (D) installment sale or open transaction disposition made on or prior to the Closing Date; or (E) prepaid amount received on or prior to the Closing Date.

(l) DMK has not distributed stock of another Person, or has had its stock distributed by another Person, in a transaction that was purported or intended to be governed in whole or in part by Section 355 of the Code.

(m) DMK has not entered into any transaction identified as a "listed transaction" for purposes of Treasury Regulations Sections 1.6011-4(b)(2) or 301.6111-2(b)(2) or any analogous provision of state or local Law.

2.14. Employee Benefit Plans.

(a) <u>DMK Employee Plans</u>. The DMK Disclosure Schedule contains a complete and accurate list of each material plan, program, policy, practice, contract, agreement or other arrangement providing for retirement, deferred compensation, severance, separation, termination pay, performance awards, bonus, incentive, stock option, stock purchase, stock bonus, phantom stock, stock appreciation right, restricted stock unit, supplemental retirement, fringe benefits, cafeteria benefits or other benefits, including, without limitation, each "employee benefit plan" within the meaning of Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended ("*ERISA*"), which is sponsored, maintained, contributed to, or required to be contributed to by DMK and, with respect to any such plans which are subject to Code Section 401(a), any trade or business (whether or not incorporated) that is or at any relevant time was treated as a single employer with DMK within the meaning of Section 414(b), (c), (m) or (o) of the Code, (an "*ERISA Affiliate*") for the benefit of any person who performs or who has performed services for DMK as an employee or with respect to which DMK or any ERISA Affiliate has or may have any liability (including, without limitation, contingent liability) or obligation (collectively, the "*DMK Employee Plans*").

(b) <u>Compliance</u>. Each DMK Employee Plan has been administered in material compliance with its terms and with the requirements of applicable law; and DMK and each ERISA Affiliate have performed all material obligations required to be performed by them under, and are not in any material respect in default under or violation of, any of DMK Employee Plans. No DMK Employee Plan is intended to be qualified under Section 401(a) of the Code. To DMK's Knowledge, no "prohibited transaction," within the meaning of Section 4975 of the Code or Sections 406 and 407 of ERISA, and not otherwise exempt under Section 408 of ERISA, has occurred with respect to any DMK Employee Plan.

(c) <u>Future Commitments</u>. No DMK Employee Plan provides (except at no cost to DMK), or reflects or represents any liability of DMK to provide, retiree life insurance, retiree health benefits or other retiree employee welfare benefits to any Person for any reason, except as may be required by COBRA or other applicable Law. Other than commitments made that involve no future costs to DMK, DMK has never represented, promised or contracted (whether in oral or written form) to any current or former employee of DMK or any other Person that such employee or other Person would be provided with retiree life insurance, retiree health benefit or other retiree employee welfare benefits, except to the extent required by applicable Law.

(d) <u>Effect of Transaction</u>. The consummation of the transactions contemplated by this Agreement will not (i) entitle any current or former employee or other service provider of DMK or any ERISA Affiliate to severance benefits or any other payment (including, without limitation, unemployment compensation, golden parachute, bonus or benefits under any DMK Employee Plan), except as expressly provided in this Agreement; or (ii) accelerate the time of payment or vesting of any such benefits or increase the amount of compensation due any such employee or service provider.

2.15. <u>Employee Matters</u>. The employment of each of DMK's employees is terminable by DMK at will, subject to any notice provisions of applicable employment agreements. DMK is in material compliance with all currently applicable laws and regulations respecting terms and conditions of employment. There are no proceedings pending or, to DMK's Knowledge, threatened, between DMK, on the one hand, and any or all of its current or former employees, on the other hand, which would reasonably be expected to have a Material Adverse Effect on DMK. DMK has provided all employees with all wages, benefits, relocation benefits, stock options, bonuses and incentives, and all other compensation that became due and payable through the date of this Agreement.

2.16. <u>Insurance</u>. The DMK Disclosure Schedule sets forth all policies of insurance maintained by, at the expense of or for the benefit of DMK. All premiums due and payable under all such policies have been paid and, to DMK's Knowledge, DMK is otherwise in compliance with the terms of such policies. To DMK's Knowledge, there is no threatened termination of, or material premium increase with respect to, any of such policies. No officer or director is a party to or is subject to any decree, order, judgment, or agreement that would prevent such person from serving as a director or officer of a public company under any applicable Law.

2.17. <u>Compliance with Law</u>.

(a) For all periods of time during which the respective applicable statute of limitations periods have not expired, (i) to its Knowledge, DMK has complied in all material respects with, is not in material violation of, and has not received any written or, to DMK's Knowledge, other notices of material violation with respect to, any applicable Law or regulation with respect to the conduct of its business, or the ownership or operation of its business; and (ii) DMK has not received any written notices from any Governmental Entity regarding (A) any actual, alleged, possible, or potential violation of, or failure to comply with, any applicable Law, or (B) any actual, alleged, possible, or potential obligation on the part of DMK to undertake, or to bear all or any portion of the cost of, any remedial action related to compliance or non-compliance with any applicable Law, in each of the above cases which if determined adversely to DMK would reasonably be expected to have a Material Adverse Effect on DMK.

(b) To the knowledge of DMK, DMK and Persons acting in concert with and on behalf of DMK:

(i) have not used in any capacity the services of any individual or entity debarred, excluded, or disqualified under 21 U.S.C. Section 335a, 42 U.S.C. Section 1320a-7, 21 C.F.R. Section 312.70, or any similar laws, rules or regulations; and

(ii) have not been convicted of any crime or engaged in any conduct that has resulted, or would reasonably be expected to result, in debarment, exclusion, or disqualification under 21 U.S.C. Section 335a, 42 U.S.C. Section 1320a-7, 21 C.F.R. Section 312.70, or any similar laws, rules regulations.

(c) All pre-clinical and clinical studies relating to product or product candidates, conducted by or on behalf of DMK have been, or are being, conducted in all material respects in compliance with the applicable requirements of the FDA's Good Laboratory Practice and Good Clinical Practice requirements, including regulations under 21 C.F.R. Parts 50, 54, 56, 58, 312 and applicable guidance documents, as amended from time to time, the Animal Welfare Act, and all applicable similar requirements in other jurisdictions, including all requirements relating to protection of human subjects participating in any such clinical studies; provided, however, that the foregoing representation and warranty is made only to DMK's knowledge with respect to clinical and pre-clinical studies conducted by any third party on behalf of DMK. DMK has filed with the FDA, any other Governmental Body, and any institutional review board or comparable body, all required notices, supplemental applications, and annual or other reports, including adverse experience reports, with respect to each investigational new drug application or any comparable foreign regulatory application, related to the manufacture, testing, study, or sale of any of its products or product candidates, as applicable.

(d) To DMK's Knowledge, DMK has implemented and maintains reasonable written policies and procedures, satisfying the requirements of applicable Laws, concerning the privacy security, collection and use of information about identified or identifiable individuals.

2.18. <u>Environmental Matters</u>. To the Knowledge of DMK, DMK is, and at all times has been, in compliance in all material respects with all Environmental Laws and is not subject to any material liability under any Environmental Law. DMK has not received, nor to DMK's Knowledge has any other Person for whose conduct it is or may be held responsible, received, any order, written notice, or other written communication from (a) any Governmental Entity or private citizen acting in the public interest, or (b) the current or prior owner or operator of any Facilities, asserting or alleging any actual or potential violation of or failure to comply with any Environmental Law, or any obligation to undertake or bear the cost of any Environmental, Health, and Safety Liabilities.

2.19. Legal Proceedings. There is no pending Legal Proceeding that has been commenced by or against DMK. There is no judgment, decree or order against DMK, or, to DMK's Knowledge, any of its directors or officers (in their capacities as such), that could prevent, enjoin, or materially alter or delay any of the transactions contemplated by this Agreement, or any Ancillary Agreement, or that would reasonably be expected to have a Material Adverse Effect on DMK. To DMK's Knowledge, no event has occurred, and no claim, dispute or other condition or circumstance exists, that will, or that would reasonably be expected to, give rise to or serve as a basis for the commencement of any such Legal Proceeding.

2.20. Contracts; No Defaults.

(a) The DMK Disclosure Schedule sets forth a list of all DMK Material Contracts. Each Material Contract of DMK is enforceable against DMK in accordance with its terms, subject to (i) laws of general application relating to bankruptcy, insolvency and the relief of debtors, and (ii) rules of law governing specific performance, injunctive relief and other equitable remedies. There are no DMK Material Contracts that are not in written form.

(b) DMK has not violated or breached, or committed any material default under, any Material Contract to which DMK is a party or by which its assets are bound, in each of the above cases where such violation, beach or default would have a Material Adverse Effect on DMK. Except as disclosed or reflected in the DMK Disclosure Schedule, DMK has not received any written notice or other written or, to DMK's Knowledge, oral communication regarding any actual or possible material violation or breach of, or default under, any Material Contract of DMK.

(c) The DMK Disclosure Schedule sets forth a list of all material consents or waivers of, or notifications to, any Governmental Entity or any third party that are required or provided for under any Material Contract of DMK in connection with the execution and delivery of this Agreement and the Ancillary Agreements by DMK and the consummation of the transactions contemplated hereby and thereby.

2.21. Labor Matters. DMK is not a party to, or bound by, any collective bargaining agreement, contract or other agreement or understanding with a labor union or labor organization. DMK is not the subject of any Legal Proceeding asserting that DMK has committed an unfair labor practice or seeking to compel it to bargain with any labor organization as to wages or conditions of employment. There is no strike, work stoppage or other labor dispute involving DMK pending or, to DMK's Knowledge, threatened against DMK.

2.22. <u>Regulatory Compliance</u>.

(a) Each DMK Product has been and is being manufactured, tested, distributed and/or marketed in compliance in all material respects with all applicable requirements under FDCA and similar Law, including those relating to investigational use, good manufacturing practices, labeling, advertising, record keeping, and filing of report, except where failure to be in such compliance would not have any Material Adverse Effect on DMK.

(b) No DMK Product has been recalled, withdrawn, suspended or discontinued by DMK in the United States or outside the United States (whether voluntarily or otherwise) (other than as a result of decisions made in the ordinary course of business for business or economic reasons not to pursue research or development of one or more DMK Products). There are no proceedings in the United States or outside the United States (whether completed or pending) seeking the recall, withdrawal, suspension or seizure of any DMK Product pending, nor have any such proceedings been pending at any time.

(c) As to each DMK Product for which a human biological license application, human establishment license application, human product license application, new human drug application, investigational new human drug application, abbreviated or supplemental new human drug application, or abbreviated or supplemental new animal drug application, new animal drug application, or similar state or foreign regulatory application has been approved, DMK and each licensee of any such biological or drug (a "*Product Licensee*") have not been determined to be in violation of 21 U.S.C. sec. 355, 360b, 42 U.S.C. sec. 351, and 21 U.S.C. sec. 822, and 21 C.F.R. Parts 312, 314, 511, 514, 601, and 1301 et seq.

(d) All manufacturing, warehousing, distributing, and testing operations conducted on DMK Products for human use by or for the benefit of DMK or each Product Licensee are not in violation of and have been and are being conducted in material compliance with the good manufacturing practice regulations set forth in 21 C.F.R. Parts 210 and 211 and similar applicable Law.

(e) Neither DMK, nor to the Knowledge of DMK, any officer, employee or agent of DMK, has been convicted of any crime or engaged in any conduct for which debarment is mandated by 21 U.S.C. Section 335a(a) or any similar applicable Law, or authorized by 21 U.S.C. Section 335a(b) or any similar applicable Law.

(f) There are no Legal Proceedings pending or, to the Knowledge of DMK, threatened with respect to an alleged material violation by DMK of the Federal Food, Drug, and Cosmetic Act, the Public Health Service Act, the U.S. Food and Drug Administration regulations adopted thereunder, the Controlled Substances Act or any other similar Law promulgated by any Governmental Entity responsible for regulation of the development, testing, manufacturing, processing, storage, labeling, sale, marketing, advertising, distribution and importation or exportation of drug products.

2.23. <u>Unlawful Payments</u>. To DMK's Knowledge, none of DMK, or any officer, director, employee, agent or representative of DMK has made, directly or indirectly, any bribe or kickback, illegal political contribution, payment from corporate funds which was incorrectly recorded on the books and records of DMK, or unlawful payment from corporate funds, to governmental or municipal officials in their individual capacities for the purpose of affecting their action or the actions of the jurisdiction which they represent to obtain favorable treatment in securing business or licenses or to obtain special concessions of any kind whatsoever, or illegal payment from corporate funds to obtain or retain any business.

2.24. <u>Financial Advisor</u>. No broker, finder or investment banker is entitled to any commission or brokerage or finder's fee in connection with the Merger or any of the other transactions contemplated by this Agreement based upon arrangements made by or on behalf of DMK.

2.25. Takeover Statutes; No Rights Plan; Appraisal Rights. DMK is not subject to the prohibition on certain business combinations set forth in Section 14A:10A-4, -5 and -6 of the NJBCA and, to DMK's Knowledge, any similar "moratorium," "control share," "fair price," "takeover" or "interested shareholder" Law (any such Laws, a "*Takeover Statute*") applicable to DMK. There is no shareholder rights plan, "poison pill", antitakeover plan or other similar agreement or plan in effect to which DMK is a party or is otherwise bound.

2.26. <u>No Other Representations</u>. Except for the representations and warranties contained in this Article II (including the related portions of the DMK Disclosure Schedule), neither DMK nor any other Person has made or makes any other express or implied representation or warranty, either written or oral, on behalf of DMK. Without limiting the generality of the foregoing, neither DMK nor any other Person has made or makes any representation or warranty with respect to any projections, estimates, or budgets of future revenues, future results of operations, future cash flows, or future financial condition (or any component of any of the foregoing) of DMK, including any information made available in the electronic data room maintained by DMK for purposes of the transactions contemplated by this Agreement, teasers, marketing materials, consulting reports or materials, confidential information memoranda, management presentations, functional "break-out" discussions, responses to questions submitted on behalf of Adamis, Merger Sub, or any of their respective Representatives, or in any other form in connection with the transactions contemplated by this Agreement.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF ADAMIS AND MERGER SUB

Adamis, and Merger Sub (with respect to the representations, warranties and covenants of Merger Sub), represent and warrant to DMK that the statements contained in this Article III are true and correct as set forth herein and as qualified by the disclosure schedules separately delivered to DMK concurrently herewith (collectively, the "*Adamis Disclosure Schedule*"). The disclosures set forth in Adamis Disclosure Schedule shall be arranged in paragraphs corresponding to the numbered and lettered paragraphs contained in this Article III. The disclosures in any section or subsection of the Adamis Disclosure Schedule shall qualify other sections and subsections in this Article III to the extent it is reasonably apparent from a reading of the disclosure that such disclosure is applicable to such other sections and subsections.

3.1. Organization and Qualification; Charter Documents.

(a) Each of Adamis and Merger Sub is a corporation duly organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation, with requisite corporate power and authority to conduct its business as now being conducted and to own or use its properties and assets. Adamis is duly qualified to do business as a foreign corporation and is in good standing under the laws of each state or other jurisdiction in which either the ownership or use of the properties owned or used by it, or the nature of the activities conducted by it, requires such qualification, except where the failure to be so qualified or in good standing would not have a Material Adverse Effect on Adamis.

(b) The copies of the restated certificate of incorporation and bylaws of Adamis which are incorporated by reference as exhibits to the Adamis Annual Report on Form 10-K for the year ended December 31, 2021, together with any subsequent amendments thereto or certificates of designation of preferences, rights and limitations that have been filed as exhibits to the Adamis SEC Reports, are complete and correct copies of such documents.

3.2. Subsidiaries. The Adamis SEC Reports set forth all direct and indirect significant Subsidiaries of Adamis. Adamis owns all of the equity of each such Subsidiary. Except as set forth in the Adamis SEC Reports, Adamis does not have any significant Subsidiaries and, other than Merger Sub, does not otherwise own any shares in the capital of or any interest in, or control, directly or indirectly, any Entity. Merger Sub and each significant Subsidiary of Adamis: (a) is a corporation duly organized and validly existing under the laws of its jurisdiction of incorporation, (b) has all requisite corporate power and authority to own, operate or lease the properties and assets owned, operated or leased by such Subsidiary and to carry on its business as it has been and is currently conducted by such Subsidiary, and (c) is duly qualified to do business and is in good standing in each jurisdiction in which the properties owned or leased by it or the operation of its business makes such license and qualification necessary, except, in each of clauses (a), (b) and (c), for such failures which, when taken together with all other such failures, would not have a Material Adverse Effect on Adamis and its Subsidiaries, when considered together.

3.3. Authority. Each of Adamis and Merger Sub has all requisite corporate power and authority to enter into this Agreement and the Ancillary Agreements to which it is a party, to perform its obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby. The execution and delivery of this Agreement and the Ancillary Agreements and the consummation of the transactions contemplated hereby have been duly authorized by all necessary corporate action on the part of Adamis and Merger Sub, subject only to the approval of this Agreement by the stockholders of Adamis and Merger Sub. The Board of Directors of Adamis and Merger Sub have unanimously approved this Agreement and the Merger. This Agreement has been (and the Ancillary Agreements will be at the Closing) duly executed and delivered by Adamis and Merger Sub, and this Agreement constitutes (and the Ancillary Agreements will constitute at the Closing) the valid and binding obligations of Adamis and Merger Sub in accordance with their terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting or relating to creditors' rights generally, and subject to general principles of equity. Merger Sub has been formed solely for the purpose of executing and delivering this Agreement and the related documents and instruments. Merger Sub has no assets or properties or debts, liabilities or obligations of any kind whatsoever, and with the exception of this Agreement and the related documents and instruments, is not a party to any contract, agreement or undertaking of any nature.

3.4. <u>No Conflict</u>. The execution and delivery by Adamis of this Agreement and the Ancillary Agreements to which Adamis is a party, does not, and the consummation of the transactions contemplated hereby and thereby will not, (a) conflict with, or result in any material violation of, any provision of the Adamis Restated Certificate, as it may be amended immediately before the Effective Time, or the Adamis Bylaws, (b) result in any material violation of or default under (with or without notice or lapse of time, or both), or give rise to a right of termination, cancellation or acceleration of any obligation or loss of any benefit under, any material mortgage, indenture, lease, contract, grant, funding arrangement, or other agreement or instrument, permit, concession, franchise or license of Adamis, (c) subject to obtaining the approval of Adamis' stockholders and, except as would not reasonably be expected to have a Material Adverse Effect on Adamis, conflict with, or result in any violation of any judgment, order, decree, statute, law, ordinance, rule or regulation applicable to Adamis or any of its properties or assets, or (d) conflict with, or result in a violation of any resolution adopted by Adamis' stockholders, Adamis' board of directors or any committee of Adamis' board of directors.

3.5. <u>Non-Contravention; Consents</u>.

(a) Subject to compliance with obtaining the required approvals of the stockholders of Adamis and the filing of the Certificate of Merger with the Secretary of State of Delaware pursuant to the DGCL and the Secretary of State of New Jersey pursuant to the NJBCA, neither (x) the execution, delivery or performance of this Agreement by Adamis, nor (y) the consummation of the Contemplated Transactions, will directly or indirectly (with or without notice or lapse of time):

(i) contravene, conflict with or result in a material violation of, or give any Governmental Entity or other Person the right to challenge the Contemplated Transactions or to exercise any material remedy or obtain any material relief under, any Law or any Order by which Adamis, or any of the material assets owned or used by Adamis, is subject;

(ii) contravene, conflict with or result in a material violation of any of the terms or requirements of, or give any Governmental Entity the right to revoke, withdraw, suspend, cancel, terminate or modify, any Governmental Authorization that is held by and material to Adamis;

(iii) contravene, conflict with or result in a violation or breach of, or result in a default under, any provision of any Adamis Material Contract, or give any Person the right to: (A) declare a default or exercise any remedy under any Adamis Material Contract, (B) any material payment, rebate, chargeback, penalty or change in delivery schedule under any Adamis Material Contract, (C) accelerate the maturity or performance of any Adamis Material Contract, or (D) cancel, terminate or modify any term of any Adamis Material Contract, except in each of the above cases as would not reasonably be expected to have a Material Adverse Effect on Adamis; or

(iv) result in the imposition or creation of any material Encumbrance upon or with respect to any asset owned or used by Adamis (except for Permitted Encumbrances) that is material to its business.

(b) Except for (i) the Required Adamis Stockholder Vote, (ii) the filing of the Certificate of Merger with the Secretary of State of Delaware pursuant to the DGCL and the Secretary of State of the State of New Jersey pursuant to the NJBCA, (iii) such filings as may be required under applicable securities laws, and (iv) such consents, waivers, approvals, orders, authorizations, registrations, declarations and filings which, if not obtained or made, would not have a Material Adverse Effect on Adamis, Adamis is not and will not be required to make any filing with or give any notice to, or to obtain any waiver or Consent from, any Governmental Entity or Person in connection with (x) the execution, delivery or performance of this Agreement, or (y) the consummation of the Contemplated Transactions.

3.6. <u>Governmental Authorizations</u>.

(a) Adamis and its Subsidiaries have obtained each material federal, state, county, local or foreign governmental consent, license, permit, grant, or other authorization of a Governmental Entity (i) pursuant to which Adamis or its Subsidiaries currently operates or holds any interest in any of its properties, or (ii) that is required for the operation of Adamis and its Subsidiaries' business or the holding of any such interest, and all of such authorizations are in full force and effect, except for such consents, licenses, permits, grants or other authorizations, which if not obtained would not have a Material Adverse Effect on Adamis.

(b) Neither Adamis, its Subsidiaries nor any of its or their officers or employees are: (i) a party to any material Action involving Adamis' activities in the healthcare industry; (ii) subject to any Actions or sanctions by any payor (including Medicare or Medicaid); and (iii) listed by a federal agency as excluded, disbarred, suspended, or otherwise ineligible to participate in federal programs, including Medicare or Medicaid, or listed on the General Service Administration list of parties excluded from Federal Procurement and Non-Procurement Programs.

3.7. <u>Capitalization</u>.

(a) The authorized capital stock of Adamis consists of 200,000,000 shares of Common Stock, \$.0001 par value, and 10,000,000 shares of Preferred Stock, \$.0001 par value, of which there were issued and outstanding as of September 30, 2022, 149,983,265 shares of Common Stock and 3,000 shares of Series C Convertible Preferred Stock. The shares of Adamis Common Stock to be issued pursuant to the Merger will be duly authorized, validly issued, fully paid, and non-assessable, free of any liens or encumbrances other than any liens or encumbrances created by or imposed by the holders thereof, and shall be issued in material compliance with all applicable federal and state securities laws. All of the outstanding shares of Adamis Common Stock (i) have been duly authorized and validly issued, and are fully paid and non-assessable, (ii) are not subject to preemptive rights or rights of first refusal created by statute, Adamis Restated Certificate, Adamis Bylaws or any agreement to which Adamis is party or by which it is bound, and (iii) have been issued in compliance in all material respects with federal and state securities laws. There are no declared or unpaid dividends with respect to any shares of Adamis Common Stock.

(b) The information in the Adamis SEC Reports concerning the Adamis Stock Plans, and outstanding options and warrants to purchase shares of Adamis Common Stock, and outstanding restricted stock units or other rights to acquire shares of Adamis Common Stock, is accurate in all material respects as of the dates of such Adamis SEC Reports.

3.8. <u>Title to Assets</u>.

(a) Except as disclosed or reflected in the Adamis SEC Reports, Adamis owns, and has good, valid and marketable title to, all material assets purported to be owned by it, including all assets reflected on its balance sheet as of September 30, 2022; and all other material assets reflected in Adamis' books and records as being owned by Adamis. All of said assets are owned by Adamis free and clear of any liens or other Encumbrances, except for (x) any lien for current taxes not yet due and payable, and (y) minor liens that have arisen in the ordinary course of business and that do not (in any case or in the aggregate) materially detract from the value of the assets subject thereto or materially impair the operations of Adamis.

(b) Except as disclosed or reflected in the Adamis SEC Reports, Adamis does not own any real property and Adamis is not party to any material lease for real property either as a lessee or lessor.

3.9. SEC Reports; Financial Statements; Listing and Maintenance Requirements.

(a) As of their respective filing dates, all annual, quarterly or current reports, filed by Adamis with the SEC since January 1, 2022 (including those that Adamis may file subsequent to the date hereof) (such reports, as amended "*Adamis SEC Reports*"), (i) were prepared in accordance in all material respects with the requirements of the Securities Act and the Exchange Act, as the case may be, and the rules and regulations thereunder, except as may be reflected in any amendments to such reports that Adamis has filed with the SEC, and (ii) as the same may have been amended, did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading.

(b) The consolidated financial statements (including any related notes thereto) contained in the Adamis SEC Reports (in the form, as applicable, in any amendments to such Adamis SEC Reports) (the "*Adamis Financial Statements*"): (i) complied as to form in all material respects with the published rules and regulations of the SEC applicable thereto; (ii) were prepared in accordance with GAAP applied on a consistent basis throughout the periods covered, except as may be indicated in the notes to such financial statements and (in the case of unaudited statements) as permitted by Form 10-Q of the SEC, and except that unaudited financial statements may not contain footnotes and are subject to year-end audit adjustments; and (iii) fairly present in all material respects the consolidated financial position of Adamis as of the respective dates thereof and the consolidated results of operations and cash flows of Adamis for the periods covered thereby.

(c) Adamis maintains a system of internal accounting controls and disclosure controls and procedures sufficient, in the judgment of Adamis' board of directors, to provide reasonable assurance that (i) transactions are executed in accordance with management's general or specific authorizations, (ii) transactions are recorded as necessary to permit preparation of financial statements in conformity with GAAP and to maintain asset accountability, (iii) access to assets is permitted only in accordance with management's general or specific authorization, and (iv) the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any differences.

3.10. <u>Absence of Certain Changes</u>. Since September 30, 2022, and except as set forth in the Adamis Disclosure Schedule or as disclosed or reflected in the Adamis SEC Reports (and except for matters relating to the execution and performance of this Agreement and the discussions, negotiations and transactions related thereto), there has not occurred any change, event or condition (whether or not covered by insurance) that has resulted in, or would reasonably be expected to result in, a Material Adverse Effect on Adamis.

3.11. <u>Interested Party Transactions</u>. Except as disclosed or reflected in the Adamis SEC Reports, Adamis is not indebted to any director, officer or employee of Adamis (except for amounts due as normal compensation and bonuses and in reimbursement of ordinary expenses), and no such person is indebted to Adamis, for amounts that would be required to be disclosed in the Adamis SEC Reports. Except as disclosed or reflected in the Adamis SEC Reports, Adamis is not a party to any transaction involving more than \$120,000 in which any director, officer or stockholder known to Adamis to be a greater than 5% stockholder of Adamis (or a member of such person's immediate family) had a direct or indirect material interest, except for transactions not required to be disclosed in response to Item 404(a) of Regulation S-K, or where such person's interest arises solely from his or her ownership of Adamis Capital Stock.

3.12. Intellectual Property.

(a) Except as disclosed or reflected in the Adamis SEC Reports, Adamis owns or possesses the right to use the Intellectual Property utilized in connection with the conduct of its business and intended business, all of which is owned by or licensed to Adamis (the "*Adamis Patent and Proprietary Rights*"), except where the failure to own or possess such rights would not have a Material Adverse Effect on Adamis. Such Adamis Patent and Proprietary Rights are sufficient in all material respects for the conduct of Adamis' business and proposed business. Adamis has taken all reasonable measures to protect and maintain the confidentiality of the trade secrets included in the Adamis Patent and Proprietary Rights.

(b) Since January 1, 2021, except as described or reflected in the Adamis SEC Reports or as set forth in the Adamis Disclosure Schedule, Adamis has not received any notice of any asserted rights with respect to any of Adamis Patent and Proprietary Rights which, if determined unfavorably with respect to the interests of Adamis, would have a Material Adverse Effect on Adamis. Except as described or reflected in the SEC Reports, Adamis is not bound by, and no Adamis Patent and Proprietary Rights are subject to, any Contract containing any covenant or other provision that in any material respect limits or restricts the ability of Adamis to use, exploit, assert, defend, or enforce any Adamis Patent and Proprietary Rights.

(c) To Adamis' Knowledge, Adamis has never infringed (directly, contributorily, by inducement, or otherwise), misappropriated, or otherwise violated or made unlawful use of any right to Intellectual Property of any other Person or engaged in unfair competition, which infringement, misappropriation, violation or use (if the subject of any unfavorable decision, ruling or finding), individually or in the aggregate, would result in a Material Adverse Effect on Adamis. No material infringement, misappropriation, or similar claim or Legal Proceeding is pending or, to Adamis' Knowledge, threatened against Adamis or any other Person who is or may be entitled to be indemnified, defended, held harmless, or reimbursed by Adamis with respect to such claim or Legal Proceeding, which claim or Legal Proceeding (if the subject of any unfavorable decision, ruling or finding), individually or in the aggregate, would result in a Material Adverse Effect on Adamis.

(d) To Adamis' Knowledge, Adamis has not engaged in patent or copyright misuse or any fraud or inequitable conduct in connection with any Adamis Patent and Proprietary Rights, and no trademark or trade name owned, used, or applied for by Adamis conflicts or interferes in any material respect with any trademark or trade name owned, used, or applied for by any other Person.

3.13. <u>Taxes</u>.

(a) Adamis has prepared and timely filed all material Tax Returns relating to any and all material Taxes concerning or attributable to Adamis and such Tax Returns are true and correct in all material respects and have been completed in accordance with applicable law in all material respects. Adamis has delivered or made available to DMK correct and complete copies of all federal income Tax Returns, examination reports, and statements of deficiencies assessed against or agreed to by Adamis filed or received since January 1, 2019.

(b) Adamis (i) is not delinquent in any material respects in the payment of any Taxes due and owing by Adamis, and (ii) has withheld and timely paid all material Taxes required to have been withheld and paid with respect to any amounts paid or owing to any employee, independent contractor, creditor, stockholder, or other third party.

(c) There is no material Tax deficiency outstanding or assessed or, to Adamis' Knowledge, proposed against Adamis that is not reflected as a liability on the Adamis Financial Statements, nor has Adamis executed any agreements or waivers extending any statute of limitations on or extending the period for the assessment or collection of any Tax (other than extensions which have expired). No claim (other than claims that have been resolved) has ever been made by an authority in a jurisdiction where Adamis does not file Tax Returns that it is or may be subject to taxation by that jurisdiction. There are no liens for Taxes (other than Taxes not yet due and payable) upon any of the assets of Adamis.

(d) To Adamis' Knowledge, Adamis has no material liabilities for unpaid Taxes that have not been accrued for or reserved on the Adamis Financial Statements, whether asserted or unasserted, contingent or otherwise.

(e) Adamis has not received from any Governmental Entity any (i) notice indicating an intent to open an audit or other review, (ii) request for information related to Tax matters, or (iii) notice of deficiency or proposed adjustment of or any amount of Tax proposed, asserted, or assessed by any Governmental Entity against Adamis, in each case other than such as have been resolved before the date of this Agreement.

(f) Adamis is not a party to any tax-sharing agreement or similar arrangement with any other party, and Adamis has not assumed any obligation to pay any Tax obligations of, or with respect to any transaction relating to, any other person or agreed to indemnify any other person with respect to any Tax.

(g) Adamis has not been a member of an affiliated group of corporations filing a consolidated federal income tax return other than a group of which Adamis was the parent.

(h) Adamis has not been at any time a United States Real Property Holding Corporation within the meaning of Section 897(c)(2) of the Code during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code.

(i) Adamis is not aware of any facts, and has not knowingly taken or agreed to take or refrain from taking any action, in each case, that would reasonably be expected to prevent or impede the Merger from qualifying as a "reorganization" within the meaning of Section 368(a) of the Code.

(j) Adamis has not filed a consent under Section 341(f) of the Code concerning collapsible corporations. Except as disclosed or reflected in the Adamis SEC Reports, is not a party to any contract, agreement, plan or arrangement, including but not limited to the provisions of this Agreement, covering any employee or former employee of Adamis that, individually or collectively, could give rise to the payment of (i) any "excess parachute payment" within the meaning of Section 280G of the Code (or any corresponding provisions of state, local or foreign Tax law) and (ii) any amount that will not be fully deductible as a result of Section 162(m) of the Code (or any corresponding provisions of state, local or foreign Tax law).

(k) Adamis will not be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion there) ending after the Closing Date as a result of any: (A) change in method of accounting for taxable period ending on or prior to the Closing Date; (B) "closing agreement" as described in Section 7121 of the Code (or any corresponding or similar provision of state, local or foreign income Tax law) executed on or prior to the Closing Date; (C) intercompany transactions or any excess loss account described in Treasury Regulations under Section 1502 of the Code (or any corresponding or similar provisions of state, local or foreign income Tax law); (D) installment sale or open transaction disposition made on or prior to the Closing Date; or (E) prepaid amount received on or prior to the Closing Date.

(l) Adamis has not distributed stock of another Person, or has had its stock distributed by another Person, in a transaction that was purported or intended to be governed in whole or in part by Section 355 of the Code.

(m) Adamis has not entered into any transaction identified as a "listed transaction" for purposes of Treasury Regulations Sections 1.6011-4(b)(2) or 301.6111-2(b)(2) or any analogous provision of state or local Law.

3.14. Employee Benefit Plans.

(a) <u>Employee Plans</u>. The Adamis SEC Reports describe or reflect, or Adamis has otherwise made available to DMK descriptions or copies of, each Adamis Employee Agreement and each material plan, program, policy, practice, contract, agreement or other arrangement providing for retirement, deferred compensation, severance, separation, termination pay, performance awards, bonus, incentive, stock option, stock purchase, stock bonus, phantom stock, stock appreciation right, supplemental retirement, fringe benefits, cafeteria benefits or other benefits, without limitation, each "employee benefit plan" within the meaning of Section 3(3) of ERISA, which is sponsored, maintained, contributed to, or required to be contributed to by Adamis and, with respect to any such plans which are subject to Code Section 401(a), any ERISA Affiliate for the benefit of any person who performs or who has performed services for Adamis as an employee or with respect to which Adamis or any ERISA Affiliate has or may have any liability (including, without limitation, contingent liability) or obligation (collectively, the "Adamis Employee Plans").

(b) <u>Compliance</u>. Each Adamis Employee Plan has been administered in material compliance with its terms and with the requirements of applicable law; and Adamis and each ERISA Affiliate have performed all material obligations required to be performed by them under, and are not in any material respect in default under or violation of, any of the Adamis Employee Plans.

(c) <u>Future Commitments</u>. Except as disclosed or reflected in the Adamis SEC Reports, the Adamis Employee Agreements or in the Adamis Proxy Statement, no Adamis Employee Plan provides (except at no cost to Adamis), or reflects or represents any liability of Adamis to provide, retiree life insurance, retiree health benefits or other retiree employee welfare benefits to any Person for any reason, except as may be required by COBRA or other applicable Law. Except as disclosed or reflected in the Adamis SEC Reports or in the Adamis Proxy Statement, other than commitments made that involve no future costs to Adamis, Adamis has never represented, promised or contracted (whether in oral or written form) to any current or former employee of Adamis or any other Person that such employee or other Person would be provided with retiree life insurance, retiree health benefit or other retiree employee welfare benefits, except to the extent required by applicable Law.

(d) <u>Effect of Transaction</u>. Except as disclosed in the SEC Reports, the Adamis Proxy Statement or in the Adamis Disclosure Schedule, the consumation of the transactions contemplated by this Agreement will not (i) entitle any current or former employee or other service provider of Adamis or any ERISA Affiliate to severance benefits or any other payment (including, without limitation, unemployment compensation, golden parachute, bonus or benefits under any Adamis Employee Plan), except as expressly provided in this Agreement; or (ii) accelerate the time of payment or vesting of any such benefits or increase the amount of compensation due any such employee or service provider.

3.15. <u>Employee Matters</u>. The employment of each of Adamis' employees is terminable by Adamis at will, subject to any notice provisions of applicable employment agreements. Adamis is in material compliance with all currently applicable laws and regulations respecting terms and conditions of employment. There are no material proceedings pending or, to Adamis' Knowledge, threatened, between Adamis, on the one hand, and any or all of its current or former employees, on the other hand, which would reasonably be expected to have a Material Adverse Effect on Adamis. Adamis has provided all employees with all material wages, benefits, relocation benefits, stock options, bonuses and incentives, and all other compensation that became due and payable through the date of this Agreement.

3.16. <u>Insurance</u>. The Adamis Disclosure Schedule sets forth all policies of insurance maintained by, at the expense of or for the benefit of Adamis. All premiums due and payable under all such policies have been paid and, to Adamis' Knowledge, Adamis is otherwise in compliance with the terms of such policies.

3.17. <u>Compliance with Law</u>.

(a) For all periods of time during which the respective applicable statute of limitations periods have not expired, and except as disclosed or reflected in the Adamis SEC Reports or the Adamis Disclosure Schedule, (i) to its Knowledge, Adamis has complied in all material respects with, is not in material violation of, and has not received any written or, to Adamis' Knowledge, other notices of material violation with respect to, any applicable Law or regulation with respect to the conduct of its business, or the ownership or operation of its business, and (ii) Adamis has not received any written notices from any Governmental Entity regarding (A) any actual, alleged, possible, or potential violation of, or failure to comply with, any applicable Law, or (B) any actual, alleged, possible, or potential obligation on the part of Adamis to undertake, or to bear all or any portion of the cost of, any remedial action related to compliance or non-compliance with any applicable Law, in each of the above cases which if determined adversely to Adamis would reasonably be expected to have a Material Adverse Effect on Adamis.

(b) To the knowledge of Adamis, Adamis and Persons acting in concert with and on behalf of Adamis:

(i) have not used in any capacity the services of any individual or entity debarred, excluded, or disqualified under 21 U.S.C. Section 335a, 42 U.S.C. Section 1320a-7, 21 C.F.R. Section 312.70, or any similar laws, rules or regulations; and

(ii) have not been convicted of any crime or engaged in any conduct that has resulted, or would reasonably be expected to result, in debarment, exclusion, or disqualification under 21 U.S.C. Section 335a, 42 U.S.C. Section 1320a-7, 21 C.F.R. Section 312.70, or any similar laws, rules regulations.

(c) All pre-clinical and clinical studies relating to product or product candidates, conducted by or on behalf of Adamis have been, or are being, conducted in all material respects in compliance with the applicable requirements of the FDA's Good Laboratory Practice and Good Clinical Practice requirements, including regulations under 21 C.F.R. Parts 50, 54, 56, 58, 312 and applicable guidance documents, as amended from time to time, the Animal Welfare Act, and all applicable similar requirements in other jurisdictions, including all requirements relating to protection of human subjects participating in any such clinical studies; provided, however, that the foregoing representation and warranty is made only to Adamis' knowledge with respect to clinical and pre-clinical studies conducted by any third party on behalf of Adamis. Adamis has filed with the FDA, any other Governmental Body, and any institutional review board or comparable body, all required notices, supplemental applications, and annual or other reports, including adverse experience reports, with respect to each investigational new drug application or any comparable foreign regulatory application, related to the manufacture, testing, study, or sale of any of its products or product candidates, as applicable.

(d) To Adamis' Knowledge, Adamis has implemented and maintains reasonable written policies and procedures, satisfying the requirements of applicable Laws, concerning the privacy security, collection and use of information about identified or identifiable individuals.

3.18. <u>Environmental Matters</u>. To Adamis' Knowledge, Adamis is, and at all times has been, in compliance in all material respects with all Environmental Laws and is not subject to any material liability under any Environmental Law. Adamis has not received, nor to Adamis' Knowledge has any other Person for whose conduct it is or may be held responsible, received, any order, written notice, or other written communication from (i) any Governmental Entity or private citizen acting in the public interest, or (ii) the current or prior owner or operator of any Facilities, asserting or alleging any actual or potential violation of or failure to comply with any Environmental Law, or any obligation to undertake or bear the cost of any Environmental, Health, and Safety Liabilities.

3.19. Legal Proceedings. Except as disclosed in the Adamis SEC Reports, there is no pending Legal Proceeding that has been commenced by or against Adamis that is required to be disclosed in such SEC Reports. There is no judgment, decree or order against Adamis or, to Adamis' Knowledge, any of its directors or officers (in their capacities as such), that could prevent, enjoin, or materially alter or delay any of the transactions contemplated by this Agreement, or any Ancillary Agreement, or that would reasonably be expected to have a Material Adverse Effect on Adamis.

3.20. <u>Contracts; No Defaults</u>.

(a) Each Adamis Material Contract is enforceable against Adamis in accordance with its terms, subject to (i) laws of general application relating to bankruptcy, insolvency and the relief of debtors, and (ii) rules of law governing specific performance, injunctive relief and other equitable remedies.

(b) Adamis has not violated or breached, or committed any material default under, any Adamis Material Contract, in each of the above cases where such violation, beach or default would have a Material Adverse Effect on Adamis. Except as disclosed or reflected in the Adamis SEC Reports, Adamis has not received any notice or other written or, to Adamis' Knowledge, oral communication regarding any actual or possible material violation or breach of, or default under, any Material Contract of Adamis that was required to be disclosed in the Adamis SEC Reports.

(c) The Adamis Disclosure Schedule sets forth a list of all material consents or waivers of, or notifications to, any Governmental Entity or any third party that are required under any Adamis Material Contract in connection with the execution and delivery of this Agreement and the Ancillary Agreements by Adamis and the consummation of the transactions contemplated hereby and thereby.

3.21. Labor Matters. Adamis is not a party to, or bound by, any collective bargaining agreement, contract or other agreement or understanding with a labor union or labor organization. Adamis is not the subject of any material Legal Proceeding asserting that Adamis has committed an unfair labor practice or seeking to compel it to bargain with any labor organization as to wages or conditions of employment. There is no strike, work stoppage or other labor dispute involving Adamis pending or, to Adamis' Knowledge, threatened against Adamis.

3.22. <u>Regulatory Compliance</u>.

(a) Except as disclosed or reflected in the Adamis SEC Reports, each current Adamis Product has been and is being manufactured, tested, distributed and/or marketed in compliance in all material respects with all applicable requirements under FDCA and similar Law, including those relating to investigational use, good manufacturing practices, labeling, advertising, record keeping, and filing of report, except where failure to be in such compliance would not have any Material Adverse Effect on Adamis.

(b) Except as disclosed or reflected in the Adamis SEC Reports, no Adamis Product has been recalled, withdrawn, suspended or discontinued by Adamis in the United States or outside the United States (whether voluntarily or otherwise) (other than as a result of decisions made in the ordinary course of business for business or economic reasons not to pursue research or development of one or more Adamis Products). Except as disclosed or reflected in the Adamis SEC Reports, there are no proceedings in the United States or outside the United States or outside the United States or outside the United States (whether completed or pending) seeking the recall, withdrawal, suspension or seizure of any Adamis Product pending.

(c) As to each Adamis Product for which a human biological license application, human establishment license application, human product license application, new human drug application, investigational new human drug application, abbreviated or supplemental new human drug application, new animal drug application, or similar state or foreign regulatory application has been approved, Adamis and each licensee of any such biological or drug (a *"Product Licensee"*) have not been determined to be in violation of 21 U.S.C. sec. 355, 360b, 42 U.S.C. sec. 351, and 21 U.S.C. sec. 822, and 21 C.F.R. Parts 312, 314, 511, 514, 601, and 1301 et seq.

(d) Except as disclosed or reflected in the Adamis SEC Reports, all manufacturing, warehousing, distributing, and testing operations conducted on Adamis Products for human use by or for the benefit of Adamis or each Product Licensee are not in violation of and have been and are being conducted in material compliance with the good manufacturing practice regulations set forth in 21 C.F.R. Parts 210 and 211 and similar applicable Law.

(e) Neither Adamis, nor to the Knowledge of Adamis, any officer, employee or agent of Adamis, has been convicted of any crime or engaged in any conduct for which debarment is mandated by 21 U.S.C. Section 335a(a) or any similar applicable Law, or authorized by 21 U.S.C. Section 335a(b) or any similar applicable Law.

3.23. <u>Unlawful Payments</u>. To Adamis' Knowledge, except as disclosed or reflected in the SEC Reports, none of Adamis, or any officer, director, employee, agent or representative of Adamis has made, directly or indirectly, any bribe or kickback, illegal political contribution, payment from corporate funds which was incorrectly recorded on the books and records of Adamis, or unlawful payment from corporate funds, to governmental or municipal officials in their individual capacities for the purpose of affecting their action or the actions of the jurisdiction which they represent to obtain favorable treatment in securing business or licenses or to obtain special concessions of any kind whatsoever, or illegal payment from corporate funds to obtain or retain any business.

3.24. <u>Representations Complete</u>. This Agreement (as limited and qualified by the Adamis Disclosure Schedule) does not contain any representation, warranty or information that (i) contains an untrue statement of a material fact, or (ii) omits to state any material fact necessary in order to make the statements herein (in the light of the circumstances under which such statements have been made) not misleading.

3.25. <u>**Financial Advisor**</u>. Except as disclosed in the Adamis SEC Reports or otherwise disclosed to DMK, no broker, finder or investment banker is entitled to any commission or brokerage or finder's fee in connection with the Merger or any of the other transactions contemplated by this Agreement based upon arrangements made by or on behalf of Adamis.

3.26. Disclaimer of Reliance. Notwithstanding anything contained in this Agreement to the contrary, Adamis and Merger Sub acknowledge and agree that neither DMK nor any other Person has made or is making, and Adamis and Merger Sub expressly disclaim reliance upon, any representations, warranties, or statements relating to DMK whatsoever, express or implied, beyond those expressly given by DMK in Article II, including any implied representation or warranty as to the accuracy or completeness of any information regarding DMK furnished or made available to Adamis, Merger Sub, or any of their respective Representatives. Without limiting the generality of the foregoing, Adamis and Merger Sub acknowledge that no representations or warranties are made with respect to any projections, forecasts, estimates, budgets, or prospects information that may have been made available to Adamis, Merger Sub, or any of their respective Representatives (including in certain "data rooms," "electronic data rooms," management presentations, or in any other form in expectation of, or in connection with, the Merger or the other transactions contemplated by this Agreement).

ARTICLE IV CONDUCT BEFORE THE EFFECTIVE TIME

4.1. Access and Investigation. Subject to the terms of the Confidentiality Agreement which the Parties agree will continue in full force following the date of this Agreement, during the period commencing on the date of this Agreement and ending at the Effective Time, unless this Agreement is earlier terminated pursuant to the terms hereof (the "*Pre-Closing Period*"), upon reasonable written notice each Party shall, and shall use commercially reasonable efforts to cause such Party's Representatives to: (a) provide the other Party and such other Party's Representatives with reasonable access during normal business hours to such Party's Representatives, personnel and assets and to all existing books, records, Tax Returns, work papers and other documents and information relating to such Party; (b) provide the other Party and such other Party's Representatives of the existing books, records, Tax Returns, work papers, product data, and other documents and information relating to such Party (b) provide the other Party may reasonably request; and (c) permit the other Party's officers and other employees to meet, upon reasonable notice and during normal business hours, with the chief financial officer and other officers and managers of such Party responsible for such Party's financial statements and the internal controls of such Party to discuss such matters as the other Party may restrict the foregoing access to the extent that any Law applicable to such party requires such Party to restrict or prohibit access to any such properties or information or if such restriction is needed to protect attorney-client privilege or other applicable privilege. No information or knowledge obtained in any investigation pursuant to this Section shall affect or be deemed to modify any representation or warranty contained herein or the conditions to the obligations of the Parties to consummate the Merger.

4.2. **Operation of Adamis' Business**.

Except as contemplated or permitted by this Agreement, or as disclosed or reflected in the Adamis SEC Reports, or with the (a) prior written consent of DMK, during the Pre-Closing Period, Adamis shall: (i) use its commercially reasonable efforts to conduct its business and operations in the Ordinary Course of Business in compliance in material respects with all applicable Law and the requirements of all Contracts that constitute Material Contracts of Adamis; (ii) use its commercially reasonable efforts to preserve intact its current business organization, use commercially reasonable efforts to keep available the services of its current employees, officers and consultants and maintain its relations and goodwill with all material suppliers, customers, landlords, creditors, licensors, licensees, employees and other Persons having material business relationships with Adamis; (iii) not acquire, lease or license any material right or other material asset from any other Person; (iv) not change any of its methods of accounting or accounting practices in any material respect, except as may be required by GAAP; (v) not make any Tax election that would be material to Adamis; or (vi) not agree or commit to take any of the actions described in clauses (iii) through (v) above. Notwithstanding the foregoing, without the prior written consent of DMK, Adamis may (i) continue to have discussions with third parties, and may enter into agreements or arrangements and may consummate one or more transactions regarding, its assets, business or securities as long as such agreement, arrangements or transactions do not constitute an Alternative Transaction. In addition, for purposes of clarification and without limitation, during the Pre-Closing Period Adamis may have discussions regarding, and enter into agreements or consummate transactions regarding or with respect to: (i) sales or other dispositions of assets of Adamis relating to the compounding pharmaceuticals business formerly conducted by Adamis or its Subsidiaries and (ii) commercial, license or other agreements relating to the sale, license or distribution of one or more of Adamis' FDA approved commercial drug products.

(b) Adamis shall promptly notify DMK of: (i) any notice or other communication from any Person alleging that the Consent of such Person is or may be required in connection with any of the Contemplated Transactions; (ii) any material notice or other communication that Adamis receives from any Governmental Body in connection with the Merger or other transactions contemplated by this Agreement; (iii) any material litigation relating to or involving or otherwise affecting DMK or Adamis that relates to the Merger or other transactions contemplated by this Agreement; and (iv) any event of which it becomes aware that would be considered reasonably likely to have a Material Adverse Effect on Adamis or the Surviving Corporation.

4.3. **Operation of DMK's Business**.

(a) Except as contemplated by this Agreement, during the Pre-Closing Period, DMK shall: (i) use commercially reasonable efforts to conduct its business and operations in compliance with all applicable Law and the requirements of all Contracts that constitute Material Contracts of DMK; and (ii) use its commercially reasonable efforts to preserve intact its current business organization, use commercially reasonable efforts to keep available the services of its current Key Employees, officers and other employees and maintain its relations and goodwill with all suppliers, customers, landlords, creditors, licensors, licensees, employees and other Persons having business relationships with DMK.

(b) DMK shall promptly notify Adamis of: (i) any notice or other communication from any Person alleging that the Consent of such Person is or may be required in connection with any of the Contemplated Transactions; (ii) any material notice or other communication that it receives from any Governmental Body in connection with the Merger or other transactions contemplated by this Agreement; (iii) any material litigation relating to or involving or otherwise affecting DMK or Adamis that relates to the Merger or other transactions contemplated by this Agreement; and (iv) any event of which it becomes aware that would be considered reasonably likely to have a Material Adverse Effect on DMK or the Surviving Corporation.

4.4. Notification of Certain Matters. During the Pre-Closing Period, DMK on the one hand, and Adamis on the other, shall promptly notify the other Party (and, if in writing, furnish copies of) if any of the following occurs: (a) any notice or other communication is received from any Person alleging that the Consent of such Person is or may be required in connection with any of the Contemplated Transactions; (b) any Legal Proceeding against or involving or otherwise affecting such Party is commenced or, to the Knowledge of such Party, threatened against such Party or, to the Knowledge of such Party, any director or officer of such Party, relating to the Contemplated Transactions; (c) such Party becomes aware of any material inaccuracy in any representation or warranty made by such Party in this Agreement; or (d) the failure of such Party to comply with any covenant or obligation of such Party; in each of the above cases that could reasonably be expected to make the timely satisfaction of any of the closing conditions set forth in Articles VI, VII or VIII, as applicable, not possible. No notification given pursuant to this Section shall change, limit or otherwise affect any of the representations, warranties, covenants or obligations of the notifying Party contained in this Agreement or its Disclosure Schedule for purposes of Section 7.1 or 7.2 in the case of DMK, or Section 8.1 or 8.2 in the case of Adamis; and provided further, that the failure to give such notice will not be treated as a breach of covenant for the purposes of Articles VI, VII or VIII, unless the failure to give such notice results in material prejudice to the other party.

4.5. <u>No Solicitation</u>.

Each Party agrees that during the Pre-Closing Period, it shall not, and shall not authorize and shall use its reasonable best (a) efforts to cause its Representatives not to, directly or indirectly, (i) solicit, initiate, knowingly encourage, induce or facilitate the communication, making, submission or announcement of any Acquisition Proposal or Acquisition Inquiry or take any action that could reasonably be expected to lead to an Acquisition Proposal or Acquisition Inquiry; (ii) furnish any information regarding such Party to any Person in connection with or in response to an Acquisition Proposal or Acquisition Inquiry; (iii) engage in discussions or negotiations with any Person other than the other Party hereto and its Representatives regarding any Acquisition Proposal or Acquisition Inquiry; (iv) approve, endorse or recommend any Acquisition Proposal or, with respect to Adamis effect any Change in the Adamis Board Recommendation or, with respect to DMK, effect any change in the DMK Board Recommendation; or (v) enter into any letter of intent or similar document or any agreement contemplating or otherwise relating to any Acquisition Proposal (other than a confidentiality agreement containing terms similar in material respects to the Confidentiality Agreement) or enter into any agreement in principle requiring such Party to abandon, terminate or fail to consummate the Merger or breach its obligations hereunder or propose or agree to do any of the foregoing. For purposes of clarification, during the Pre-Closing Period, Adamis may continue to have discussions with third parties, and may enter into agreements or arrangements and may consummate one or more transactions regarding, its assets, business or securities as long as such agreements, arrangements or transactions do not constitute an Acquisition Proposal, Acquisition Inquiry or Alternative Transaction. If any Party or any Representative of such Party receives an Acquisition Proposal or Acquisition Inquiry at any time during the Pre-Closing Period, then such Party shall promptly (and in no event later than one business day after such Party becomes aware of such Acquisition Proposal or Acquisition Inquiry) advise the other Party orally and in writing of such Acquisition Proposal or Acquisition Inquiry (including the identity of the Person making or submitting such Acquisition Proposal or Acquisition Inquiry, and the terms thereof). Each Party shall immediately cease and cause to be terminated any existing discussions with any Person that relate to any Acquisition Proposal or Acquisition Inquiry as of the date of this Agreement, and shall instruct its Representatives accordingly. Each Party shall not terminate, release or permit the release of any Person from, or waive or permit the waiver of any provision of or right under, any confidentiality, nonsolicitation, no hire, "standstill" or similar agreement (whether entered into before or after the date of this Agreement) to which such Party is a party or under which such Party has any rights, and shall enforce or cause to be enforced each such agreement to the fullest extent possible.

(b) Notwithstanding anything contained in this Section:

before obtaining the Required Adamis Stockholder Vote, Adamis may furnish nonpublic information regarding Adamis (i) to, and enter into discussions or negotiations with, any Person in response to an unsolicited, bona fide written Acquisition Proposal made or received after the date of this Agreement, which the Adamis Board determines in good faith constitutes, or is reasonably likely to result in, a Superior Proposal (and is not withdrawn) if: (A) neither Adamis nor any Representative of Adamis shall have failed to comply with this Section; (B) the Adamis Board concludes in good faith, after consultation with outside counsel, that there is a reasonable risk that the failure to take such action would result in a breach of the fiduciary duties of the Adamis Board under applicable law; (C) within one business day following the furnishing of any such nonpublic information to, or entering into discussions with, such Person, Adamis gives DMK written notice of the identity of such Person and that Adamis intends to furnish nonpublic information to, or enter into discussions with, such Person or has furnished, or entered into discussions with, such Person; (D) Adamis receives from such Person an executed confidentiality agreement containing provisions (including nondisclosure provisions, use restrictions, non-solicitation provisions, no hire provisions and "standstill" provisions) at least as favorable to Adamis as those contained in the Confidentiality Agreement; and (E) within one business day following the furnishing of any such nonpublic information to such Person, Adamis furnishes such nonpublic information to DMK (to the extent such nonpublic information has not been previously furnished by Adamis to DMK). Without limiting the generality of the foregoing, Adamis acknowledges and agrees that in the event any Representative of Adamis takes any action that, if take by Adamis, would constitutes a failure to comply with this Section by Adamis, the taking of such action by such Representative shall be deemed to constitute a failure to comply with this Section by Adamis for purposes of this Agreement; and

(ii) Notwithstanding anything to the contrary set forth in this Agreement, if at any time before obtaining the Required Adamis Stockholder Vote, Adamis receives an unsolicited bona fide written Acquisition Proposal that did not relate to a breach of this Section and which the Adamis Board determines in good faith constitutes a Superior Proposal, and Adamis and its Representatives have otherwise complied in all material respects with its obligations under this Section 4.5, the Adamis Board may on four (4) Business Days' prior written Notice of Superior Proposal to DMK ("*Notice of Superior Proposal*") (which notice shall include the forms of agreements pursuant to which the Superior Proposal would be implemented or, if no such agreements have been proposed, a written summary of the material terms and conditions of such Superior Proposal) (it being understood that Adamis must deliver a new Notice of Superior Proposal and thereafter negotiate as provided herein in the event of any modification to an Acquisition Proposal if such modification results in the determination that such Acquisition Proposal is a Superior Proposal), take any action otherwise prohibited by Section 4.5(a) and cause Adamis to terminate this Agreement pursuant to Section 9.1(i) if (A) the Adamis Board shall have first determined in good faith, after consultation with outside counsel, that there is a reasonable risk that the failure to take such action would result in a breach of its fiduciary duties under the DGCL, and (B) Adamis shall have notified DMK of such determination and offered to discuss in good faith with DMK (and, if DMK accepts, thereafter negotiates in good faith), for a period of no less than four (4) Business Days, any adjustments in the terms and conditions of this Agreement proposed by DMK, and the Adamis Board shall have resolved, after taking into account the results of such discussions and proposals by DMK, if any, that the Acquisition Proposal remains a Superior Proposal.

(c) Nothing contained in this Section or in Section 5.3 shall prohibit Adamis from taking and disclosing to its stockholders a position with respect to a tender offer contemplated by Rule 14d-9 or Rule 14e-2 promulgated under the Exchange Act or from making any disclosure to Adamis' stockholders if, in the good faith judgment of the Adamis Board, after consultation with outside counsel, that there is a reasonable risk that the failure to so disclose would result in a breach of its fiduciary duties under the DGCL; provided that disclosure to stockholders pursuant to Rule 14e-2 relating to an Acquisition Proposal or Acquisition Inquiry shall be deemed to be a Change in the Adamis Board Recommendation unless the Adamis Board expressly, and without qualification, concurrently with such disclosure reaffirms the Adamis Board Recommendation.

(d) For purposes of clarification, during the Pre-Closing Period and thereafter, Adamis may continue to have discussions with third parties, and may enter into agreements or arrangements and may consummate one or more transactions regarding, its assets, business or securities as long as such agreement, arrangements or transactions do not constitute an Acquisition Transaction.

ARTICLE V ADDITIONAL AGREEMENTS

5.1. <u>Proxy Statement</u>.

(a) As promptly as reasonably practicable after the execution of this Agreement, Adamis, in cooperation with DMK, shall prepare and file with the SEC the Proxy Statement. The Proxy Statement shall, among other things, include the Adamis Board Recommendation and (i) solicit the approval of and include the recommendation of the Adamis Board to Adamis' stockholders that they vote in favor of the Adamis Proposals and the Other Adamis Proposals. DMK shall deliver to Adamis audited financial statements as of and for the years ended December 31, 2020 and 2021 (and, if determined to be required in Adamis' reports and filings with the SEC under applicable securities laws and SEC rules and regulations, 2022), with a report thereon from DMK's independent accounting firm (the "DMK Audited Financial Statements"), together with updated and reviewed DMK Interim Financial Statements. DMK shall promptly furnish to Adamis all other information concerning DMK, and shall use its commercially reasonable efforts to cause to be finished all information with respect to its stockholders, that is required to be disclosed in the Proxy Statement, or in subsequent filings that Adamis may make with the SEC.

(b) Adamis shall use all reasonable efforts to cause the Proxy Statement to comply with the applicable rules and regulations promulgated by the SEC, and shall respond promptly to any comments of the SEC or its staff and shall use all reasonable efforts to resolve any comments of SEC on the Proxy Statement as promptly as reasonably practicable. Adamis shall use its commercially reasonable efforts to cause the definitive Proxy Statement to be mailed to Adamis' stockholders after review by the SEC has been completed. Adamis shall notify DMK promptly upon the receipt of any comments from the SEC or its staff or any other government officials and of any request by the SEC or its staff or any other government officials for amendments or supplements to the Proxy Statement and shall supply DMK with copies of all substantive correspondence between Adamis or any of its Representatives, on the one hand, and the SEC, or its staff or any other government officials, on the other hand, with respect to the Proxy Statement. Whenever any event occurs which is required to be set forth in an amendment or supplement to the Proxy Statement, DMK or Adamis, as the case may be, shall use commercially reasonable efforts to promptly inform the other of such occurrence and cooperate in filing with the SEC or its staff or any other government officials, and/or mailing to stockholders of Adamis or DMK, such amendment or supplement. If Adamis becomes aware that any information in the Proxy Statement is or has become false or misleading in any material respect, Adamis shall take all reasonable steps to make such corrections as are required by applicable law or otherwise deemed appropriate by Adamis.

5.2. DMK Stockholder Approval.

(a) DMK shall take all action necessary under all applicable Law to call, give notice of and hold a meeting of the holders of DMK Capital Stock to vote on the approval of the Merger, adoption of this Agreement and the related plan of merger, and related matters (in either case, the "DMK Stockholders Meeting"). The DMK Stockholders Meeting shall be held reasonably after the date of this Agreement and after the DMK Information Statement is available. DMK shall use commercially reasonable efforts to ensure that all proxies and consents solicited in connection with the DMK Stockholders Meeting are solicited in compliance with all applicable Law. The Parties agree that the holders of DMK Capital Stock may take all action required under this Agreement by means of action by written consent as permitted by the NJBCA in lieu of an actual meeting of the holders of DMK Capital Stock.

(b) DMK agrees that: (i) the Board of Directors of DMK shall recommend that the holders of DMK Capital Stock vote to approve the Merger and adopt this Agreement and the related plan of merger, and such other matters as are contemplated by this Agreement, and shall use commercially reasonable efforts to solicit such approval (the recommendation of the board of directors of DMK that the stockholders of DMK vote to approve and adopt the Merger, this Agreement and the related plan of merger, and such other matters contemplated by this Agreement being referred to as the "**DMK Board Recommendation**"); and (ii) the Board of Directors of DMK shall not make or effect any change, withdrawal, qualification or modification of the DMK Board Recommendation in any manner that would reasonably be expected to prevent, delay or materially impair the consummation of the Merger or any of the other Contemplated Transactions.

5.3. <u>Adamis Stockholders Meeting; Change in the Adamis Board Recommendation; Adoption of Agreement by Adamis as Sole</u> <u>Stockholder of Merger Sub</u>.

(a) Adamis shall take all action necessary under applicable Law to call, give notice of and hold a meeting of the stockholders of Adamis to vote on the Adamis Proposals (such meeting, the "*Adamis Stockholders Meeting*"). Adamis shall use its commercially reasonable efforts to ensure that all proxies solicited in connection with the Adamis Stockholders Meeting are solicited in compliance with all applicable Law.

(b) Adamis agrees that, subject to Sections 4.5 and 5.6(c): (i) the Adamis Board shall recommend that the stockholders of Adamis vote to approve and adopt (as the case may be), (A) the Adamis Share Issuance Proposal, and (B) the Adamis Reverse Stock Split Amendment (proposals (A) and (B) being referred to as the "*Adamis Proposals*"), and the Adamis Board may also approve and recommend that the stockholders approve additional proposals, and such other matters as the Adamis Board may determine are reasonably necessary or appropriate to effect the Merger and the other Contemplated Transactions (the "*Other Adamis Proposals*"), and shall use commercially reasonable efforts to solicit such approval or adoption, as the case may be; (ii) the Proxy Statement shall include a statement to the effect that the Board of Directors of Adamis recommends that the stockholders of Adamis vote to approve and adopt (as the case may be) the Adamis Proposals (such recommendation being referred to herein as the "*Adamis Board Recommendation*"); and (iii) except as expressly permitted by this Agreement, the Board of Directors of Adamis shall not make or effect any Change in the Adamis Board Recommendation.

(c) Subject to Sections 4.5 and 5.6(c), Adamis shall take all action that is both reasonable and lawful to solicit the approval of its stockholders of the Adamis Proposals and shall take all other action reasonably necessary or advisable to secure the vote or consent of the stockholders of Adamis required by the DGCL to obtain such approvals. If, on the date of Adamis Stockholders Meeting or any subsequent adjournment thereof pursuant to this Section, a quorum for the conduct of business does not exist or Adamis has not received proxies representing a sufficient number of shares of Adamis voting stock to approve the Adamis Proposals, Adamis may adjourn the Adamis Stockholders Meeting to solicit additional proxies for such proposals, and shall continue to use its commercially reasonable efforts, together with its proxy solicitor, to assist in the solicitation of proxies from stockholders relating to the Adamis Proposals.

(d) Adamis, as sole stockholder of Merger Sub, shall adopt this Agreement and the related plan of merger as soon as practicable following the Execution Date, including by means of action by written consent, as permitted by the NJBCA in lieu of an actual meeting of the stockholders of Merger Sub.

5.4. Registration Statement. Within sixty (60) days following the Closing Date, or as soon thereafter as is reasonably practicable including, without limitation, (a) taking into account any restrictions, limitations or prohibitions on the filing of registration statements by Adamis that may be contained in any agreement entered into by Adamis in connection with any financing transaction completed after the Agreement Date, and (b) applicable financial statement, audit and auditor requirements concerning the auditor reports and financial statements that are required to be included in the Registration Statement, Adamis will prepare and file with the SEC (x) a registration statement on Form S-3 (or if Form S-3 is not available, such other form as may provide for a resale of the Merger Consideration shares, but with such registration obligations otherwise consistent with the requirements of this Section) (such a new registration statement, together with all amendments and supplements thereto, including post-effective amendments, all exhibits thereto and all material incorporated by reference therein, the "*Registration Statement*"), or (y) a prospectus supplement pursuant to an existing effective registration statement of Adamis, covering the resale of the Merger Consideration shares of Adamis Common Stock (including the Series E Preferred Conversion Shares) issued pursuant to Section 1.6. If Adamis files such a Registration Statement, Adamis will use commercially reasonable efforts to cause the Registration Statement to be declared effective as soon as reasonably possible following the filing of the Registration Statement and be maintained (or, if applicable, the prospectus supplement maintained) until, with respect to an individual former DMK stockholder, the earlier to occur of (i) the date that all such Merger Consideration shares held by such stockholder may be publicly resold without restrictions pursuant to the provisions of Rule 144, or (ii) the date that all of such Merger Consideration shares have actually been resold; provided, however, that Adamis will have no obligation to file the Registration Statement or prospectus supplement, or have the Registration Statement declared effective or maintain such prospectus supplement, unless and until any required Form 8-K Amendment relating to the Contemplated Transactions, if required by applicable law, is filed with the SEC (and Adamis shall use its reasonable efforts to timely file the Form 8-K Amendment, if required). Notwithstanding the registration obligations set forth in this Section, in the event the SEC informs Adamis that all of the shares requested to be included in the registration cannot, as a result of the application of Rule 415, be registered for resale on such registration statement, Adamis agrees to promptly inform the holders of such shares and use its commercially reasonable efforts to file amendments to the Registration Statement as required by the SEC covering the maximum number of Registrable Securities that are permitted to be registered by the SEC, and the limitation on the number of such Merger Consideration shares included in the Registration Statement as a result of the application of Rule 415 shall not be deemed to be a breach of any provision of this Agreement. For not more than sixty (60) consecutive days or for a total of not more than ninety (90) days in any twelve (12) month period, Adamis may suspend the use of any prospectus included in the Registration Statement (or the use of the applicable prospectus supplement) if Adamis determines in good faith that such suspension is necessary to (x) delay the disclosure of material non-public information concerning Adamis, the disclosure of which at the time is not, in the good faith opinion of Adamis, in the best interests of Adamis and its stockholders, or (y) amend or supplement the Registration Statement or the related prospectus or prospectus supplement so that the Registration Statement, prospectus or prospectus supplement will not include an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in the case of the prospectus in light of the circumstances under which they were made, not misleading.

5.5. Information Statement; DMK Written Consent.

(a) As promptly as reasonably practicable after the date of this Agreement, DMK will prepare, with the cooperation of Adamis, and cause to be delivered to its stockholders an information statement for the holders of DMK Capital Stock (the "*Information Statement*") to solicit the written consent of its stockholders in lieu of a meeting pursuant to the NJBCA (the "*DMK Written Consent*") for purposes of (i) adopting this Agreement and the relating plan of merger and approving the Merger, and all other transactions contemplated by this Agreement, (ii) acknowledging that the approval given thereby is irrevocable, that such DMK Stockholder is aware of its rights to demand appraisal for its shares pursuant to Section 14A:11 of the NJBCA, and that by its approval of the Merger it is not entitled to appraisal rights with respect to its shares in connection with the Merger and thereby waives any rights to receive payment of the fair value of its DMK Capital Stock under the NJBCA, and (iii) acknowledging the conversion of all outstanding DMK Convertible Notes into DMK Common Stock prior to the Effective Time (collectively, the "*DMK Stockholder Matters*"). DMK shall use commercially reasonable efforts to cause the Information Statement to comply with applicable federal and state securities laws requirements. Each of Adamis and DMK agree to provide promptly to the other such information concerning its business and financial statements and affairs as, in the reasonable judgment of the providing party or its counsel, may be required or appropriate for inclusion in the Information Statement, or in any amendments or supplements thereto. DMK shall not include in the Information Statement any information with respect to Adamis or its affiliates, the form and content of which information has not been approved by Adamis prior to such inclusion.

(b) (i) The board of directors of DMK will recommend that its stockholders vote to approve the DMK Stockholder Matters (such recommendation the "*DMK Board Recommendation*"); (ii) the Information Statement will include the DMK Board Recommendation; and (iii) the DMK Board Recommendation will not be withdrawn or modified in a manner adverse to Adamis, and no resolution by the board of directors of DMK or any committee thereof to withdraw or modify the DMK Board Recommendation in a manner adverse to Adamis will be adopted or proposed.

5.6. Adamis Stockholders Meeting.

(a) Adamis will take all action necessary under applicable Law to call, give notice of and hold the Adamis Stockholders Meeting to vote on (i) the Adamis Proposals, and (ii) the Other Adamis Proposals. The Adamis Stockholders Meeting will be held on a date selected by Adamis, following the date on which the definitive Proxy Statement may be mailed to the Adamis stockholders; provided, however, notwithstanding anything to the contrary contained herein, Adamis will have the absolute discretion to adjourn the Adamis Stockholders Meeting for a period of up to sixty (60) days after the initial date of the Adamis Stockholders Meeting is scheduled or convened if Adamis determines it necessary or appropriate to solicit additional proxies in favor of the Adamis Proposals.

(b) Subject to Section 5.6(c): (i) the board of directors of Adamis will recommend that its stockholders vote to approve the Adamis Proposals (such recommendation, the "*Adamis Board Recommendation*"); (ii) the Proxy Statement will include the Adamis Board Recommendation; and (iii) the Adamis Board Recommendation will not be withdrawn or modified in a manner adverse to DMK.

Notwithstanding anything to the contrary contained in this Agreement, at any time before the approval and adoption of the (c) Adamis Proposals by the Required Adamis Stockholder Vote, the Board of Directors of Adamis may effect a Change in the Adamis Board Recommendation in accordance with the provisions of Section 4.5(b), provided that DMK must receive not less than four (4) Business Days prior written notice from Adamis confirming that Adamis' Board of Directors has determined to make a Change in the Adamis Board Recommendation. For purposes of this Agreement, "Change in the Adamis Board Recommendation" means any: (i) withholding, withdrawal, qualification or modification of (or any proposal or resolution to withhold, withdraw, qualify or modify) the Adamis Board Recommendation in any manner adverse to DMK; (ii) action or statement by Adamis in connection with Adamis Stockholders Meeting contrary to the Adamis Board Recommendation; (iii) taking any position other than opposition (including making no recommendation), by the Adamis Board with respect to an Acquisition Proposal that has been publicly disclosed or otherwise become publicly known to any Person other than Adamis, DMK and their respective Representatives after a reasonable amount of time has elapsed for the Adamis Board to review and make a recommendation with respect thereto (and in no event more than ten (10) Business Days after being so publicly disclosed or otherwise become public known); (iv) failure of the Adamis Board to (A) if a tender offer, take-over bid or exchange offer that constitutes or would constitute an Acquisition Proposal (other than by DMK) is commenced, recommend that the Adamis stockholders not accept such tender offer, take-over bid or exchange offer after a reasonable amount of time has elapsed for the Adamis Board to review and make a recommendation with respect thereto (and in no event more than ten Business Days following commencement of such tender offer, take-over bid or exchange offer), or (B) reaffirm in writing the Adamis Board Recommendation in connection with a disclosure pursuant to Section 4.5(c) or otherwise within two (2) Business Days of a request by DMK to do so; or (v) approval, adoption or recommendation, or publicly disclosed proposal to approve, adopt or recommend, an Acquisition Proposal.

(d) Nothing contained in this Agreement will prohibit Adamis or its board of directors from complying with Rules 14d-9 and 14e-2(a) promulgated under the Exchange Act; provided, however, that any disclosure made by Adamis or its board of directors pursuant to Rules 14d-9 and 14e-2(a) will be limited to a statement that Adamis is unable to take a position with respect to the bidder's tender offer unless the board of directors of Adamis determines in good faith, after consultation with its outside legal counsel, that there is a reasonable risk that the failure to make such a statement would result in a breach of its fiduciary duties under applicable Law.

5.7. <u>Regulatory Approvals</u>. Each Party shall use commercially reasonable efforts to file or otherwise submit, as soon as practicable after the date of this Agreement, all applications, notices, reports and other documents reasonably required to be filed or otherwise submitted by such Party with or to any Governmental Entity with respect to the Merger and the other Contemplated Transactions, and to submit promptly any additional information requested by any such Governmental Entity.

5.8. Indemnification of Officers and Directors.

(a) From and after the Effective Time through the third anniversary of the date the Effective Time occurs, Adamis shall and shall cause the Surviving Corporation to, fulfill and honor in all respects the obligations of Adamis and DMK pursuant to any indemnification provisions under their respective certificates of incorporation and bylaws regarding indemnification of directors and officers that are in effect on the date of this Agreement (the persons entitled to be indemnified pursuant to such provisions being referred to collectively as the "**D&O Indemnified Parties**"); provided, however, that after the Closing and the Effective Time of the Merger, the DMK directors and officers who become directors and officers of Adamis will enter into the Adamis standard indemnification agreement which will supersede any other contractual rights to indemnification by the Surviving Corporation.

(b) The certificate of incorporation and bylaws of Adamis and the Surviving Corporation, as the case may be, shall contain provisions no less favorable with respect to indemnification and elimination of liability for monetary damages of present and former directors and officers than are presently set forth in the certificate of incorporation and bylaws of DMK, which provisions shall not be amended, modified or repealed for a period of six (6) years from the Effective Time in a manner that would adversely affect the rights thereunder of the D&O Indemnified Parties.

(c) Adamis, at its election, may purchase a "tail" policy on DMK's existing directors and officers liability insurance policy (if any) for a period of six (6) years after the Closing. Adamis will maintain either a directors and officers liability insurance policy or a "tail" policy on Adamis' existing directors and officers.

(d) The provisions of this Section are intended to be in addition to the rights otherwise available to the D&O Indemnified Parties by law, charter, statute, by-law or agreement, and shall operate for the benefit of, and shall be enforceable by, each of the D&O Indemnified Parties, their heirs and their representatives.

(e) Adamis shall cause the Surviving Corporation to perform all of the obligations of the Surviving Corporation under this Section.

5.9. Additional Agreements.

(a) Subject to Sections 4.5, 5.2, 5.3, 5.5 and 5.6, the Parties shall use commercially reasonable efforts to cause to be taken all actions necessary to consummate the Merger and make effective the other Contemplated Transactions. Without limiting the generality of the foregoing, but subject to paragraph (b) below, each Party to this Agreement: (i) shall make all filings and other submissions (if any) and give all notices (if any) reasonably required to be made and given by such Party in connection with the Merger and the other Contemplated Transactions; (ii) shall use commercially reasonable efforts to obtain each Consent (if any) reasonably required to be obtained (pursuant to any applicable Law or Material Contract) by such Party in connection with the Merger or any of the other Contemplated Transactions or for such Contract to remain in full force and effect; (iii) shall use commercially reasonable efforts to lift any injunction prohibiting, or any other legal bar to, the Merger or any of the other Contemplated Transactions; and (iv) shall use commercially reasonable efforts to satisfy the conditions precedent to the consummation of this Agreement. Each Party shall provide to the other Party a copy of each proposed filing with or other submission to any Governmental Entity relating to any of the Contemplated Transactions, and shall (if reasonably practicable) give the other Party a reasonable time before making such filing or other submission in which to review and comment on such proposed filing or other submission. Each Party shall promptly deliver to the other Party a copy of each such filing or other submission made, each notice given and each Consent obtained by such Party during the Pre-Closing Period.

(b) Notwithstanding anything to the contrary contained in this Agreement, no Party shall have any obligation under this Agreement: (i) to dispose of or transfer any assets; (ii) to discontinue offering any product or service; (iii) to license or otherwise make available to any Person any Intellectual Property; (iv) to hold separate any assets or operations (either before or after the Closing Date); (v) to make any commitment (to any Governmental Entity or otherwise) regarding its future operations; or (vi) to contest any Legal Proceeding or any order, writ, injunction or decree relating to the Merger or any of the other Contemplated Transactions if such Party determines in good faith that contesting such Legal Proceeding or order, writ, injunction or decree might not be advisable.

(c) Prior to the Closing, Adamis shall have the right to control the defense and settlement of any stockholder litigation against Adamis or any of its directors or officers relating to this Agreement or the Contemplated Transactions, and shall, subject to applicable considerations of privilege and confidentiality, keep DMK reasonably apprised of any material developments in connection with any such litigation.

5.10. <u>Disclosure; Public Announcements</u>. The Parties shall use their commercially reasonable efforts to agree on a mutually acceptable joint initial press release and Adamis Form 8-K announcing the execution and delivery of this Agreement. Without limiting any of either Party's obligations under the Confidentiality Agreement, each Party shall not issue any press release or make any public disclosure regarding the Merger or any of the other Contemplated Transactions unless: (a) the other Party shall have approved such press release or disclosure (which approval shall not be unreasonably withheld or delayed); or (b) such Party shall have determined in good faith, upon the advice of outside legal counsel, that such disclosure is required by applicable Law. Notwithstanding the foregoing, without consulting with the other Party, Adamis and its officers may make public statements, including in response to specific questions by the press, analysts, investors or those attending industry conferences or financial analyst conference calls, so long as any such statements are consistent in material respects with previous press releases, public disclosures or public statements made by DMK or Adamis in compliance with this Section.

5.11. Directors; Officers. Prior to the Effective Time, and subject to the receipt of any required stockholder vote, Adamis shall use its best efforts to take all actions necessary (i) to cause the authorized number of members of the Board of Directors of Adamis to remain at five (5) or to be such other number as Adamis and DMK may mutually agree, effective at the Effective Time; (ii) to obtain the resignations, effective at the Effective Time, of the directors of Adamis who will not continue as directors after the Effective Time as contemplated by this Agreement, and (iii) so that the Persons listed in <u>Annex 5.11</u> are elected or appointed, as applicable, to the positions of officers and directors of Adamis, as set forth therein, to serve in such positions effective as of the Effective Time. If during the Pre-Closing period any Person listed in <u>Annex 5.11</u> indicates that he or she is unable or unwilling to serve as officer or director of Adamis or the Surviving Corporation, then Adamis and DMK will use their best efforts to agree on a successor. Notwithstanding the foregoing, the Parties acknowledge that so long as Adamis remains a public reporting company, Adamis and DMK will use their best efforts such that the board of directors of Adamis will continue to satisfy applicable securities laws and listing requirements of any exchange or market on which the Adamis Common Stock is then listed or traded, including, without limitation, maintaining an independent audit committee and having an audit committee financial expert serve on such committee, and the actions taken by Adamis and DMK hereunder regarding the composition of the Board of Directors of Adamis immediately after the Effective Time will be taken so as to allow Adamis to comply with such applicable Law.

5.12. Tax Matters.

(a) Adamis, Merger Sub and DMK shall use their respective commercially reasonable efforts to cause the Merger to qualify, and shall use their respective commercially reasonable efforts not to, and not to permit or cause any affiliate to, take any actions or cause any action to be taken which would prevent the Merger from qualifying, as a "reorganization" under Section 368(a) of the Code.

(b) This Agreement is intended to constitute, and the Parties hereto hereby adopt this Agreement as, a "plan of reorganization" within the meaning Treasury Regulation Sections 1.368-2(g) and 1.368-3(a). Adamis, Merger Sub and DMK shall report the Merger as reorganization within the meaning of Section 368(a) of the Code, unless otherwise required pursuant to a "determination" within the meaning of Section 1313(a) of the Code.

(c) DMK shall (and shall cause its Affiliates to) provide any information reasonably requested to allow Adamis to comply with any information reporting or withholding requirements contained in the Code or other applicable Laws with respect to the transactions contemplated by, or any payment made in connection with, this Agreement.

(d) All transfer, documentary, sales, use, value added, goods and services, stamp, registration, notarial fees and other similar Taxes and fees (collectively, "**Transfer Taxes**"), shall be paid by the Surviving Corporation. After the Closing Date, the Surviving Corporation will prepare and file all necessary Tax Returns and other documentation with respect to all such Transfer Taxes that are required to be filed after the Closing Date, and, if required by applicable Law, Adamis will, and will cause its Affiliates to, cooperate and join in the execution of any such Tax Returns and other documentation, as applicable. Each Party shall (and shall cause its Affiliates to) provide certificates or forms, and timely execute any Tax Returns, that are necessary or appropriate to establish an exemption for (or reduction in) any Transfer Tax.

(e) Each of the Parties shall (and shall cause their respective Affiliates to) cooperate fully, as and to the extent reasonably requested by another Party, in connection with the filing of relevant Tax Returns, and any audit or tax proceeding. Such cooperation shall include the retention and (upon the other Party's request) the provision (with the right to make copies) of records and information reasonably relevant to any tax proceeding or audit, making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder.

5.13. <u>Adamis Amendments</u>. Subject to Section 5.6(c), Adamis agrees to recommend to its stockholders that the Adamis Restated Certificate be amended by means of the Adamis Reverse Stock Split Amendment to effect a reverse stock split of the Adamis Common Stock by a ratio determined by the Adamis Board (the "*Adamis Reverse Stock Split Proposal*").

5.14. DMK's Auditors. DMK will use its commercially reasonable efforts to cause its management and its independent auditors to facilitate on a timely basis (a) the preparation of financial statements (including audited and pro forma financial statements if required) as required by Adamis to comply with Law, (b) the review of DMK's audit work papers for up to the past two years or such lesser period of which DMK has been in existence, including the examination of selected interim financial statements and data, and (c) such auditor reports and consents relating to the including to the inclusion of the DMK Financial Statements in the Proxy Statement, reports that Adamis files with the SEC, and such other documents and filings as Adamis or DMK may be determine are necessary or appropriate.

5.15. Legends. Adamis shall be entitled to place such appropriate legends on the certificates evidencing any shares of Adamis Common Stock to be received in the Merger by equity holders of DMK Capital Stock, including, without limitation, persons who may after the Effective Time be considered "affiliates" of Adamis for purposes of Rules 144 and 145 under the Securities Act, as Adamis reasonably determines is required or appropriate under applicable laws, and to issue appropriate stop transfer instructions to the transfer agent for such shares of Adamis Common Stock.

5.16. <u>**Confidentiality**</u>. Each of Adamis and DMK hereby agrees that the information obtained in any investigation pursuant to this Agreement, or pursuant to the negotiation and execution of this Agreement or the effectuation of the transaction contemplated hereby shall be governed by the terms of the Confidentiality Agreement dated as of April 29, 2022, previously executed by and between DMK and Adamis (the "*Confidentiality Agreement*").

5.17. Section 16 Matters. Subject to the following sentence, prior to the Effective Time, Adamis and DMK will take all such steps as may be required (to the extent permitted under applicable Law and no-action letters issued by the SEC) to cause any acquisition of Adamis Common Stock (including derivative securities with respect to Adamis Common Stock) by each officer or director of DMK (in their capacity as officer or director) who is or will become subject to the reporting requirements of Section 16(a) of the Exchange Act with respect to Adamis, to be exempt under Rule 16b-3 under the Exchange Act. Prior to the Closing Date, DMK will furnish the following information to Adamis for each director, officer or securityholder of DMK who, immediately after the Effective Time, will become subject to the reporting requirements of Section 16(a) of the Exchange Act with respect to Adamis: (a) the number of shares of DMK Capital Stock held by such individual and expected to be exchanged for shares of Adamis Common Stock pursuant to the Merger; (b) the number of other derivative securities (if any) with respect to Adamis Common Stock in connection with the Merger; and (c) such other information as may be reasonably required in connection with compliance with Section 16(a) of the Exchange Act.

5.18. Listing. If the Adamis Common Stock is then listed on the Nasdaq Capital Market ("*Nasdaq*"), Adamis will (a) to the extent required by the rules and regulations of Nasdaq, prepare and submit to Nasdaq a notification form for the listing of the shares of Adamis Common Stock to be issued and issuable pursuant to the Merger and use all reasonable efforts to cause such shares to be approved for listing (subject to notice of issuance), and (b) to the extent required by Nasdaq Marketplace Rules, file an initial listing for the Adamis Common Stock (the "*Nasdaq Listing Application*") and use its reasonable best efforts to cause such Nasdaq Listing Application to be conditionally approved prior to the Effective Time. DMK will cooperate with Adamis as reasonably requested by Adamis with respect to such Nasdaq Listing Application and will promptly furnish to Adamis all information concerning DMK and its stockholders that may be required or reasonably requested in connection with any action contemplated by this Section, or in any appeal by Adamis to Nasdaq or a Nasdaq hearings panel concerning any de-listing determination that Nasdaq may make.

5.19. Private Placement. In connection with the solicitation of the DMK Stockholder Approval, DMK shall deliver to each holder of DMK Capital Stock and each holder of a DMK Convertible Note a customary form of investor questionnaire reasonably satisfactory to Adamis. In reliance on such investor questionnaires, Adamis and DMK shall take such action as may be necessary or appropriate so that the issuance of shares of Adamis Common Stock and Series E Preferred as Merger Consideration in the Merger shall validly qualify for an exemption from the registration and prospectus delivery requirements of the Securities Act and the equivalent state "blue-sky" laws.

5.20. <u>Termination of Certain DMK Agreements</u>. Prior to the Closing Date, DMK will obtain the necessary written consent or agreement of its stockholders or other counterparties, as applicable, to, effective upon the Closing Date, terminate the agreements to which DMK is a party or by which DMK is bound, if any, that are set forth on Schedule 5.20.

5.21. DMK Options.

At the Effective Time, each DMK Option that is outstanding and unexercised immediately prior to the Effective Time (a) (a) under the DMK Stock Plan, whether or not vested, will be converted into and become an option to purchase Adamis Common Stock, in a manner consistent with the requirements of Section 409A and, for DMK Options gualified under Section 422 of the Code, Section 424 of the Code, and Adamis shall assume the DMK Stock Plan. All rights with respect to DMK Common Stock under DMK Options assumed by Adamis will thereupon be converted into rights with respect to Adamis Common Stock. Accordingly, from and after the Effective Time: (i) each DMK Option assumed by Adamis may be exercised solely for shares of Adamis Common Stock; (ii) the number of shares of Adamis Common Stock subject to each DMK Option assumed by Adamis will be determined by multiplying (x) the number of shares of DMK Common Stock that were subject to such DMK Option, as in effect immediately prior to the Effective Time by (y) the Exchange Ratio and rounding the resulting number down to the nearest whole number of shares of Adamis Common Stock; (iii) the per share exercise price for the Adamis Common Stock issuable upon exercise of each DMK Option assumed by Adamis will be determined by dividing (x) the per share exercise price of DMK Common Stock subject to such DMK Option, as in effect immediately prior to the Effective Time, by (y) the Exchange Ratio and rounding the resulting exercise price up to the nearest whole cent; and (iv) any restriction on the exercise of any DMK Option assumed by Adamis will continue in full force and effect and the term, exercisability, vesting schedule, status as an "incentive stock option" under Section 422 of the Code, if applicable, and other provisions of such DMK Option will otherwise remain unchanged; provided, however, that: (1) to the extent provided under the terms of a DMK Option, such DMK Option assumed by Adamis in accordance with this Section will, in accordance with its terms, be subject to further adjustment as appropriate to reflect any stock split, division or subdivision of shares, stock dividend, reverse stock split, consolidation of shares, reclassification, recapitalization or other similar transaction with respect to Adamis Common Stock subsequent to the Effective Time; and (2) Adamis' board of directors or a committee thereof will succeed to the authority and responsibility of DMK's board of directors or any committee thereof with respect to each DMK Option assumed by Adamis. Notwithstanding anything to the contrary in this Section, the conversion of each DMK Option (regardless of whether such option qualifies as an "incentive stock option" within the meaning of Section 422 of the Code) into an option to purchase shares of Adamis Common Stock will be made in a manner consistent with Treasury Regulation Section 1.424-1, such that to the maximum extent possible, the conversion of a DMK Option will not constitute a "modification" of such DMK Option for purposes of Section 409A or Section 424 of the Code. It is the intention of the Parties that each DMK Option so assumed by Adamis shall qualify following the Effective Time as an incentive stock option as defined in Section 422 of the Code to the extent permitted under Section 422 of the Code and to the extent such DMK Option qualified as an incentive stock option prior to the Effective Time.

(b) After the Effective Time, Adamis will file with the SEC a registration statement on Form S-8, or a post-effective amendment to an existing Form S-8 registration statement, relating to the DMK Stock Plan and the shares of Adamis Common Stock issuable with respect to DMK Options assumed by Adamis in accordance with Section, to the extent permitted by federal securities laws, and Adamis shall use its commercially reasonable efforts to maintain the effectiveness of such registration statement or registration statements (and maintain the current status of the prospectus or prospectuses delivered with respect to such shares) for so long as such options remain outstanding.

(c) After the Effective Time, Adamis will use all reasonable efforts to enter into an assumption agreement or other agreement reasonably satisfactory to Adamis with each person who, immediately prior to the Effective Time, was a holder of a DMK Option, evidencing the foregoing assumption of such DMK Options by Adamis.

5.22. <u>Allocation Certificate</u>. DMK will prepare and deliver to Adamis at least two (2) Business Days (if practicable) prior to the Closing Date a certificate signed by the Chief Financial Officer and Secretary of DMK in a form reasonably acceptable to Adamis which sets forth (a) a true and complete list of the DMK Stockholders immediately prior to the Effective Time (including holders of DMK Convertible Notes that will convert into shares of DMK Common Stock) and the number and type of shares of DMK Capital Stock owned by each such DMK Stockholder, and (b) the allocation of the Merger Consideration among the DMK Stockholders pursuant to the Merger (the "Allocation Certificate").

5.23. Employee Benefit Matters. All employees of DMK shall continue in their existing benefit plans until such time as, in Adamis' discretion, an orderly transition can be accomplished to employee benefit plans and programs maintained by or on behalf of Adamis for its employees. Adamis shall take such reasonable actions, to the extent permitted by Adamis' benefits programs, as are necessary to allow eligible employees of DMK to participate in the health, welfare and other benefit programs of Adamis or alternative benefits programs in the aggregate that are substantially similar to those applicable to employees of Adamis in similar functions and positions on similar terms (it being understood that equity incentive plans are not considered employee benefits). Pending such action, Adamis and DMK shall cooperate to maintain the effectiveness of the DMK's benefit plans. To the extent permitted by and consistent with Adamis' benefit plans, employees of DMK shall be given credit for all service with DMK for purposes of eligibility and vesting (but not for purposes of benefit accrual) under all employee benefit plans, programs, policies and arrangements and employment policies maintained by Adamis in which they become participants. In connection with such transition, if DMK maintains any Code Section 401(k) arrangements or plans (each a "DMK 401(k) Plan"), then at Adamis' request, Adamis and DMK will cooperate with respect to the termination of the DMK 401(k) Plan.

5.24. <u>Takeover Statutes</u>. If any takeover statute is or may become applicable to the Contemplated Transactions, each of DMK, the DMK Board, Adamis and the Adamis Board, as applicable, shall grant such approvals and take such actions as are necessary so that the Contemplated Transactions may be consummated as promptly as practicable on the terms contemplated by this Agreement and otherwise act to eliminate or minimize the effects of such statute or regulation on the Contemplated Transactions.

5.25. <u>Merger Sub</u>. Adamis will take all action that are necessary to cause Merger Sub to perform its obligations under this Agreement and to consummate the Merger, subject to the terms and conditions set forth in this Agreement.

5.26. Supplement to Disclosure Schedules. (i) From time to time prior to the Closing, DMK shall have the right (but not the obligation) to supplement or amend the DMK Disclosure Schedule hereto with respect to any matter hereafter arising or of which it becomes aware after the date hereof (each a "Schedule Supplement"). Any disclosure in any such Schedule Supplement shall not be deemed to have cured any inaccuracy in or breach of any representation or warranty contained in this Agreement, including for purposes of the termination rights contained in this Agreement or of determining whether or not the conditions set forth in Article VII have been satisfied; provided, however, that if as a result of matters disclosed in such Schedule Supplement, Adamis has the right to, but does not elect to, terminate this Agreement within ten (10) Business Days of its receipt of such Schedule Supplement, then Adamis shall be deemed to have irrevocably waived any right to terminate this Agreement"). Any disclosure in any such Schedule Supplement or amend the Adamis Disclosure Schedule hereto with respect to any matter hereafter arising or of which it becomes aware after the date hereof (each a "Schedule Supplement"). Any disclosure in any such Schedule Supplement shall not be deemed to have cured any inaccuracy in or breach of any representation or warranty contained in this Agreement, including for purposes of the termination rights contained in this Agreement or of determining whether or not the conditions set forth in Article VIII have been satisfied; provided, however, that if as a result of matters disclosed in such Schedule Supplement, DMK has the right to, but does not elect to, terminate this Agreement, DMK has the right to, but does not elect to, terminate this Agreement within ten (10) Business Days of its receipt of such Schedule Supplement, DMK shall be deemed to have irrevocably waived any right to terminate this Agreement with respect to such matter.

5.27. Series E Preferred Certificate of Designation. The Series E Preferred Certificate of Designation shall be substantially in the form attached hereto as <u>Exhibit C</u>, with such changes thereto as Adamis and DMK may mutually agree.

ARTICLE VI CONDITIONS PRECEDENT TO OBLIGATIONS OF EACH PARTY

The obligations of each Party to effect the Merger and otherwise consummate the transactions to be consummated at the Closing are subject to the satisfaction or, to the extent permitted by applicable law, the written waiver by each of the Parties, at or before the Closing, of each of the following conditions:

6.1. <u>Stockholder Approval</u>. This Agreement and the Merger shall have been duly adopted by the Required DMK Stockholder Vote; the board of directors and shareholders of Merger Sub shall have approved and adopted this Agreement and the Merger; and the Adamis Proposals shall have been duly approved or adopted, as the case may be, by the Required Adamis Stockholder Vote.

6.2. No Restraints. No temporary restraining order, preliminary or permanent injunction or other order preventing the consummation of the Merger shall have been issued by any court of competent jurisdiction or other Governmental Entity and remain in effect, and there shall not be any Law which has the effect of making the consummation of the Merger illegal.

6.3. <u>Governmental Authorization</u>. Any Governmental Authorization or other Consent required to be obtained by any of the Parties under any applicable antitrust or competition law or regulation or other Law shall have been obtained and shall remain in full force and effect.

6.4. <u>No Governmental Proceedings Relating to Contemplated Transactions or Right to Operate Business</u>. There shall not be any Legal Proceeding pending, or overtly threatened in writing by an official of a Governmental Entity in which such Governmental Entity indicates that it intends to conduct any Legal Proceeding or taking any other action: (a) challenging or seeking to restrain or prohibit the consummation of the Merger or any of the other Contemplated Transactions; (b) relating to the Merger and seeking to obtain from Adamis, Merger Sub or DMK any damages or other relief that would have a Material Adverse Effect on the Combined Company; (c) seeking to prohibit or limit in any material and adverse respect a Party's ability to vote, transfer, receive dividends with respect to or otherwise exercise ownership rights with respect to the stock of Adamis; (d) that could have a Material Adverse Effect on the right or ability of the Combined Company to own the assets or operate the business of the Combined Company; or (e) seeking to compel DMK or Adamis to dispose of or hold separate any assets that are material to the Combined Company as a result of or following the Merger or any of the Contemplated Transactions.

6.5. <u>Series E Preferred Certificate of Designation</u>. The Series E Preferred Certificate of Designation shall have been filed with the Secretary of State of Delaware, which shall continue to be in full force and effect as of the Closing.

ARTICLE VII ADDITIONAL CONDITIONS PRECEDENT TO OBLIGATIONS OF ADAMIS AND MERGER SUB

The obligations of Adamis and Merger Sub to effect the Merger and otherwise consummate the transactions to be consummated at the Closing are subject to the satisfaction or the written waiver by Adamis, at or before the Closing, of each of the following conditions:

7.1. <u>Accuracy of Representations</u>. The representations and warranties of DMK contained in this Agreement shall have been true and correct as of the date of this Agreement and shall be true and correct on and as of the Closing Date with the same force and effect as if made on the Closing Date, except (A) in each case, or in the aggregate, where the failure to be true and correct would not reasonably be expected to have a Material Adverse Effect on the Combined Company, or (B) for those representations and warranties which address matters only as of a particular date (which representations shall have been true and correct, subject to the qualifications as set forth in the preceding clause (A), as of such particular date).

7.2. Performance of Covenants. Each of the covenants and obligations in this Agreement that DMK is required to comply with or to perform at or before the Closing shall have been complied with and performed by DMK in all material respects, except where the failure to perform such covenants or obligations would not have a Material Adverse Effect on the Combined Company.

7.3. No Material Adverse Effect. From the Execution Date through the Effective Time, there shall not have occurred any Material Adverse Effect on DMK that shall be continuing as of the Effective Time and that would have a Material Adverse Effect on the Combined Company.

7.4. <u>Agreements and Other Documents</u>. Adamis shall have received the following agreements and other documents, each of which shall be in full force and effect:

(a) a certificate of DMK executed on its behalf by the Chief Executive Officer and Chief Financial Officer of DMK confirming that the conditions set forth in Sections 7.1, 7.2, and 7.3 have been duly satisfied; and

(b) certificates of good standing (or equivalent documentation) of DMK in its jurisdiction of incorporation and the various foreign jurisdictions in which it is qualified (except where the failure to have obtained such certificates would not result in a Material Adverse Effect on the Combined Company), certified charter documents, a certificate as to the incumbency of officers and the adoption of resolutions of the Board of Directors of DMK authorizing the execution of this Agreement and the consummation of the Contemplated Transactions to be performed by DMK hereunder.

7.5. <u>Conversion of DMK Convertible Notes</u>. DMK shall have taken all required actions such that all DMK Convertible Notes that are outstanding immediately prior to the Closing that are being converted into shares of DMK Common Stock in connection with the Closing shall convert into shares of DMK Common Stock before the Effective Date and Closing, and no DMK Convertible Notes or other convertible securities shall be outstanding as of the Closing.

7.6. <u>FIRPTA Certificate</u>. On or no more than thirty (30) days prior to the Closing Date (or such other date as Adamis and DMK may mutually agree), DMK shall deliver to Adamis a properly executed certification, in form and substance reasonably satisfactory to Adamis, that shares of DMK are not "U.S. real property interests" in accordance with the Treasury Regulations under Sections 897 and 1445 of the Code, together with a notice to the IRS (which shall be filed by Adamis with the IRS following the Closing within the time period specified in Treasury Regulations Section 1.897-2(h)(2)) in accordance with the provisions of Treasury Regulations Section 1.897-2(h)(2).

7.7. Third Party Consents. DMK shall have obtained and delivered to Adamis all consents and approvals of third parties listed in Schedule 7.7.

7.8. <u>Securities Law Matters</u>. The issuance of the Merger Consideration Shares to the DMK Stockholders pursuant to this Agreement shall be exempt from the registration requirements of the Securities Act.

7.9. Dissenting Shares. The holders of no more than three percent (3.0)% of the outstanding shares of DMK Common Stock, including shares into which the DMK Convertible Notes are convertible, will have demanded and not lost or withdrawn, or will be eligible to demand, appraisal rights.

7.10. DMK License Agreements. All of the patents and related intellectual property rights and other rights held by the licensor party relating to the intellectual property license agreements set forth on <u>Section 2.12(a)</u> of the DMK Disclosure Schedules (the "*DMK License Agreements*") shall have been transferred, assigned and conveyed, effective as of the Closing to DMK pursuant to instruments of conveyance reasonably satisfactory to Adamis, and the DMK License Agreements shall have been terminated effective upon such transfer, assignment and conveyance.

ARTICLE VIII

ADDITIONAL CONDITIONS PRECEDENT TO OBLIGATION OF DMK

The obligations of DMK to effect the Merger and otherwise consummate the transactions to be consummated at the Closing are subject to the satisfaction or the written waiver by DMK, at or before the Closing, of each of the following conditions:

8.1. <u>Accuracy of Representations</u>. The representations and warranties of Adamis and Merger Sub contained in this Agreement shall have been true and correct as of the date of this Agreement and shall be true and correct on and as of the Closing Date with the same force and effect as if made on the Closing Date, except (A) in each case, or in the aggregate, where the failure to be true and correct would not reasonably be expected to have a Material Adverse Effect on the Combined Company, or (B) for those representations and warranties which address matters only as of a particular date (which representations shall have been true and correct, subject to the qualifications as set forth in the preceding clause (A), as of such particular date).

8.2. <u>Performance of Covenants</u>. All of the covenants and obligations in this Agreement that Adamis or Merger Sub is required to comply with or to perform at or before the Closing shall have been complied with and performed in all material respects, except where the failure to perform such covenants or obligations would not have a Material Adverse Effect on the Combined Company.

8.3. No Material Adverse Effect. From the Execution Date through the Effective Time, there shall not have occurred any Material Adverse Effect on Adams that continues as of the Effective Time and that would have a Material Adverse Effect on the Combined Company.

8.4. Documents. DMK shall have received the following documents:

(a) a certificate of Adamis executed on its behalf by the Chief Executive Officer and Chief Financial Officer of Adamis confirming that the conditions set forth in Sections 8.1, 8.2 and 8.3 have been duly satisfied;

(b) certificates of good standing (or equivalent documentation) of each of Adamis and Merger Sub in Delaware, and the various foreign jurisdictions in which it is qualified (except where the failure to have obtained such certificates would not result in a Material Adverse Effect on the Combined Company), certified charter documents, a certificate as to the incumbency of officers and the adoption of resolutions of the Boards of Directors of Adamis and Merger Sub authorizing the execution of this Agreement and the consummation of the Contemplated Transactions to be performed by Adamis and Merger Sub hereunder; and

(c) Written resignations in forms reasonably satisfactory to DMK, dated on or before the Closing Date and effective as of the Closing, executed by the directors and officers of Adamis whose names are set forth on <u>Annex 5.11</u>.

8.5. <u>Sarbanes-Oxley Certifications</u>. Neither the principal executive officer nor the principal financial officer of Adamis shall have failed to provide, with respect to any Adamis SEC Document filed (or required to be filed) with the SEC on or after the date of this Agreement, any necessary certification in the form required under Rule 13a-14 under the Exchange Act and 18 U.S.C. §1350.

8.6. Board of Directors. Adamis shall have caused the Board of Directors of Adamis to be constituted as set forth on <u>Annex 5.1</u> of this Agreement.

8.7. Officers. Each of the individuals identified on <u>Annex 5.11</u> shall have been appointed officers of Adamis as of the Effective Time.

8.8. <u>Adamis Reverse Stock Split Certificate of Amendment</u>. The Reverse Stock Split Amendment shall have become effective under the

DGCL.

8.9. SEC Reports. Adamis shall have filed with the SEC all reports required to be filed under the Securities Act or Exchange Act.

8.10. <u>Nasdaq Listing</u>. Adamis shall have (a) conducted a hearing before the Hearings Panel of the Listing Qualifications Department of Nasdaq (the "*Nasdaq Panel*"), including the submission of a plan of compliance (the "*Plan of Compliance*") setting forth Adamis' plan to regain compliance with the minimum bid price requirement set forth in Nasdaq Listing Rule 5550(a)(2) (the "*Rule*"), (b) the Nasdaq Panel shall have issued a ruling approving the Plan of Compliance, (c) the Adamis Common Stock shall have maintained its listing on Nasdaq as of the Closing Date and (d) since the effectiveness of the Reverse Stock Split, the Adamis Common Stock shall have maintained the minimum bid price required under the Rule.

ARTICLE IX TERMINATION

9.1. Termination. This Agreement may be terminated before the Effective Time (whether before or after receipt of the Required DMK Stockholder Vote or Required Adamis Stockholder Vote, unless otherwise specified below):

(a) by mutual written consent duly authorized by the Boards of Directors of Adamis and DMK;

(b) by either Adamis or DMK if the Merger shall not have been consummated by (i) June 30, 2023 (the "*Outside Date*"); provided, however, that the right to terminate this Agreement under this Section 9.1(b) shall not be available to any Party whose failure to fulfill or diligently pursue fulfillment of any material obligation under this Agreement has been a principal cause of or resulted in the failure of the Merger to occur on or before the Outside Date;

(c) by either Adamis or DMK if a court of competent jurisdiction or other Governmental Entity shall have issued a nonappealable final order, decree or ruling or taken any other nonappealable final action, in each case having the effect of permanently restraining, enjoining or otherwise prohibiting the Merger; provided, however, that neither Party may terminate this Agreement pursuant to this Section 9.1(c) unless that party first shall have used its reasonable best efforts to procure the removal, reversal, dissolution, setting aside or invalidation of any such order, decree, ruling or action;

(d) by either Adamis or DMK if (i) the Adamis Stockholders Meeting (including any adjournments and postponements thereof) shall have been held and completed and Adamis' stockholders shall have taken a final vote on the Adamis Proposals, and (ii) any of the Adamis Share Issuance Proposal or the Adamis Reverse Stock Split Proposal shall not have been approved or adopted at Adamis Stockholders Meeting (and shall not have been approved or adopted at any adjournment or postponement thereof) by the Required Adamis Stockholder Vote; provided, however, that the right to terminate this Agreement under this Section 9.1(d) shall not be available to Adamis where the failure to obtain the Required Adamis Stockholder Vote shall have been caused by the action or failure to act of Adamis and such action or failure to act constitutes a breach by Adamis of this Agreement;

occurred;

(e)

(f)

by DMK (at any time before the receipt of the Required Adamis Stockholder Vote) if an Adamis Triggering Event shall have

occurred;

by Adamis (at any time before the receipt of the Required Adamis Stockholder Vote) if a DMK Triggering Event shall have

(g) by DMK, upon a material breach of any representation, warranty, covenant or agreement on the part of Adamis or Merger Sub set forth in this Agreement, or if any representation or warranty of Adamis or Merger Sub shall have become inaccurate, in either case such that the conditions set forth in Section 8.1 or Section 8.2 would not be satisfied as of the time of such breach or as of the time such representation or warranty shall have become inaccurate, provided that if such inaccuracy in Adamis' or Merger Sub's representations and warranties or breach by Adamis or Merger Sub is curable by Adamis or Merger Sub, then this Agreement shall not terminate pursuant to this Section 9.1(g) as a result of such particular breach or inaccuracy in Merger Sub (as applicable) ceasing to exercise commercially reasonable efforts to cure such breach (it being understood that this Agreement shall not terminate pursuant to this Section 9.1(g) as a result of such particular breach or inaccuracy if such breach (it being understood that this Agreement shall not terminate pursuant to this Section 9.1(g) as a result of such particular breach or inaccuracy if such breach (it being understood that this Agreement shall not terminate pursuant to this Section 9.1(g) as a result of such particular breach or inaccuracy if such breach (it being understood that this Agreement shall not terminate pursuant to this Section 9.1(g) as a result of such particular breach or inaccuracy if such breach by Adamis or Merger Sub is cured before such terminate pursuant to this Section 9.1(g) as a result of such particular breach or inaccuracy if such breach by Adamis or Merger Sub is cured before such terminate pursuant to this Section 9.1(g) as a result of such particular breach or inaccuracy if such breach by Adamis or Merger Sub is cured before such termination becoming effective);

(h) by Adamis, upon a material breach of any representation, warranty, covenant or agreement on the part of DMK set forth in this Agreement, or if any representation or warranty of DMK shall have become inaccurate, in either case such that the conditions set forth in Section 7.1 or Section 7.2 would not be satisfied as of the time of such breach or as of the time such representation or warranty shall have become inaccurate, provided that if such inaccuracy in DMK's representations and warranties or breach by DMK is curable by DMK then this Agreement shall not terminate pursuant to this Section 9.1(h) as a result of such particular breach or inaccuracy until the earlier of (i) the Outside Date; (ii) the expiration of a thirty (30) day period commencing upon delivery of written notice from Adamis to DMK of such breach or inaccuracy; and (iii) DMK ceasing to exercise commercially reasonable efforts to cure such breach (it being understood that this Agreement shall not terminate pursuant to this Section 9.1(h) as a result of such particular breach or before such termination becoming effective); or

(i) by Adamis in accordance with the terms and subject to the conditions of Section 4.5(b)(ii).

9.2. <u>Effect of Termination</u>. In the event of the termination of this Agreement as provided in Section 9.1, this Agreement shall be of no further force or effect; provided, however, that (i) this Section 9.2, Section 9.3, and Section 10 and the Confidentiality Agreement shall survive the termination of this Agreement and shall remain in full force and effect, and (ii) the termination of this Agreement shall not relieve any Party from any liability for any breach of any representation, warranty, covenant, obligation or other provision contained in this Agreement.

9.3. <u>Expenses</u>.

(a) Except as set forth in this Section 9.3, all fees and expenses incurred in connection with this Agreement and the Contemplated Transactions shall be paid by the Party incurring such expenses, whether or not the Merger is consummated.

(b) Notwithstanding the above, Adamis agrees to pay or reimburse DMK for the amount of the actual out-of-pocket legal and accounting expenses incurred by DMK in connection with the negotiation and entering into of this Agreement and the Contemplated Transactions (and the DMK financial statements required to be included in the Adamis Proxy Statement) and the consummation of the transactions contemplated by this Agreement (the "*Expense Payment Amount*"). DMK will provide, or arrange for its counsel and independent accounting firm to provide, to Adamis copies of invoices or other customary documentation relating to such expenses. Adamis shall make such payments and reimbursements promptly upon receipt of such invoices or other customary documentation and, in any case, no later than fifteen (15) days after receipt thereof.

ARTICLE X MISCELLANEOUS PROVISIONS

10.1. <u>Non-Survival of Representations and Warranties</u>. The representations and warranties of DMK, Merger Sub and Adamis contained in this Agreement or any certificate or instrument delivered pursuant to this Agreement shall terminate at the Effective Time, and only the covenants that by their terms survive the Effective Time and this Article X shall survive the Effective Time.

10.2. <u>Amendment</u>. This Agreement may be amended with the approval of the respective Boards of Directors of DMK and Adamis at any time (whether before or after the receipt of the Required DMK Stockholder Vote or Required Adamis Stockholder Vote); provided, however, that after any such adoption and approval of this Agreement by a Party's stockholders, no amendment shall be made which by law requires further approval of the stockholders of such Party without the further approval of such stockholders. This Agreement may not be amended, except by an instrument in writing signed on behalf of each of DMK and Adamis.

10.3. <u>Waiver</u>.

(a) No failure on the part of any Party to exercise any power, right, privilege or remedy under this Agreement, and no delay on the part of any Party in exercising any power, right, privilege or remedy under this Agreement, shall operate as a waiver of such power, right, privilege or remedy; and no single or partial exercise of any such power, right, privilege or remedy shall preclude any other or further exercise thereof or of any other power, right, privilege or remedy.

(b) No Party shall be deemed to have waived any claim arising out of this Agreement, or any power, right, privilege or remedy under this Agreement, unless the waiver of such claim, power, right, privilege or remedy is expressly set forth in a written instrument duly executed and delivered on behalf of such Party; and any such waiver shall not be applicable or have any effect, except in the specific instance in which it is given.

10.4. Entire Agreement; Counterparts; Exchanges by Facsimile. This Agreement and the other agreements referred to in this Agreement constitute the entire agreement and supersede all prior agreements and understandings, both written and oral, among or between any of the Parties with respect to the subject matter hereof and thereof; provided, however, that the Confidentiality Agreement shall not be superseded and shall remain in full force and effect in accordance with its terms. This Agreement may be executed by means of electronic signatures and electronic transmission, and in several counterparts, each of which shall be deemed an original and all of which shall constitute one and the same instrument. The exchange of a fully executed Agreement (in counterparts or otherwise) by all Parties by facsimile shall be sufficient to bind the Parties to the terms and conditions of this Agreement.

10.5. <u>Applicable Law; Jurisdiction</u>. This Agreement shall be governed by, and construed in accordance with, the laws of the State of Delaware, except to the extent that the NJBCA mandatorily applies to matters relating to the Merger, regardless of the laws that might otherwise govern under applicable principles of conflicts of laws. In any action or suit between any of the Parties arising out of or relating to this Agreement or any of the Contemplated Transactions: (a) each of the Parties irrevocably and unconditionally consents and submits to the exclusive jurisdiction and venue of the state and federal courts located in the State of Delaware; (b) if any such action or suit is commenced in a state court, then, subject to applicable Law, no Party shall object to the removal of such action or suit to any federal court located in Delaware, and (c) the Parties agree that service of progress may be made in the manner provided for in this Agreement for delivery of notices.

10.6. <u>Waiver of Jury Trial</u>. EACH PARTY TO THIS AGREEMENT HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OF THE TRANSACTIONS CONTEMPLATED HEREBY. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) IT UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) IT MAKES THIS WAIVER VOLUNTARILY, AND (D) IT HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 10.6.

10.7. Notices. Any notice or other communication required or permitted to be delivered to any party under this Agreement shall be in writing and shall be given by means of hand delivery, registered mail, courier or express delivery service, or facsimile. Notices shall be deemed delivered and received (i) upon delivery by hand, (ii) three (3) Business Days after deposit in the U.S. mails, certified or registered mail, (iii) one (1) Business Day after delivery to a reputable overnight courier service for next business-day delivery (with confirmation of delivery), or (iv) one (1) Business Day after transmission by email, or other electronic transmission, to the number set forth beneath the name of such party below (or to such other address or facsimile telephone number as such party shall have specified in written notice given to the other Parties here), with confirmation of successful transmission:

if to Adamis:

Adamis Pharmaceuticals Corporation 11682 El Camino Real, Suite 300 San Diego, CA 92130 Attention: Chief Executive Officer Telephone No.: (858) 997-2400

with a copy to:

C. Kevin Kelso, Esq. Weintraub Tobin 400 Capitol Mall, 11th Floor Sacramento, CA 95814 Telephone No.: (916) 558-6000 Facsimile No. (916) 446-1611 Email: <u>kkelso@weintraub.com</u>

if to DMK:

DMK Pharmaceuticals Corporation 50 Division Street, Suite 501 Somerville, NJ 08876 Attention: Chief Executive Officer Telephone No.: (908) 698-2390 Email: <u>eversi@DMKPharma.com</u>

Nelson Mullins Glenlake One, Suite 200 4140 Parklake Avenue Raleigh, NC 27612 Attention: David Mannheim, Esq

| Telephone No.: | (919) 329-3804 |
|----------------|----------------------------------|
| Facsimile No.: | (919) 329-3799 |
| Email: | david.mannheim@nelsonmullins.com |

10.8. <u>**Cooperation**</u>. Each Party agrees to cooperate fully with the other Party and to execute and deliver such further documents, certificates, agreements and instruments and to take such other actions as may be reasonably requested by the other Party to evidence or reflect the Contemplated Transactions and to carry out the intent and purposes of this Agreement.

10.9. <u>Severability</u>. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions of this Agreement or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If a final judgment of a court of competent jurisdiction declares that any term or provision of this Agreement is invalid or unenforceable, the Parties hereto agree that the court making such determination.

10.10. <u>Other Remedies; Specific Performance</u>. Except as otherwise provided herein, any and all remedies herein expressly conferred upon a Party will be deemed cumulative with and not exclusive of any other remedy conferred hereby, or by law or equity upon such Party, and the exercise by a Party of any one remedy will not preclude the exercise of any other remedy. The Parties hereto agree that irreparable damage would occur in the event that any of the material provisions of the Confidentiality Agreement, as modified by Section 5.10 of this Agreement, were not performed in material respects or were otherwise breached in material respects (a "*Confidentiality Action*"). It is accordingly agreed that the Parties shall be entitled to seek an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court of the United States or any state having jurisdiction, this being the addition to any other remedy to which they are entitled at law or in equity, without, solely with respect to a Confidentiality Action, the necessity of proving actual damages and without posting bond or other security.

10.11. Conflict Waiver; Attorney-Client Privilege.

(a) Each of the Parties hereto acknowledges and agrees, on its own behalf and on behalf of its directors, managers, members, shareholders, partners, officers, employees, and Affiliates, that (i) Nelson Mullins Riley & Scarborough LLP has acted as counsel to the DMK Stockholders and DMK, in connection with the negotiation, preparation, execution, and delivery of this Agreement and the Ancillary Agreements and the consummation of the transactions contemplated hereby and thereby. Adamis agrees, and shall cause the Surviving Corporation to agree, that, following consummation of the transactions contemplated hereby, such representation and any prior representation of the Company by Nelson Mullins Riley & Scarborough LLP (or any successor) (the "DMK Law Firm") shall not preclude DMK Law Firm from serving as counsel to the DMK Stockholders or any director, manager, member, shareholder, partner, officer, or employee of DMK and the DMK Stockholders, in connection with any litigation, claim, or obligation arising out of or relating to this Agreement or the transactions contemplated hereby and (ii) Adamis shall not, and shall cause the Surviving Corporation not to, seek or have DMK Law Firm disqualified from any such representation based on the prior representation of DMK by DMK Law Firm. Each of the Parties hereto hereby consents thereto and waives any conflict of interest arising from such prior representation, and each of such Parties shall cause any of its Affiliates to consent to waive any conflict of interest arising from such representation. Each of the Parties acknowledges that such consent and waiver is voluntary, that it has been carefully considered, and that the Parties have consulted with counsel or have been advised they should do so in connection herewith. The covenants, consent, and waiver contained in this Section 10.11(a) shall not be deemed exclusive of any other rights to which DMK Law Firm is entitled whether pursuant to law, contract, or otherwise.



All communications prior to Closing between or among any of DMK Stockholders and DMK, on the one hand, and DMK Law (b) Firm, on the other hand, relating to the negotiation, preparation, execution, and delivery of this Agreement and the Ancillary Agreements and the consummation of the transactions contemplated hereby and thereby (the "Privileged Communications") shall be deemed to be attorney-client privileged and the expectation of client confidence relating thereto shall survive Closing, and from and after Closing shall belong solely to the DMK Stockholders and shall not pass to or be claimed by Adamis or the Surviving Corporation. Accordingly, Adamis and the Surviving Corporation shall not have access to any Privileged Communications or to the files of DMK Law Firm relating to such engagement from and after Closing. Without limiting the generality of the foregoing, from and after the Closing, (i) the DMK Stockholders (and not Adamis or the Surviving Corporation) shall be the sole holders of the attorneyclient privilege with respect to such engagement, and none of Adamis nor the Surviving Corporation shall be a holder thereof, (ii) to the extent that files of DMK Law Firm in respect of such engagement constitute property of the client, only the DMK Stockholders (and not Adamis nor the Surviving Corporation) shall hold such property rights and (iii) DMK Law Firm shall have no duty whatsoever to reveal or disclose any such attorney-client communications or files to Adamis or the Surviving Corporation by reason of any attorney-client relationship between DMK Law Firm and the Surviving Corporation or otherwise. Notwithstanding the foregoing, in the event that after Closing a dispute arises between Adamis or its Affiliates (including the Surviving Corporation), on the one hand, and a third party other than any of the DMK Stockholders, on the other hand, Adamis and its Affiliates (including the Surviving Corporation) may assert the attorney-client privilege to prevent disclosure of confidential communications to such third party; provided, however, that neither Adamis nor any of its Affiliates (including the Surviving Corporation) may waive such privilege without the prior written consent of the DMK Stockholders representing the Required DMK Stockholder Vote. In the event that Adamis or any of its Affiliates (including the Surviving Corporation) is legally required by Governmental Order or otherwise legally required to access or obtain a copy of all or a portion of the Privileged Communications, to the extent (x) permitted by applicable Law, and (y) advisable in the opinion of Adamis' counsel, then Adamis shall immediately (and, in any event, within ten (10) days) notify the DMK Stockholders in writing so that the DMK Stockholders can seek a protective order. In furtherance of the foregoing, each of the Parties agrees that (i) no waiver is intended by failing to remove all Privileged Communications from the Surviving Corporation's files and computer systems, and (ii) after Closing the Parties hereto will use commercially reasonable efforts to take the steps necessary to ensure the Privileged Communications are held and controlled by the DMK Stockholders. Adamis agrees that after Closing none of Adamis, the Surviving Corporation, nor their Affiliates will (i) access or review the Privileged Communications in connection with any action, litigation, claim, or dispute against or involving any of the DMK Stockholders or any of their Affiliates or (ii) use or assert the Privileged Communications against any member of the DMK Stockholders or any of their Affiliates in any action, litigation, claim, or dispute against or involving any of the DMK Stockholders or any of their Affiliates.

(c) This <u>Section 10.11</u> is intended for the benefit of, and shall be enforceable by, DMK Law Firm. This <u>Section 10.11</u> shall be irrevocable, and no term of this <u>Section 10.11</u> may be amended, waived, or modified, without the prior written consent of DMK Law Firm.

(d) By virtue of the approval and adoption of this Agreement by the requisite vote of the DMK Stockholders, each DMK Stockholder (other than a Dissenting Stockholder) agrees to the provisions of this section:

Each DMK Stockholder (other than a Dissenting Stockholder) hereby appoints Dr. Ebrahim Versi as the Stockholder's (i) representative (the "Stockholders' Representative") to act as the authorized representative of the Stockholders with respect to all matters relating to this Section 10.11 under or in connection with this Section 10.11 requiring any action or decision by any of the Stockholders. The Stockholders' Representative shall thereupon be authorized to serve as agent and attorney-in-fact for and on behalf of each Stockholder to (i) execute and deliver for and on behalf of any or all Stockholders all agreements, instruments and documents necessary or desirable to carry out the intent of this Section and any other documents, instruments and/or agreements contemplated hereby and thereby, (ii) make any claim following the Closing against Adamis on behalf of the Stockholders (or any of them) relating to this Section, (iii) act as the Stockholders' exclusive agent for the receipt of any notice of claim by Adamis, pursuant to this Section, (iv) undertake the defense or settlement of any claim relating to this Section, (v) give and receive on behalf of the Stockholders, or any of them, any and all other notices from or to any other Stockholder under this Section, and (vi) and to take all other actions that are necessary or appropriate in the reasonable judgment of the Stockholders' Representative for the accomplishment of the foregoing. The Stockholders' Representative may, from time to time, be replaced by the Stockholders upon the written approval of the holders representing a majority in interest of the Stockholders based on their respective ownership of DMK Capital Stock prior to the Closing. The Stockholders' Representative may resign at any time by providing written notice of intent to resign to the Stockholders, which resignation shall be effective upon the earlier of (A) thirty (30) days following delivery of such written notice, or (B) the appointment of a successor by the holders of a majority in interest of the Stockholders based on their respective ownership of DMK Capital Stock prior the Closing.

(ii) The Stockholders' Representative shall not be liable for any act done or omitted hereunder as Stockholders' Representative except in the case of gross negligence, bad faith or willful misconduct by the Stockholders' Representative. The Stockholders' Representative may in good faith rely conclusively upon information, reports, statements and opinions prepared or presented by professionals, and any action taken by the Stockholders' Representative based on such reliance shall be deemed conclusively to have been taken in good faith and in the exercise of reasonable judgment. A decision, act, consent, instruction or agreement of the Stockholders' Representative shall constitute a decision or waiver of this Section, act (or election not to act), agreement or deemed agreement of the Stockholders' Representative shall constitute a decision, act, consent, instruction or agreement of the Stockholders and shall be final, binding and conclusive upon the Stockholders.

(iii) Each party hereto and each DMK Stockholder shall be entitled to rely exclusively upon any communication or instruction given or other action taken by the Stockholders' Representative on behalf of the Stockholders pursuant to this Section, and shall not be liable for any action taken or not taken in good faith reliance on a communication or other instruction from the Stockholders' Representative on behalf of the Stockholders (or any of them). The Stockholders Representative shall promptly notify Adamis of any replacement of the Stockholders' Representative. Adamis shall be entitled to assume that any such notice received by them is valid and correct, without any duty or obligation to investigate whether such replacement Stockholders' Representative was properly appointed.

10.12. Construction.

(a) For purposes of this Agreement, whenever the context requires: the singular number shall include the plural, and vice versa; the masculine gender shall include the feminine and neuter genders; the feminine gender shall include the masculine and neuter genders; and the neuter gender shall include masculine and feminine genders.

(b) The Parties hereto agree that any rule of construction to the effect that ambiguities are to be resolved against the drafting Party shall not be applied in the construction or interpretation of this Agreement.

(c) As used in this Agreement, the words "include" and "including," and variations thereof, shall not be deemed to be terms of limitation, but rather shall be deemed to be followed by the words "without limitation."

(d) Except as otherwise indicated, all references in this Agreement to "Sections," "Exhibits" and "Schedules" are intended to refer to Sections of this Agreement and Exhibits and Schedules to this Agreement.

(e) The bold-faced headings contained in this Agreement are for convenience of reference only, shall not be deemed to be a part of this Agreement and shall not be referred to in connection with the construction or interpretation of this Agreement.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the Parties have caused this Agreement and Plan of Merger and Reorganization to be executed as of the date first above written.

ADAMIS PHARMACEUTICALS CORPORATION

| By: | /s/ David J. Marguglio |
|--------|-------------------------|
| Name: | David J. Marguglio |
| Title: | Chief Executive Officer |

AARDVARK MERGER SUB, INC.

| By: | /s/ David J. Marguglio |
|--------|-------------------------|
| Name: | David J. Marguglio |
| Title: | Chief Executive Officer |

_

[Signature Page to Agreement and Plan of Merger and Reorganization]

DMK PHARMACEUTICALS CORPORATION

By: /s/ Ebrahim Versi Name: Ebrahim Versi

Title: Chief Executive Officer

[Signature Page to Agreement and Plan of Merger and Reorganization]

EXHIBIT A

CERTAIN DEFINITIONS

For purposes of this Agreement:

"Acquisition Inquiry" shall mean, with respect to a Party, an inquiry, indication of interest or request for information (other than an inquiry, indication of interest or request for information made or submitted by DMK or its Representatives, on the one hand or Adamis or its Representatives, on the other hand, to the other Party) that would reasonably be expected to lead to an Acquisition Proposal from such Party.

"*Acquisition Proposal*" shall mean any offer or proposal (other than an offer or proposal made or submitted by DMK, on the one hand or Adamis, on the other hand to the other Party) contemplating or otherwise relating to any Acquisition Transaction with such Party.

"Acquisition Transaction" shall mean any transaction or series of transactions (except for the Contemplated Transactions) involving:

(a) any direct or indirect acquisition of assets of a Party (including any voting equity interests of a Party's subsidiaries) equal to 50% or more of the fair market value of such Party's consolidated assets or to which 50% or more of a Party's net revenues or net income on a consolidated basis are attributable;

(b) direct or indirect acquisition of 50% or more of the shares of Adamis common stock other than in a transaction primarily for purposes of financing;

(c) tender offer or exchange offer that if consummated would result in any person beneficially owning 50% or more of the shares of a Party's common stock; or

(d) merger, consolidation, other business combination or similar transaction involving a Party or any of its Subsidiaries, pursuant to which the other party to the transaction or its affiliates or shareholders would own 51% or more of the consolidated assets, net revenues or net income of the Party and its Subsidiaries, taken as a whole; in all such cases where such transaction is to be entered into with any person or group of persons other than a Party or its affiliates.

(e) Notwithstanding the foregoing, with respect to Adamis, "Acquisition Transaction" shall not include discussions or agreements regarding or with respect to, or consummation of, one or more transactions regarding or with respect to: (i) sales or other dispositions of assets of Adamis relating to the compounding pharmaceuticals business formerly conducted by Adamis or its Subsidiaries, (ii) commercial, license or other agreements relating to the sale, license or distribution of one or more of Adamis' FDA approved commercial drug products; (iii) modifications to existing outstanding securities, (iv) issuance of equity or debt securities or agreements in transactions primarily for purposes of financing; or (iv) bankruptcy, dissolution or liquidation proceedings, and no such agreement or transaction shall be deemed to be an "Acquisition Transaction."

"*Agreement*" shall mean the Agreement and Plan of Merger and Reorganization to which this <u>Exhibit A</u> is attached, as it may be amended from time to time.

"Adamis Bylaws" shall mean the bylaws of Adamis as currently in effect.

"Adamis Employee Agreement" shall mean each management, employment, severance, consulting, relocation, repatriation or expatriation agreement or other contract between Adamis and any current employee thereof, other than any such management, employment, severance, consulting, relocation, repatriation or expatriation agreement or other contract with such employee which is terminable "at will" without any obligation on the part of Adamis to make any payments or provide any benefits in connection with such termination.

"*Adamis' Knowledge*" shall mean (a) the actual knowledge, after reasonable diligence, of Adamis' officers and directors, and (b) such facts and circumstances each of the officers and directors of Adamis should have known given his involvement in Adamis and the information available to him.

"Adamis Material Contract" means a Material Contract to which Adamis or any of its Subsidiaries is a party that as of the date of this Agreement is a material contract as defined in Item 601(b)(10) of Regulation S-K as promulgated under the Securities Act.

"Adamis Options" shall mean options or other rights to purchase or acquire shares of Adamis Common Stock issued by Adamis.

"Adamis Preferred Stock" shall mean shares of preferred stock, par value \$0.0001 per share, of Adamis.

"Adamis Product" shall mean those products, compounds, proteins or other biological materials that are under research or development, manufactured, tested, the subject of trials or studies, distributed and/or marketed by or on behalf of Adamis (and for purposes of clarification shall not include any product or compound formerly manufactured, marketed or sold by Adamis' U.S. Compounding, Inc. subsidiary.

"Adamis Share Issuance Proposal" means the Proposal that Adamis will include in its Proxy Statement relating to the Adamis Stockholders Meeting to approve the issuance of shares of Adamis Common Stock and Series E Preferred to the securityholders of DMK pursuant to the Merger Agreement and the Merger.

"Adamis Stock Plan" shall mean the Adamis 2020 Equity Incentive Plan.

"Adamis Stock Plans" shall mean the Adamis 2009 Equity Incentive Plan and the 2020 Equity Incentive Plan, considered together.

"Adamis Triggering Event" shall be deemed to have occurred if: (i) there shall have occurred a Change in the Adamis Board Recommendation; (ii) Adamis shall have failed to convene the Adamis Stockholders Meeting within sixty (60) days after the definitive Proxy Statement is filed with the SEC (other than to the extent that Adamis determines, in good faith, that the Required Adamis Stockholder Vote will not be obtained at a meeting held within such time, in such case the sixty (60) day period shall be tolled until such time as Adamis determines, in good faith, that the Required Adamis determines, in good faith, that the Required Adamis or any of its Representatives shall have failed to comply with the provisions set forth in Section 4.5 of the Agreement in any material respect, or (iv) Adamis shall have delivered a Notice of Superior Proposal under Section 4.5(b).

"Alternative Transaction" means any (a) direct or indirect acquisition of assets of a Party or any of its subsidiaries (including any voting equity interests of a Party's subsidiaries) equal to 50% or more of the fair market value of such Party's consolidated assets or to which 50% or more of a Party's net revenues or net income on a consolidated basis are attributable, (b) direct or indirect acquisition of 50% or more of the shares of Adamis common stock other than in a transaction primarily for purposes of financing, (c) tender offer or exchange offer that if consummated would result in any person beneficially owning 50% or more of the shares of a Party's common stock, or (d) merger, consolidation, other business combination or similar transaction involving a Party or any of its subsidiaries, pursuant to which the other party to the transaction or its affiliates or shareholders would own 51% or more of the consolidated assets, net revenues or net income of the Party and its Subsidiaries, taken as a whole; in all such cases where such transaction is to be entered into with any person or group of persons other than a Party or its affiliates. Notwithstanding the foregoing, with respect to Adamis, "Alternative Transaction" shall not include discussions or agreements regarding or with respect to, or consummation of one or more transactions regarding or with respect to: (i) sales or other dispositions of assets of Adamis relating to the compounding pharmaceuticals business formerly conducted by Adamis or its Subsidiaries, (ii) commercial, license or other agreements relating to the sale, license or distribution of one or more of Adamis' FDA approved commercial drug products; (iii) modifications to existing outstanding securities, (iv) issuance of equity or debt securities or agreements in transactions primarily for purposes of financing; or (iv) bankruptcy, dissolution or liquidation proceedings, and no such agreement or transaction shall be deemed to be an "Alternative Transaction."

"Business" shall mean the business and operations of a party.

"Business Day" shall mean any day other than a day on which banks in the State of California are authorized or obligated to be closed.

"Code" shall mean the Internal Revenue Code of 1986, as amended.

"*Combined Company*" shall mean Adamis and DMK and their respective Subsidiaries (and, after the Closing, the Surviving Corporation), taken together as a whole.

"Consent" shall mean any approval, consent, ratification, permission, waiver or authorization (including any necessary Governmental Authorization).

"Contemplated Transactions" shall mean the Merger and the other transactions and actions expressly contemplated by the Agreement.

"*Contract*" shall, with respect to any Person, mean any written, oral or other agreement, contract, subcontract, lease (whether real or personal property), mortgage, understanding, arrangement, instrument, note, option, warranty, purchase order, license, sublicense, insurance policy, benefit plan or legally binding commitment or undertaking of any nature to which such Person is a party or by which such Person or any of its assets are bound or affected under applicable law.

"*Convertible Securities*" shall mean and include options, warrants and other rights for the purchase of common stock or any stock or security convertible into or exchangeable for common stock.

"DGCL" shall mean the General Corporation Law of the State of Delaware.

"DMK" shall have the meaning set forth in the Background section.

"DMK Board" shall have the meaning set forth in the Background section.

"DMK Bylaws" shall mean the bylaws of DMK as currently in effect.

"DMK Capital Stock" shall mean shares of DMK Common Stock and, if any, DMK Preferred Stock.

"DMK Charter" shall mean the certificate of incorporation of DMK, as in effect on the date of this Agreement.

"*DMK's Knowledge*" shall mean (a) the actual knowledge, after reasonable diligence, of Ebrahim Versi, and (b) such facts and circumstances Ebrahim Versi should have known given his involvement in DMK and the information available to him.

"*DMK Options*" shall mean all options, warrants or other rights, if any, that may be outstanding to purchase, acquire or otherwise receive shares of DMK Capital Stock (whether or not vested) held by current or former employees or directors of or consultants to DMK.

"*DMK Product*" shall mean those products, compounds, proteins or other biological materials that are under research or development, manufactured, tested, the subject of trials or studies, distributed and/or marketed by or on behalf of DMK.

"DMK Preferred Stock" shall mean shares of preferred stock of DMK.

"*DMK Registered IP*" means all DMK Patent and Proprietary rights or other intellectual property rights that are owned by or exclusively licensed to DMK that are registered, filed or issued under the authority of, with or by any Governmental Body, including all patents, registered copyrights and registered trademarks and all applications for any of the foregoing.

"DMK Stock Plan" means the DMK 2016 Stock Plan.

"DMK Stockholder" shall mean each holder of any DMK Capital Stock immediately before the Effective Time.

"*DMK Triggering Event*" shall be deemed to have occurred if (a) DMK or any of its Representatives shall have failed to comply with the provisions set forth in Section 4.5 of the Agreement in any material respect, or (b) DMK or any of its Representatives shall change the DMK Board Recommendation or not convene the DMK Stockholders Meeting (or obtain the Required DMK Stockholder Vote by written consent).

"*Encumbrances*" shall mean any lien, pledge, hypothecation, charge, mortgage, security interest, encumbrance, claim, infringement, interference, option, right of first refusal, preemptive right, community property interest or restriction of any nature (including any restriction on the voting of any security, any restriction on the transfer of any security or other asset, any restriction on the receipt of any income derived from any asset, any restriction on the use of any asset and any restriction on the possession, exercise or transfer of any other attribute of ownership of any asset).

"*Environment*" shall mean soil, land surface or subsurface strata, surface waters (including navigable waters, ocean waters, streams, ponds, drainage basins, and wetlands), ground waters, drinking water supply, stream sediments, ambient air (including indoor air), plant and animal life, and any other environmental medium or natural resource.

"*Environmental, Health, and Safety Liabilities*" shall mean any cost, damages, expense, liability, obligation, or other responsibility arising out of any Environmental Law or Occupational Safety and Health Law and consisting of or relating to:

(a) any fines, penalties, judgments, awards, settlements, legal or administrative Legal Proceedings, damages, losses, claims, demands and response, investigative, remedial, compliance, corrective or inspection costs and expenses arising under Environmental Law or Occupational Safety and Health Law (including on-site or off-site contamination, occupational safety and health, and regulation of chemical substances or products); or

(b) financial responsibility under Environmental Law or Occupational Safety and Health Law for cleanup costs or corrective action, including any investigation, cleanup, removal, containment, or other remediation or response actions ("*Cleanup*") required by applicable Environmental Law or Occupational Safety and Health Law (whether or not such Cleanup has been required or requested by any Governmental Entity or any other Person) and for any natural resource damages. The terms "removal," "remedial," and "response action," include the types of activities covered by the United States Comprehensive Environmental Response, Compensation, and Liability Act, 42 U.S.C. Section 9601 et seq., as amended ("*CERCLA*").

"Environmental Law" shall mean all federal, state and local laws, statutes, regulations, ordinances, codes, rules and other governmental restrictions and requirements relating to the discharge of air pollutants, water pollutants or processed waste water or otherwise relating in any manner to the environment, pollutants or hazardous substances or materials, including but not limited to the Federal Solid Waste Disposal Act; the Federal Clean Air Act including, without limitation, the Clean Air Act Amendments of 1990; the Federal Water Pollution Control Act; the Hazardous Materials Transportation Act; the Federal Toxic Substances Control Act; the Federal Resource Conservation and Recovery Act of 1976; the National Environmental Policy Act; CERCLA; all amendments to any of the foregoing statutes, and all regulations promulgated by any federal or state agencies, including the Environmental Protection Agency, regulations of the Nuclear Regulatory Agency, and regulations of any state department of natural resources or state environmental protection agency previously, now or at any time hereafter in effect.

"Exchange Act" shall mean the Securities Exchange Act of 1934, as amended.

"*Facilities*" shall mean any real property, leaseholds, or other interests currently or formerly owned or operated by a Party and any buildings, plants, structures, or equipment (including motor vehicles, tank cars, and rolling stock) currently or formerly owned or operated by any Party.

"FDA" shall mean the U.S. Food and Drug Administration.

"FDCA" shall mean the Federal Food, Drug and Cosmetic Act and the regulations thereunder.

"Form 8-K Amendment" means the amendment to the Form 8-K that Adamis will be required to file with the SEC after the Closing and the Effective Time pursuant to Items 2.01 and 9.01 of Form 8-K, including in order to file certain financial information.

"GAAP" shall mean United States generally accepted accounting principles.

"Governmental Authorization" shall mean any: (a) permit, license, certificate, franchise, grant, funding arrangement, permission, variance, clearance, registration, qualification, approval or authorization issued, granted, given or otherwise made available by or under the authority of any Governmental Entity or pursuant to any applicable Law; or (b) right under any Contract with any Governmental Entity.

"*Governmental Entity*" shall mean any: (a) nation, state, commonwealth, province, territory, county, municipality, district or other jurisdiction of any nature; (b) federal, state, local, municipal, foreign or other government; or (c) governmental or quasi-governmental authority of any nature (including any governmental division, department, agency, commission, instrumentality, official, ministry, fund, foundation, center, organization, unit, body or Entity and any court or other tribunal, and for the avoidance of doubt, any Taxing authority).

"*Hazardous Materials*" shall mean any pollutant, chemical, substance and any toxic, infectious, carcinogenic, reactive, corrosive, ignitable or flammable chemical, or chemical compound, or hazardous substance, material or waste, whether solid, liquid or gas, that is subject to regulation, control or remediation under any Environmental Law Requirement, including, without limitation, crude oil or any fraction thereof, and petroleum products or by-products.

"Intellectual Property" shall mean all domestic and foreign intellectual property and proprietary rights, including but not limited to all (a) inventions (whether or not patentable and whether or not reduced to practice), all improvements thereto, and all patents and patent applications, (b) trademarks, service marks, trade names, domain names, trade dress, logos, corporate names and brand names, together will all goodwill associated therewith, and all applications and registrations in connection therewith, (c) all works of authorship (whether or not published), copyrights and designs, and all applications and registrations in connection therewith, (d) source code and object code versions of computer software (including data and related documentation) and website content, and (e) trade secrets and confidential business information (including ideas, know-how, formulas, compositions, processes and techniques, research and development information, technical data, designs, drawings, specifications, research records, records of inventions, test information, financial, marketing and business data, pricing and cost information, business and marketing plans and proposals and customer and supplier lists and information, including all membership lists and databases and related information and profiles).

"IRS" shall mean the United States Internal Revenue Service.

"*Law*" means any federal, state, national, supra-national, foreign, local or municipal or other law, statute, constitution, principle of common law, resolution, ordinance, code, edict, decree, rule, regulation, ruling or requirement issued, enacted, adopted, promulgated, implemented or otherwise put into effect by or under the authority of any Governmental Entity (including under the authority of Nasdaq or the Financial Industry Regulatory Authority).

"*Legal Proceeding*" shall mean any action, suit, litigation, arbitration, proceeding (including any civil, criminal, administrative, investigative or appellate proceeding), hearing, inquiry, audit, examination or investigation commenced, brought, conducted or heard by or before, or otherwise involving, any court or other Governmental Entity or any arbitrator or arbitration panel.

"*Material Adverse Effect*" shall mean any fact, change, event, factor, condition, circumstance, development or effect that, individually or in the aggregate, has, or would reasonably be expected to have, a material adverse effect on the business, assets, liabilities, condition (financial or otherwise) or results of operations of a Party (including, following the Merger, the Surviving Corporation and its Subsidiaries), taken as a whole, other than to the extent such effects are due to: (i) the announcement of the transactions contemplated by this Agreement; (ii) economic factors affecting the national, regional or world economy; (iii) any act or threat of terrorism or war anywhere in the world, any armed hostilities or terrorist activities anywhere in the world or any governmental or other response or reaction to any of the foregoing; (iv) factors generally affecting the industry or market in which a Party operates; (v) changes in law, rules or regulations applicable to a Party; (vi) changes in GAAP or the interpretation thereof, in each case to the extent required by GAAP; (vii) the Reverse Stock Split; (viii) any change in the stock price or trading volume of Adamis Common Stock (it being understood that the facts and circumstances giving rise to such change may be deemed to constitute, and may be taken into account in determining whether there has been, a Material Adverse Effect if such facts and circumstances are not otherwise excluded by clauses (i) – (vii) of this definition); or (ix) any failure of DMK to obtain additional financing or commitments for additional financing before the closing date.

"*Material Contract*" shall mean, with respect to DMK, any agreement, instrument or document now in effect (including any amendment to any of the foregoing) to which DMK is a party or by which such Party's assets are bound:

(a) with any director, or executive officer of DMK, including any Contract that would give rise to or otherwise result in proxy statement disclosure pursuant to Item 404 of Regulation S-K (assuming that DMK was subject to the requirements of the Exchange Act);

(b) evidencing, governing or relating to indebtedness for borrowed money or which provides for the imposition of any material lien on any of DMK's assets;

(c) that involves expenditures or receipts in excess of \$50,000 per annum;

(d) that in any material way purports to restrict the business activity of DMK or limit the freedom of DMK to engage in any line of business or to compete with any Person;

(e) other than customary employee offer letters, relating to the employment of, or the performance of services by, any employee or consultant; or pursuant to which a party is or may become obligated to make any severance, termination or similar payment to any employee or director; or pursuant to which a party is or may become obligated to make any bonus or similar payment (other than payments constituting base salary) to any employee or director;

(f) (i) relating to the acquisition, issuance, voting, registration, sale or transfer of any securities of a party, (ii) providing any Person with any preemptive right, right of participation, right of maintenance or any similar right with respect to any securities of DMK, or (C) providing a Person with any right of first refusal with respect to, or right to repurchase or redeem, any securities, except for Contracts pursuant to the DMK Stock Plan and Contracts between DMK and any Person that provide a right of first refusal, right of repurchase or cancellation or similar right in favor of DMK;

(g) relating to any DMK guaranty or any indemnity or similar obligation;

(h) (i) imposing any confidentiality obligation on DMK, or (ii) containing "standstill" provisions, in each case other than under agreements entered into in the Ordinary Course of Business that are not material to DMK;

(i) (i) to which any Governmental Entity is a party or under which any Governmental Entity has any rights or obligations, or (ii) directly or indirectly benefiting any Governmental Entity (including any subcontract or other Contract between DMK and any contractor or subcontractor to any Governmental Entity), or (iii) relating to any funding, grant or similar agreement, proposal or commitment relating to product or product candidate of DMK, or pursuant to which any funding, grant or similar funding is or has been provided to DMK, whether or not from a Governmental Entity or from a different third party;

(j) relating to any equity or convertible debt funding of DMK;

(k) relating to the license to DMK of any intellectual property rights other than with respect to "off-the-shelf" or "shrink-wrap" licenses used by DMK; and

(l) that if terminated or breached would reasonably be expected to have a Material Adverse Effect on DMK or on any of the transactions contemplated by this Agreement or any of the Ancillary Agreements.

"Ordinary Course of Business" shall mean, in the case of each of DMK and Adamis, such actions taken in the ordinary course of its normal operations and consistent with its past practices.

"*Party*" or "*Parties*" shall mean DMK, Merger Sub and Adamis.

"Person" shall mean any individual, Entity or Governmental Entity.

"Pre-Reverse Stock Split Adamis Shares" shall mean shares of Adamis Common Stock before the effective time of the Reverse Stock Split.

"*Proxy Statement*" shall mean the Proxy Statement to be filed with the SEC by Adamis in connection with the Merger, as said statement may be amended, and mailed to the Adamis stockholders in connection with Adamis Stockholders Meeting.

"Representatives" shall mean officers, directors, employees, attorneys, accountants, investment bankers, advisors and representatives.

"*Required Adamis Stockholder Vote*" shall mean the vote of the Adamis stockholders that is required under the DGCL or other applicable law to approve the Adamis Proposals.

"*Required DMK Stockholder Vote*" shall mean the vote or written consent of the DMK Stockholders that is required under applicable law to approve the Merger and the transactions contemplated by this Agreement.

"*Reverse Stock Split Ratio*" shall mean the number of shares of Pre-Reverse Stock Split Adamis Shares that are combined and converted into one share of New Adamis Common Stock in the Reverse Stock Split, as determined by the Adamis Board and specified in the Adamis Reverse Stock Split Amendment.

"Sarbanes-Oxley Act" shall mean the Sarbanes-Oxley Act of 2002, as it may be amended from time to time.

"SEC" shall mean the United States Securities and Exchange Commission.

"*Subsidiary*" An Entity shall be deemed to be a "Subsidiary" of another Person if such Person directly or indirectly owns, beneficially or of record, (a) an amount of voting securities of other interests in such Entity that is sufficient to enable such Person to elect at least a majority of the members of such Entity's board of directors or other governing body, or (b) at least 50% of the outstanding equity, voting, beneficial or financial interests in such Entity.

"*Superior Proposal*" means an Acquisition Proposal that the board of directors of a Party determines, in its reasonable judgment, to be more favorable to such Party's stockholders than the terms of the transactions contemplated by this Agreement.

"Tax" or *"Taxes"* shall mean any and all federal, state, local or foreign taxes of any country, assessments and other similar governmental charges, duties, impositions and liabilities, including taxes based upon or measured by gross receipts, income, profits, sales, use and occupation, and value added, ad valorem, stamp transfer, franchise, withholding, payroll, recapture, employment, excise and property taxes, together with all interest, penalties and additions imposed with respect to such amounts and any obligations under any agreements or arrangements with any other person with respect to such amounts and including any liability for taxes of a predecessor Entity.

"*Tax Return*" shall mean any return (including any information return), report, statement, declaration, estimate, schedule, notice, notification, form, election, certificate or other document or information filed with or submitted to, or required to be filed with or submitted to, any Governmental Entity in connection with the determination, assessment, collection or payment of any Tax or in connection with the administration, implementation or enforcement of or compliance with any Law relating to any Tax.

"Treasury Regulations" shall mean the official interpretations of the Code promulgated by the United States Department of the Treasury.

Additionally, the following terms have the meanings assigned to such terms in the Sections of this Agreement set forth below opposite such term:

| Defined Term | Section of Agreement |
|---|--------------------------------|
| Action | 2.6(b) |
| Adamis | Preamble |
| Adamis Average Closing Price | 1.7(b) |
| Adamis Board | Background– Paragraph A |
| Adamis Board Recommendation | 5.3(b) |
| Adamis Common Stock | Background – Paragraph B |
| Adamis Common Stock Consideration Cap | 1.6(a) |
| Adamis Disclosure Schedule | First paragraph of Article III |
| Adamis Employee Plans | 3.14(a) |
| Adamis Financial Statements | 3.9(b) |
| Adamis Patent and Proprietary Rights | 3.12(a) |
| Adamis Percentage Threshold | 1.7 |
| Adamis Proposals | 5.3(b) |
| Adamis Restated Certificate | 1.4(b) |
| Adamis Reverse Stock Split Amendment | 1.5(a) |
| Adamis Reverse Stock Split Proposal | 5.13 |
| Adamis SEC Reports | 3.9(a) |
| Adamis Stockholders Meeting | 5.3(a) |
| Agreement Date | Preamble |
| Allocation Certificate | 5.22 |
| Ancillary Agreements | 2.3(a) |
| Base Date | 2.10 |
| Certificate of Merger | 1.3 |
| Change in the Adamis Board Recommendation | 5.6(c) |
| Closing | 1.3 |
| Closing Date | 1.3 |
| Closing Price | 1.7(c) |
| Confidentiality Action | 10.10 |
| Confidentiality Agreement | 5.16 |
| Current Balance Sheet | 2.9(a) |
| D&O Indemnified Parties | 5.8(a) |
| DGCL | 1.3 |
| Dissenting Shares | 1.8 |
| Dissenting Stockholder | 1.8 |
| DMK | Preamble |

| Defined Term | Section of Agreement |
|---------------------------------------|-----------------------------|
| DMK Audited Financial Statements | 5.1(a) |
| DMK Board | Background – Paragraph A |
| DMK Board Recommendation | 5.2(b) |
| DMK Common Stock | Background – Paragraph B |
| DMK Convertible Notes | 2.7(c) |
| DMK Disclosure Schedule | Article II, first paragraph |
| DMK Employee Plans | 2.14(a) |
| DMK Financial Statements | 2.9(a) |
| DMK Interim Financial Statements | 2.9(a) |
| DMK Law Firm | 10.11(a) |
| DMK License Agreements | 7.10 |
| DMK Outstanding Shares | 1.7(d) |
| DMK Patent and Proprietary Rights | 2.12(a) |
| DMK Plan | 2.7(b) |
| DMK Restricted Stock | 1.6(c) |
| DMK Stock Certificate | 1.6(e) |
| DMK Stockholder Matters | 5.5(a) |
| DMK Stockholders Meeting | 5.2(a) |
| DMK Support Agreement | Background – Paragraph G |
| DMK Valuation | 1.7(a) |
| DMK Written Consent | 5.5(a) |
| DMK 401(k) Plan | 5.23 |
| Effective Time | 1.3 |
| Entity | 2.2 |
| ERISA | 2.14(a) |
| ERISA Affiliate | 2.14(a) |
| Excess Cap Shares | 1.6(a) |
| Exchange Agent | 1.10(a) |
| Expense Payment Amount | 9.3(b) |
| Exchange Ratio | 1.7 |
| Exchange Shares | 1.10(b) |
| Grant Date | 2.7(f) |
| Information Statement | 5.5(a) |
| Merger | Background – Paragraph A |
| Merger Consideration | 1.6(a) |
| Merger Consideration Preferred Shares | 1.6(a) |
| Merger Sub | Preamble |
| Nasdaq | 5.18 |
| Nasdaq Listing Application | 5.18 |
| New Adamis Common Stock | 1.5(a)(i) |
| NJBCA | Background – Paragraph D |
| Notice of Superior Proposal | 4.5(b)(ii) |
| Other Adamis Proposals | 5.3(b) |
| Outside Date | 9.1(b) |

| Defined Term | Section of Agreement |
|---|--------------------------|
| Plan of Merger | Background – Paragraph D |
| Pre-Closing Period | 4.1 |
| Privileged Communications | 10.11(b) |
| Product Licensee | 2.22(c) |
| Registration Statement | 5.4 |
| Required DMK Stockholder Vote | 2.3(b) |
| Reverse Stock Split | 1.5(a)(i) |
| Schedule Supplement | 5.26 |
| Securities Act | 1.6(f) |
| Series E Preferred | Background – Paragraph B |
| Series E Preferred Certificate of Designation | Background – Paragraph B |
| Series E Preferred Conversion Shares | 1.6(a) |
| Specified DMK Stockholder | 1.6(a) |
| Stockholders' Representative | 10.11(d)(i) |
| Surviving Corporation | 1.1 |
| Takeover Statute | 2.25 |
| Transfer Taxes | 5.12(d) |
| U.S. Tax Treatment | 1.12 |

EXHIBIT B

FORM OF SUPPORT AGREEMENT

EXHIBIT C

CERTIFICATE OF DESIGNATION OF PREFERENCES, RIGHTS AND LIMITATIONS OF SERIES E CONVERTIBLE PREFERRED STOCK

ADAMIS PHARMACEUTICALS CORPORATION

CERTIFICATE OF DESIGNATION OF PREFERENCES, RIGHTS AND LIMITATIONS OF

SERIES E CONVERTIBLE PREFERRED STOCK

PURSUANT TO SECTION 151 OF THE DELAWARE GENERAL CORPORATION LAW

The undersigned, David J. Marguglio, does hereby certify that:

1. David J. Marguglio is the Chief Executive Officer of Adamis Pharmaceuticals Corporation, a Delaware corporation (the "Corporation").

2. The Corporation is authorized to issue 10,000,000 shares of preferred stock.

3. The following resolutions were duly adopted by the board of directors of the Corporation (the "**Board of Directors**"):

WHEREAS, the certificate of incorporation of the Corporation provides for a class of its authorized stock known as preferred stock, consisting of 10,000,000 shares, \$0.0001 par value per share, issuable from time to time in one or more series;

WHEREAS, the Board of Directors is authorized to fix the dividend rights, dividend rate, voting rights, conversion rights, rights and terms of redemption and liquidation preferences of any wholly unissued series of preferred stock and the number of shares constituting any series and the designation thereof, of any of them; and

WHEREAS, it is the desire of the Board of Directors, pursuant to its authority as aforesaid, to fix the rights, preferences, restrictions and other matters relating to a series of the preferred stock, which shall consist of 15,000 shares of the preferred stock which the Corporation has the authority to issue, as follows:

NOW, THEREFORE, BE IT RESOLVED, that the Board of Directors does hereby provide for the issuance of a series of preferred stock for cash or exchange of other securities, rights or property and does hereby fix and determine the rights, preferences, restrictions and other matters relating to such series of preferred stock as follows:

TERMS OF PREFERRED STOCK

<u>Section 1</u>. <u>Definitions</u>. For the purposes hereof, the following terms shall have the following meanings:

"Affiliate" means, as to any Person (the "subject Person"), any other Person (a) that directly or indirectly through one or more intermediaries controls or is controlled by, or is under direct or indirect common control with, the subject Person, (b) that directly or indirectly beneficially owns or holds ten percent (10%) or more of any class of voting equity of the subject Person, or (c) ten percent (10%) or more of the voting equity of which is directly or indirectly beneficially owned or held by the subject Person. For the purposes of this definition, "control" when used with respect to any Person means the power to direct the management and policies of such Person, directly or indirectly, whether through the ownership of voting securities, through representation on such Person's board of directors or other management committee or group, by contract or otherwise.

"Beneficial Ownership Limitation" shall have the meaning set forth in Section 6(c).

"**Business Day**" means any day except any Saturday, any Sunday, any day which is a federal legal holiday in the United States or any day on which banking institutions in the State of California are authorized or required by law or other governmental action to close.

"Commission" means the United States Securities and Exchange Commission.

"**Common Stock**" means the Corporation's common stock, par value \$0.0001 per share, and stock of any other class of securities into which such securities may hereafter be reclassified or changed.

"**Common Stock Equivalents**" means any securities of the Corporation which would entitle the holder thereof to acquire at any time Common Stock, including, without limitation, any convertible debt, preferred stock, rights, options, warrants or other instrument that is at any time convertible into or exercisable or exchangeable for, or otherwise entitles the holder thereof to receive, Common Stock.

"Conversion Date" shall have the meaning set forth in Section 6(a).

"Conversion Ratio" shall initially be equal to one thousand (1,000) per whole share of Series E Preferred Stock, and shall be subject to adjustment as provided herein.

"Conversion Shares" means, collectively, the shares of Common Stock issuable upon conversion of the shares of Series E Preferred Stock in accordance with the terms hereof.

"Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

"Fair Value" means the last reported closing stock price of a share of Common Stock on the principal Trading Market on which the Common Stock is listed as of the Trading Day immediately prior to the date on which a Notice of Conversion is delivered to the Company.

"Holder" means a holder of shares of Series E Preferred Stock.

"**Merger Agreement**" means the Agreement and Plan of Merger and Reorganization, dated as of February 24, 2023, by and among the Corporation, Aardvark Merger Sub Inc., a Delaware corporation, and DMK Pharmaceuticals Corporation, a New Jersey corporation, as amended, modified or supplemented from time to time in accordance with its terms.

"Notice of Conversion" shall have the meaning set forth in Section 6(a).

"Original Issue Date" means the date of the first issuance of any shares of Series E Preferred Stock.

"**Person**" means an individual or corporation, partnership, trust, incorporated or unincorporated association, joint venture, limited liability company, joint stock company, government (or an agency or subdivision thereof) or other entity of any kind.

"Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

"Series E Preferred Stock" shall have the meaning set forth in Section 2.

"Share Delivery Date" shall have the meaning set forth in Section 6(b).

"Trading Day" means a day on which the principal Trading Market is open for business.

"**Trading Market**" means any of the following markets or exchanges on which the Common Stock is listed or quoted for trading on the date in question: the NYSE MKT, the Nasdaq Capital Market, the Nasdaq Global Market, the Nasdaq Global Select Market, the New York Stock Exchange or the OTC Bulletin Board (or any successors to any of the foregoing).

"Transfer Agent" means American Stock Transfer & Trust Company, LLC or any successor entity appointed as the transfer agent of the Corporation.

Section 2. Designation, Amount and Par Value. The series of preferred stock shall be designated as the Corporation's Series E Convertible Preferred Stock (the "Series E Preferred Stock") and the number of shares so designated shall be 15,000. Each share of Series E Preferred Stock shall have a par value of \$0.0001 per share.

Section 3. <u>Dividends</u>. Except for stock dividends or distributions for which adjustments are to be made pursuant to Section 7, Holders shall be entitled to receive, and the Corporation shall pay, dividends on shares of Series E Preferred Stock equal (on an as-if-converted-to-Common-Stock basis, disregarding for such purpose any conversion limitations hereunder) to and in the same form as dividends actually paid on shares of the Common Stock when, as and if such dividends are paid on shares of the Common Stock. No other dividends shall be paid on shares of Series E Preferred Stock.

Section 4. Voting Rights. Except as otherwise provided herein or as otherwise required by law, the Holders shall be entitled to vote with the holders of outstanding shares of Common Stock, voting together as a single class, with respect to any and all matters presented to the stockholders of the Corporation for their action or consideration (whether at a meeting of stockholders of the Corporation, by written action of stockholders in lieu of a meeting or otherwise). In any such vote, each Holder shall be entitled to a number of votes equal to the number of shares of Common Stock into which the Series E Preferred Stock held by such Holder is convertible pursuant to Section 6 herein (and after giving effect to and taking into account the Beneficial Ownership Limitation set forth in Section 6(c)) as of the record date for such vote or written consent or, if there is no specified record date, as of the date of such vote or written consent. Each Holder of outstanding Series E Preferred Stock shall be entitled to notice of all stockholder meetings (or requests for written consent) in accordance with the Corporation's bylaws. In addition, as long as any shares of Series E Preferred Stock, directly or indirectly, by merger, consolidation, recapitalization or otherwise (a) alter or change adversely the powers, preferences or rights given to the Series E Preferred Stock or alter or amend this Certificate of Designation or (b) increase the number of authorized shares of Series E Preferred Stock, or (c) enter into any agreement with respect to any of the foregoing.

<u>Section 5.</u> <u>Liquidation</u>. Upon any liquidation, dissolution or winding-up of the Corporation, whether voluntary or involuntary (a "**Liquidation**"), subject to the rights of the holders of any other outstanding series of Preferred Stock of the Corporation, the Holders shall be entitled to receive, pari passu with the holders of Common Stock of the Corporation, out of the assets, whether capital or surplus, of the Corporation an amount equal to such amount per share as would have been payable had all shares of Series E Preferred Stock been converted into Common Stock pursuant to Section 6 herein (without giving effect to any limitation on conversion as a result of the Beneficial Ownership Limitation set forth in Section 6(c)) immediately prior to such liquidation, dissolution or winding up.

Section 6. Conversion.

Conversions at Option of Holder. Each share of Series E Preferred Stock (or fraction thereof) shall be convertible, at any time and a) from time to time at the option of the Holder thereof, into the number of shares of Common Stock (subject to the limitations set forth in Section 6(c)) equal to the Conversion Ratio (and giving effect proportionately to any conversion of a fraction of a share of Series E Preferred Stock). Holders shall effect conversions by providing the Corporation with a written notice of conversion (a "Notice of Conversion") substantially in the form attached hereto on the Trading Day on which such Holder wishes to effect such conversion (the "Conversion Date"). Each Notice of Conversion shall specify the number of shares of Series E Preferred Stock to be converted, the applicable Conversion Ratio, the number of shares of Common Stock to be issued, and the number of shares of Series E Preferred Stock owned subsequent to the conversion at issue. The Conversion Shares shall be deemed to have been issued, and the Holder or any other person so designated to be deemed to have become a holder of record of such shares for all purposes, as of the date of delivery to the Corporation of the Notice of Conversion. To effect conversions of shares of Series E Preferred Stock, a Holder shall not be required to surrender the certificate(s) representing the shares of Series E Preferred Stock to the Corporation unless all of the shares of Series E Preferred Stock represented thereby are so converted, in which case such Holder shall deliver the certificate representing such shares of Series E Preferred Stock promptly following the Conversion Date at issue. Conversions of less than the total amount of shares of Series E Preferred Stock represented by a certificate held by Holder will have the effect of lowering the outstanding number of Series E Preferred Stock held by such Holder by an amount equal to the number so converted, as if the original stock certificate(s) were cancelled and one or more new stock certificates evidencing the new number of shares of Series E Preferred Stock were issued; provided, however that in such cases the Holder may request that the Corporation deliver to the Holder a certificate representing such nonconverted shares of Series E Preferred Stock; provided, further, that the failure of the Corporation to deliver such new certificate shall not affect the rights of the Holder to submit a further Notice of Conversion with respect to such Series E Preferred Stock and, in any such case, the Holder shall be deemed to have submitted the original of such new certificate at the time that it submits such further Notice of Conversion. In the case of a dispute between the Corporation and a Holder as to the calculation of the Conversion Ratio or the number of Conversion Shares issuable upon a conversion (including without limitation the calculation of any adjustment to the Conversion Ratio pursuant to Section 7 below), the Corporation shall issue to such Holder the number of Conversion Shares that are not disputed within the time periods specified in Section 6(b) below and shall submit the disputed calculations to a certified public accounting firm of national reputation (other than the Corporation's regularly retained accountants) within three (3) Trading Days following the Corporation's receipt of such Holder's Notice of Conversion. The Corporation shall cause such accountant to calculate the Conversion Ratio as provided herein and to notify the Corporation and such Holder of the results in writing no later than three (3) Trading Days following the day on which such accountant received the disputed calculations. Such accountant's calculation shall be deemed conclusive absent manifest error. The fees of any such accountant shall be borne by the party whose calculations were most at variance with those of such accountant.

b) <u>Mechanics of Conversion</u>

i. <u>Delivery of Conversion Shares Upon Conversion</u>. Not later than three (3) Trading Days after each Conversion Date (the "**Share Delivery Date**"), the Corporation shall deliver, or cause to be delivered, to the converting Holder the number of Conversion Shares being acquired upon the conversion of the Series E Preferred Stock. If, in the case of any Notice of Conversion, such Conversion Shares are not delivered to or as directed by the applicable Holder by the Share Delivery Date, the Holder shall be entitled to elect by written notice to the Corporation at any time on or before its receipt of such Conversion Shares, to rescind such Conversion, in which event the Corporation shall promptly return to the Holder any original Series E Preferred Stock certificate delivered to the Corporation and the Holder shall promptly return to the Conversion Shares issued to such Holder pursuant to the rescinded Conversion Notice.

ii. <u>Reservation of Shares Issuable Upon Conversion</u>. The Corporation shall at all times reserve and keep available out of its authorized and unissued shares of Common Stock for the sole purpose of issuance upon conversion of the outstanding Series E Preferred Stock, free from preemptive rights or any other actual contingent purchase rights of Persons other than the Holder (and the other holders of the Series E Preferred Stock), not less than such aggregate number of shares of the Common Stock as shall be issuable (taking into account the adjustments and restrictions of Section 7) upon the conversion of the then outstanding shares of Series E Preferred Stock (without giving effect to any limitation on conversion as a result of the Beneficial Ownership Limitation set forth in Section 6(c)).

iii. <u>Fractional Shares</u>. No fractional shares of Common Stock shall be issued upon the conversion of the Series E Preferred Stock. In lieu of any fractional shares to which the Holder would otherwise be entitled upon such conversion, the Corporation shall at its election, either pay a cash adjustment in respect of such fractional share of Common Stock in an amount equal to such fraction multiplied by the Fair Value or round up to the next whole share (after aggregating all fractional shares).

iv. <u>Transfer Taxes and Expenses</u>. The issuance of Conversion Shares on conversion of the Series E Preferred Stock shall be made without charge to any Holder for any documentary stamp or similar taxes that may be payable in respect of the issue or delivery of such Conversion Shares, provided that the Corporation shall not be required to pay any tax that may be payable in respect of any transfer involved in the issuance and delivery of any such Conversion Shares upon conversion in a name other than that of the Holders of such shares of Series E Preferred Stock, and the Corporation shall not be required to issue or deliver such Conversion Shares unless or until the Person or Persons requesting the issuance thereof shall have paid to the Corporation the amount of such tax or shall have established to the satisfaction of the Corporation that such tax has been paid.

Beneficial Ownership Limitation. The Corporation shall not effect any conversion of the Series E Preferred Stock, and a Holder shall not have the right to convert any portion of the Series E Preferred Stock, to the extent that, after giving effect to the conversion set forth on the applicable Notice of Conversion, such Holder would beneficially own in excess of the Holder Beneficial Ownership Limitation, or such Holder together with such Holder's Affiliates and any Persons acting as a group together with such Holder or Affiliates (such Persons, "Attribution Parties") would beneficially own in excess of the Affiliates Beneficial Ownership Limitation (as defined below). For purposes of the foregoing sentence, the number of shares of Common Stock beneficially owned by such Holder and its Affiliates shall include the number of shares of Common Stock issuable upon conversion of the Series E Preferred Stock with respect to which such determination is being made, but shall exclude the number of shares of Common Stock which are issuable upon (i) conversion of the remaining, unconverted Series E Preferred Stock beneficially owned by such Holder or any of its Affiliates or Attribution Parties and (ii) exercise or conversion of the unexercised or unconverted portion of any other securities of the Corporation (including, without limitation, any Common Stock Equivalents) subject to a limitation on conversion or exercise analogous to the limitation contained herein beneficially owned by such Holder or any of its Affiliates or Attribution Parties. Except as set forth in the preceding sentence, for purposes of this Section 6(c), beneficial ownership shall be calculated in accordance with Section 13(d) of the Exchange Act and the rules and regulations promulgated thereunder, it being acknowledged by each Holder that the Corporation is not representing to such Holder that such calculation is in compliance with Section 13(d) of the Exchange Act and the Holder is solely responsible for any schedules required to be filed in accordance therewith. To the extent that the limitation contained in this Section 6(c) applies, the determination of whether the Series E Preferred Stock is convertible (in relation to other securities owned by such Holder together with any Affiliates and Attribution Parties) and of how many shares of Series E Preferred Stock are convertible shall be in the reasonable judgment of such Holder, in each case subject to the Holder Beneficial Ownership Limitation or the Affiliates Beneficial Ownership Limitation, and the Corporation shall have no obligation to verify or confirm the accuracy of such determination. To ensure compliance with this restriction, each Holder will be deemed to represent to the Corporation each time it delivers a Notice of Conversion that such Notice of Conversion has not violated the restrictions set forth in this paragraph and the Corporation shall have no obligation to verify or confirm the accuracy of such determination. In addition, a determination as to any group status as contemplated above shall be determined in accordance with Section 13(d) of the Exchange Act and the rules and regulations promulgated thereunder. For purposes of this Section 6(c), in determining the number of outstanding shares of Common Stock, a Holder may rely on the number of outstanding shares of Common Stock as stated in the most recent of the following: (i) the Corporation's most recent periodic or annual report filed with the Commission, as the case may be, (ii) a more recent public announcement by the Corporation or (iii) a more recent written notice by the Corporation or the Transfer Agent setting forth the number of shares of Common Stock outstanding. Upon the written or oral request of a Holder, the Corporation shall within two (2) Trading Days confirm orally and in writing to such Holder the number of shares of Common Stock then outstanding. In any case, the number of outstanding shares of Common Stock shall be determined after giving effect to the conversion or exercise of securities of the Corporation, including the Series E Preferred Stock, by such Holder or its Affiliates or Attribution Parties since the date as of which such number of outstanding shares of Common Stock was reported. The "Holder Beneficial Ownership Limitation" shall be 9.99% of the number of shares of the Common Stock outstanding immediately after giving effect to the issuance of shares of Common Stock issuable upon conversion of Series E Preferred Stock held by the applicable Holder. The "Affiliates Beneficial Ownership Limitation" shall be 9.99% of the number of shares of the Common Stock outstanding immediately after giving effect to the issuance of shares of Common Stock issuable upon conversion of Series E Preferred Stock held by the applicable Holder and its Affiliates. The Holder Beneficial Ownership Limitation together with the Affiliates Beneficial Ownership Limitation is collectively known as the "Beneficial Ownership Limitation." The provisions of this paragraph shall be construed and implemented in a manner otherwise than in strict conformity with the terms of this Section 6(c) to correct this paragraph (or any portion hereof) which may be defective or inconsistent with the intended Beneficial Ownership Limitation contained herein or to make changes or supplements necessary or desirable to give effect to such limitation. The limitations contained in this paragraph shall apply to a successor holder of Series E Preferred Stock.

Nasdaq Issuance Limitation. Notwithstanding anything herein to the contrary, the Corporation shall not be obligated to issue any d) shares of Common Stock, and the Holders shall not have the right to receive, upon conversion of the Series E Preferred Stock, taken as a whole, any shares of Common Stock to the extent such issuance of shares of Common Stock would exceed that number of shares of Common Stock which the Corporation may issue in the aggregate pursuant to the transactions contemplated under the Merger Agreement (including pursuant to this Certificate of Designation) without breaching the Corporation's obligations under the rules and regulations of the Nasdaq Capital Market (the "Exchange Cap"). In furtherance of the above, no holder of Series E Preferred Stock shall be issued, in the aggregate pursuant to the terms of this Certificate of Designation, shares of Common Stock in an amount greater than the product of the Exchange Cap multiplied by a fraction, the numerator of which is the number of shares of Series E Preferred Stock held by the Holder and the denominator of which is the aggregate number of shares of Series E Preferred Stock issued on the Original Issue Date to all Holders (with respect to each Holder, the "Exchange Cap Allocation"). In the event that the Holder shall sell or otherwise transfer any of the Holder's Series E Preferred Stock, the transferee shall be allocated a pro rata portion of the Holder's Exchange Cap Allocation, and the restrictions of the prior sentence shall apply to such transferee with respect to the portion of the Exchange Cap Allocation allocated to such transferee. In the event that any holder of Series E Preferred Stock shall convert all of such holder's Series E Preferred Stock into a number of shares of Common Stock which, in the aggregate, is less than such holder's Exchange Cap Allocation, then the difference between such holder's Exchange Cap Allocation and the number of shares of Common Stock actually issued to such holder shall be allocated to the respective Exchange Cap Allocations of the remaining holders of Series E Preferred Stock on a pro rata basis in proportion to the shares of Series E Preferred Stock then held by each such Holder.

⁷

Section 7. Certain Adjustments.

a) <u>Stock Dividends and Stock Splits</u>. If the Corporation, at any time while the Series E Preferred Stock is outstanding: (i) pays a stock dividend or otherwise makes a distribution or distributions payable in shares of Common Stock on shares of Common Stock or any other Common Stock Equivalents (which, for avoidance of doubt, shall not include any shares of Common Stock issued by the Corporation upon conversion of, or payment of a dividend on, this Series E Preferred Stock), (ii) subdivides outstanding shares of Common Stock into a larger number of shares, (iii) combines (including by way of a reverse stock split) outstanding shares of Common Stock into a smaller number of shares, or (iv) issues, in the event of a reclassification of shares of the Common Stock, any shares of capital stock of the Corporation, then the Conversion Ratio shall be divided by a fraction of which the numerator shall be the number of shares of Common Stock outstanding immediately before such event, and of which the denominator shall be the number of shares of Common Stock outstanding immediately after such event. Any adjustment made pursuant to this Section 7(a) shall become effective immediately after the record date for the determination of stockholders entitled to receive such dividend or distribution and shall become effective immediately after the effective date in the case of a subdivision, combination or re-classification.

b) <u>Subsequent Rights Offerings</u>. In addition to any adjustments pursuant to Section 7(a) above, if at any time the Corporation grants, issues or sells any Common Stock Equivalents pro rata to all the record holders of any class of shares of Common Stock (the "**Purchase Rights**"), then the Holder will be entitled to acquire, upon the terms applicable to such Purchase Rights, the aggregate Purchase Rights which the Holder could have acquired if the Holder had held the number of shares of Common Stock acquirable upon conversion of the Series E Preferred Stock (without regard to any limitations on conversion hereof, including without limitation, the Beneficial Ownership Limitation) immediately before the date on which a record is taken for the grant, issuance or sale of such Purchase Rights, or, if no such record is taken, the date as of which the record holders of shares of Common Stock are to be determined for the grant, issue or sale of such Purchase Rights (provided, however, to the extent that the Holder's right to participate in any such Purchase Right would result in the Holder exceeding the Beneficial Ownership Limitation, then the Holder shall not be entitled to participate in such Purchase Right to such extent (or beneficial ownership of such shares of Common Stock as a result of such Purchase Right to such extent) and such Purchase Right to such extent shall be held in abeyance for the Holder until such time, if ever, as its right thereto would not result in the Holder exceeding the Beneficial Ownership Limitation).

c) <u>Pro Rata Distributions</u>. Subject to Section 3, during such time as the Series E Preferred Stock is outstanding, if the Corporation declares or makes any dividend or other distribution of its assets (or rights to acquire its assets) to all holders of shares of Common Stock, by way of return of capital or otherwise (including, without limitation, any distribution of cash, stock or other securities, property or options by way of a dividend, spin off, reclassification, corporate rearrangement, scheme of arrangement or other similar transaction) (a "**Distribution**"), at any time after the issuance of the Series E Preferred Stock, then, in each such case, the Holder shall be entitled to participate in such Distribution to the same extent that the Holder would have participated therein if the Holder had held the number of shares of Common Stock acquirable upon complete Conversion of the Series E Preferred Stock (without regard to any limitations on Conversion hereof, including without limitation, the Beneficial Ownership Limitation) immediately before the date of which a record is taken for such Distribution (provided, however, to the extent that the Holder's right to participate in any such Distribution would result in the Holder exceeding the Beneficial Ownership Limitation, then the Holder shall not be entitled to participate in such Distribution to such extent (or in the beneficial ownership of any shares of Common Stock as a result of such Distribution to such extent) and the portion of such Distribution shall be held in abeyance for the benefit of the Holder until such time, if ever, as its right thereto would not result in the Holder exceeding the Beneficial Ownership Limitation).

d) <u>Merger; Sale of Assets</u>. If at any time while the Series E Preferred Stock is outstanding: (i) the Corporation effects any merger or consolidation of the Corporation with or into another Person pursuant to which the shares of capital stock of the Corporation outstanding immediately prior to such merger or consolidation are converted into or exchanged for shares of another corporation or entity, or are converted into or exchanged for equity securities that represent, less than a majority, by voting power, of the equity securities of (1) the surviving or resulting party or (2) if the surviving or resulting party is a wholly owned subsidiary of another party immediately following such merger or consolidation, the parent of such surviving or resulting party, immediately following such merger or consolidation; or (ii) the Corporation sells all or substantially all of its assets in a single transaction or a series of related transactions (each, a "**Merger or Sale**"), then each Holder of the Series E Preferred Stock shall be entitled to receive such number of shares of common stock of the successor or acquiring corporation and/or such other or additional consideration as are receivable by virtue of such Merger or Sale by a holder of the number of shares of Common Stock for which the Series E Preferred Stock held by the Holder is convertible immediately prior to such Merger or Sale (without regard to any limitation in Section 6(c) on the conversion of this Series E Preferred Stock).

e) <u>Calculations</u>. All calculations under this Section 7 shall be made to the nearest cent or the nearest 1/1,000th of a share, as the case may be. For purposes of this Section 7, the number of shares of Common Stock deemed to be issued and outstanding as of a given date shall be the sum of the number of shares of Common Stock (excluding any treasury shares of the Corporation) issued and outstanding.

f) <u>Notice of Adjustment</u>. Whenever the Conversion Ratio is adjusted pursuant to any provision of this Section 7, the Corporation shall promptly deliver to each Holder a notice setting forth the Conversion Ratio after such adjustment and setting forth a brief statement of the facts requiring such adjustment.

g) <u>Adjustments</u>. In the event that at any time, as a result of an adjustment made pursuant to this Section 7, the Holder shall, upon conversion of such Holder's Series E Preferred Stock, become entitled to receive securities or assets (other than Common Stock) then, wherever appropriate, all references herein to shares of Common Stock shall be deemed to refer to and include such shares and/or other securities or assets; and thereafter the number of such shares and/or other securities or assets shall be subject to adjustment from time to time in a manner and upon terms as nearly equivalent as practicable to the provisions of this Section 7.

Section 8. Miscellaneous.

a) <u>Lost or Mutilated Preferred Stock Certificate</u>. If a Holder's Series E Preferred Stock certificate shall be mutilated, lost, stolen or destroyed, the Corporation shall execute and deliver, in exchange and substitution for and upon cancellation of a mutilated certificate, or in lieu of or in substitution for a lost, stolen or destroyed certificate, a new certificate for the shares of Series E Preferred Stock so mutilated, lost, stolen or destroyed, but only upon receipt of evidence of such loss, theft or destruction of such certificate, and of the ownership hereof reasonably satisfactory to the Corporation.

b) <u>Severability</u>. If any provision of this Certificate of Designation is invalid, illegal or unenforceable, the balance of this Certificate of Designation shall remain in effect, and if any provision is inapplicable to any Person or circumstance, it shall nevertheless remain applicable to all other Persons and circumstances.

c) <u>Next Business Day</u>. Whenever any payment or other obligation hereunder shall be due on a day other than a Business Day, such payment shall be made on the next succeeding Business Day.

d) <u>Headings</u>. The headings contained herein are for convenience only, do not constitute a part of this Certificate of Designation and shall not be deemed to limit or affect any of the provisions hereof.

e) <u>Status of Converted Preferred Stock</u>. Shares of Series E Preferred Stock may only be issued pursuant to the Merger Agreement. If any shares of Series E Preferred Stock shall be converted or reacquired by the Corporation, such shares shall resume the status of authorized but unissued shares of preferred stock and shall no longer be designated as Series E Convertible Preferred Stock.

f) <u>Transfers</u>. Upon written notice to the Corporation, a Holder may sell, transfer, assign, pledge or otherwise dispose of all or any portion of the Series E Preferred Stock to any person or entity as long as such transaction is the subject of an effective registration statement under the Securities Act or is exempt from registration thereunder and otherwise is made in accordance with the terms of the Merger Agreement and any letter of transmittal executed in connection with the issuance Series E Preferred Stock pursuant to the Merger Agreement. From and after the date of any such sale or transfer, the transferee thereof shall be deemed to be a Holder. Upon any such sale or transfer, the Corporation shall, promptly following the return of the certificate or certificates representing the Series E Preferred Stock that are the subject of such sale or transfer, issue and deliver to such transferee a new certificate in the name of such transferee.

g) Notices. Any and all notices or other communications or deliveries to be provided by the Holders hereunder including, without limitation, any Notice of Conversion, shall be in writing and delivered personally, via email or sent by a nationally recognized overnight courier service, addressed to the Corporation, at 11682 El Camino Real, Suite 300, San Diego, CA 92130, Attention: Corporate Secretary, or such other email address or mailing address as the Corporation may specify for such purposes by notice to the Holders delivered in accordance with this Section. Any and all notices or other communications or deliveries to be provided by the Corporation hereunder shall be in writing and delivered personally, by email at the email address of such Holder appearing on the books of the Corporation, or if no such email address appears on the books of the Corporation, sent by a nationally recognized overnight courier service addressed to each Holder, at the principal place of business of such Holder. Any notice or other communication or deliveries hereunder shall be deemed given and effective on the earliest of (i) the date of transmission, if such notice or communication is delivered via email at the email address specified in this Section prior to 5:30 p.m. (New York City time) on any date, (ii) the date immediately following the date of transmission, if such notice or communication is delivered via email at the email address specified in this Section Business Day following the date of mailing, if sent by nationally recognized overnight courier service, or (iv) upon actual receipt by the party to whom such notice is required to be given.

RESOLVED, FURTHER, that the president and the secretary of the Corporation be and they hereby are authorized and directed to prepare and file this Certificate of Designation of Preferences, Rights and Limitations in accordance with the foregoing resolution and the provisions of Delaware law.

Name: David J. Marguglio Title: Chief Executive Officer

NOTICE OF CONVERSION

The undersigned hereby elects to convert shares of Series E Preferred Stock (the "**Preferred Stock**"), represented by stock certificate No(s) ________, into shares of common stock ("**Common Stock**") of ADAMIS PHARMACEUTICALS CORPORATION, according to the terms and conditions of the Certificate of Designation relating to the Preferred Stock (the "**Certificate of Designation**"), as of the date written below. Capitalized terms used herein and not otherwise defined shall have the respective meanings set forth in the Certificate of Designation.

| Date of Conversion: | Date of Conversion: | | |
|---|--|--|--|
| | | | |
| Preferred Stock to be Converted: | Preferred Stock to be Converted: | | |
| | Applicable Conversion Ratio: | | |
| | | | |
| Number of Shares of Common Stock to be Issued: | Number of Shares of Common Stock to be Issued: | | |
| | Preferred Stock Held Subsequent to Issuance: | | |
| Name of Holdow | | | |
| | | | |
| Address: | | | |
| | | | |
| | | | |
| Signature: | | | |
| Na | me: | | |
| Tit | le: | | |
| | | | |
| Holder Requests Delivery to be made: (check one) | | | |
| | | | |
| By Delivery of Physical Certificates to the Above Address Through Depository Trust Corporation | | | |
| (Account) | | | |
| () | | | |
| | | | |

ANNEX 5.11

DIRECTORS AND OFFICERS

Officers

In addition to the persons who are currently officers of Adamis on the date of the Agreement, effective at the Effective Time, Ebrahim Versi, M.D., Ph.D., shall become Chief Executive Officer of Adamis.

Directors

Adamis shall have the right to designate three (3) persons who will be directors of Adamis immediately after the Effective Time and DMK shall have the right to designate two (2) persons who will be directors of Adamis immediately after the Effective Time.

- 1. David J. Marguglio and Richard Williams shall resign as directors and Howard C. Birndorf, Meera J. Desai, Ph.D. and Vickie Reed shall continue as directors, as the designees of Adamis.
- 2. Ebrahim Versi, M.D., Ph.D. and Jannine Versi shall be appointed as directors of Adamis, as the designees of DMK, with Dr. Versi serving as Chairman of the Board.

SUPPORT AGREEMENT

THIS SUPPORT AGREEMENT (this "**Support Agreement**") is made and entered into as of February 24, 2023, by and between (i) each of the undersigned identified on the signature page hereof as a "Securityholder" (each, a "**Securityholder**"); (ii) DMK Pharmaceuticals Corporation, a New Jersey corporation ("**DMK**"); (iii) Adamis Pharmaceuticals Corporation, a Delaware corporation ("**APC**"); and (iv) Aardvark Merger Sub, Inc., a Delaware corporation ("**Merger Sub,**" and together with APC, "**Adamis**").

BACKGROUND

A. DMK, APC and Merger Sub have entered into an Agreement and Plan of Merger and Reorganization of even date herewith (the "**Merger Agreement**"), which provides for the merger of DMK with and into Merger Sub, a newly-created and wholly-owned subsidiary of APC, with Merger Sub as the surviving corporation (the "**Merger**"). Capitalized terms used herein but not otherwise defined shall have the meaning ascribed to them in the Merger Agreement.

B. Securityholder is the beneficial owner (as defined in Rule 13d-3 under the Securities Exchange Act of 1934, as amended (the "**Exchange Act**")) of such number of shares of the outstanding capital stock of DMK, as are indicated on <u>Exhibit A</u> to this Support Agreement. Securityholder may also receive in the future shares of Common Stock of DMK upon the exercise of outstanding options or warrants held by Securityholder, upon conversion or exchange of outstanding convertible promissory notes of DMK, or upon exercise or conversion of such other security of DMK which may be directly or indirectly convertible into or exercisable or exchangeable for shares of Common Stock of DMK.

C. In consideration of the execution of the Merger Agreement by the parties thereto and to induce DMK and Adamis to enter into the Merger Agreement, Securityholder (solely in Securityholder's capacity as such) desires to enter into this Agreement and vote the Shares (as defined below) and other such shares of capital stock of DMK over which Securityholder has voting power so as to facilitate consummation of the Merger.

AGREEMENT

NOW, THEREFORE, intending to be legally bound, the parties hereto agree as follows:

1. <u>Certain Definitions</u>. Capitalized terms not defined herein shall have the meanings ascribed to them in the Merger Agreement. For purposes of this Support Agreement:

(a) **"Expiration Date**" shall mean the earlier to occur of (i) such date and time as the Merger Agreement shall have been terminated pursuant to the terms thereof, or (ii) such date and time as the Merger has been consummated in accordance with the terms of the Merger Agreement.

(b) Securityholder shall be deemed to "**Own**" or to have acquired "**Ownership**" of a security if Securityholder: (i) is the record owner of such security; or (ii) is the "beneficial owner" (within the meaning of Rule 13d-3 under the Exchange Act) of such security.

(c) "**Person**" shall mean any (i) individual, (ii) corporation, limited liability company, partnership or other entity, or (iii) governmental authority.

(d) **"Shares**" shall mean: (i) all securities of DMK (including all shares of DMK Common Stock, and all shares of DMK Common Stock that are issuable upon exercise, conversion or exchange of outstanding options, warrants, convertible notes and other rights to acquire shares of DMK Common Stock) Owned by Securityholder as of the date of this Support Agreement or thereafter acquired by or issued to Securityholder; and (ii) all additional securities of DMK of which Securityholder acquires Ownership during the period from the date of this Support Agreement through the Expiration Date (including by way of stock dividend or distribution, split-up, recapitalization, combination, exchange of shares and the like).

(e) **"Transfer**". A Person shall be deemed to have effected a "**Transfer**" of a security if such person directly or indirectly: (i) sells, pledges, encumbers, assigns, grants an option with respect to, transfers or disposes of such security or any interest in such security; (ii) enters into an agreement or commitment providing for the sale of, pledge of, encumbrance of, assignment of, grant of an option with respect to, transfer of or disposition of such security or any interest therein; or (iii) reduces such Person's beneficial ownership of, interest in or risk relating to such security.

2. <u>Transfer of Shares</u>.

(a) <u>Transfer Restrictions</u>. Securityholder agrees that, during the period from the date of this Support Agreement through the Expiration Date, Securityholder shall not cause or permit any Transfer of any of the Shares to be effected; provided that, notwithstanding the foregoing, Securityholder shall not be restricted from effecting a Transfer of any Shares to any member of Securityholder's immediate family or to a revocable trust of which the transferor Securityholder is the trustee for the benefit of Securityholder and/or any member of Securityholder's immediate family provided that (A) each such transferee shall be an "accredited investor" as defined in Regulation D promulgated under the Securities Act of 1933, as amended, and have (i) executed a Support Agreement identical to this Agreement upon consummation of the Transfer and if requested by Adamis, a proxy in the form attached hereto as Exhibit B (with such modifications or in such other form as Adamis may reasonably request) and (ii) agreed in writing, for the benefit of Adamis, to hold such Shares, or such interest therein, subject to all of the terms and conditions set forth in this Agreement and to execute all instruments that the Merger Agreement contemplates will be executed by the DMK stockholders, and (B) the aggregate number of shares (whether outstanding or underlying outstanding options and warrants or such other security which may be directly or indirectly convertible into or exercisable or exchangeable for shares of capital stock of DMK) that may be so Transferred by Securityholder may not exceed one percent (1%) of DMK's outstanding Common Stock as of the date hereof. For purposes of this Agreement, "immediate family" means Securityholder's spouse, parents, siblings, children or grandchildren.

(b) <u>Transfer of Voting Rights</u>. Securityholder agrees that, during the period from the date of this Support Agreement through the Expiration Date, Securityholder shall not deposit (or permit the deposit of) any Shares in a voting trust or grant any proxy or enter into any voting agreement or similar agreement in contravention of the obligations of Securityholder under this Support Agreement with respect to any of the Shares.

3. <u>Agreement to Vote Shares</u>. At every meeting of the stockholders of DMK called (and at every adjournment thereof), and with respect to every requested action or approval by written consent of the stockholders of DMK, called or requested by DMK or its board of directors, to vote on matters with respect to the Merger, the Merger Agreement (and related plan of merger) and the other matters contemplated thereby, Securityholder (in his or her capacity as such) shall, or shall cause the holder of record on any applicable record date to, vote the Shares:

(a) in favor of approval of the Merger and the adoption and approval of the Merger Agreement and any related plan of merger, and in favor of each of the other actions contemplated by the Merger Agreement to be taken by DMK (including, without limitation, any amendments to the DMK Charter), any action requested to be taken in any DMK Information Statement distributed to DMK stockholders in connection with the Merger and the Merger Agreement, and any action required in furtherance thereof;

(b) in favor of any matter recommended by the board of directors of DMK to the stockholders that could reasonably be expected to facilitate the Merger;

(c) against approval of any proposal made in opposition to, or in competition or inconsistent with, consummation of the Merger or the transactions contemplated by the Merger Agreement (including, without limitation, any action or agreement that would result in a breach of any representation, warranty, covenant or obligation of DMK in the Merger Agreement); and

(d) in favor of waiving any notice that may have been or may be required relating to any reorganization of DMK or any subsidiary of DMK, any reclassification or recapitalization of the capital stock of DMK or any subsidiary of DMK, or any sale of assets, change of control, or acquisition of DMK or any subsidiary of DMK by any other person except for the transactions contemplated by the Merger Agreement, or any consolidation or merger of DMK or any subsidiary of DMK with or into any other person except pursuant to the Merger Agreement.

Securityholder further agrees that if a meeting of DMK stockholders is held, Securityholder shall, or shall cause the holder of record on any applicable record date to, appear at such meeting or otherwise cause the Shares to be counted as present thereat for purposes of establishing a quorum. Before the Expiration Date, the Securityholder shall not enter into any agreement or understanding with any person to vote or give instructions in any manner inconsistent with the terms of this <u>Section 3</u>.

4. <u>Agreement Not to Exercise Appraisal Rights</u>. Securityholder hereby irrevocably and unconditionally waives, and agrees not to exercise any rights to demand appraisal of any Shares which may arise with respect to the Merger or any related transaction.

5. <u>Directors and Officers</u>. Notwithstanding any provision of this Support Agreement to the contrary, nothing in this Support Agreement shall limit or restrict Securityholder from acting in Securityholder's capacity as a director or officer of DMK (it being understood that this Support Agreement shall apply to Securityholder solely in Securityholder's capacity as a stockholder of DMK) or voting in Securityholder's sole discretion on any matter other than those matters referred to in <u>Section 3</u>.

6. <u>Irrevocable Proxy</u>. If requested by Adamis, (a) Securityholder agrees to deliver to Adamis a proxy in the form attached hereto as <u>Exhibit B</u> or such other similar form and substance as Adamis may reasonably request (the "**Proxy**"), which shall be irrevocable to the fullest extent permissible by law, with respect to the Shares, and (b) Securityholder shall cause to be delivered to Adamis, an additional proxy (in the form attached hereto as <u>Exhibit B</u>) executed on behalf of the record owner of any Shares that are owned beneficially (within the meaning of Rule 13d-3 under the Exchange Act), but not of record, by Securityholder. Securityholder shall, at his, her or its own expense, perform such further acts and execute such further promises and other documents and instruments as may reasonably be required to vest in Adamis the power to carry out and give effect to the provisions of this Support Agreement.

7. <u>No Ownership Interest</u>. Nothing contained in this Support Agreement shall be deemed to vest in Adamis any direct or indirect ownership or incidence of ownership of or with respect to any Shares. All rights, ownership and economic benefits of and relating to the Shares shall remain vested in and belong to Securityholder, and Adamis shall have no authority to manage, direct, superintend, restrict, regulate, govern, or administer any of the policies or operations of DMK by virtue of this Support Agreement or exercise any power or authority to direct Securityholder in the voting of any of the Shares, except as otherwise provided herein.

8. <u>No Solicitation</u>. Securityholder agrees that, during the period from the date of this Agreement through the Expiration Date, Securityholder shall comply with the provisions of <u>Section 4.5</u> of the Merger Agreement as applicable to DMK and its stockholders, agents, officers, directors, employees and representatives.

9. <u>Representations and Warranties of the Securityholder</u>. Securityholder represents and warrants to Adamis as follows:

(a) <u>Power; Binding Agreement</u>. Securityholder has full power, authority and capacity to execute and deliver this Support Agreement and the Proxy, to perform Securityholder's obligations hereunder and to consummate the transactions contemplated hereby and thereby. If Securityholder is a corporation or other entity, the execution, delivery and performance by Securityholder of this Support Agreement and the consummation by it of the transactions contemplated hereby have been duly and validly authorized by Securityholder and no other actions or proceedings on the part of Securityholder are necessary to authorize the execution and delivery by it of this Support Agreement and the consummation by it of the transactions contemplated hereby. This Support Agreement and the Proxy have been duly executed and delivered by Securityholder, and constitute valid and binding obligations of Securityholder, enforceable against Securityholder in accordance with their terms.

(b) <u>No Conflicts</u>. Except for filings under the Exchange Act, no filing with, and no permit, authorization, consent, or approval of, any state or federal public body or authority ("**Governmental Entity**") is necessary for the execution of this Support Agreement and Proxy by Securityholder of the transactions contemplated by this Support Agreement and Proxy. None of the execution and delivery of this Support Agreement or Proxy by Securityholder, the consummation by Securityholder of the transactions contemplated by this Support Agreement and Proxy shall (i) if Securityholder is a corporation or other entity, conflict with or result in any breach of any organizational documents applicable to Securityholder, (ii) result in a violation or breach of, or constitute (with or without notice or lapse of time or both) a default (or give rise to any third party right of termination, cancellation, material modification or acceleration) under any of the terms, conditions or provisions of any note, loan agreement, bond, mortgage, indenture, license, contract, commitment, arrangement, understanding, agreement, or other instrument or obligation of any kind to which Securityholder is a party or by which Securityholder or any of its properties or assets may be bound, or (iii) violate any order, writ, injunction, decree, judgment, order, statute, rule, or regulation applicable to Securityholder or any of Securityholder's properties or assets.

(c) <u>Ownership of Shares</u>. Securityholder (i) is the beneficial owner of the shares of DMK Common Stock and the options, convertible notes and warrants or such other security which may be directly or indirectly convertible into or exercisable or exchangeable for shares of capital stock of DMK indicated on <u>Exhibit A</u> hereto, which are free and clear of any liens, adverse claims, charges, security interests, pledges or options, proxies, voting trusts or agreements, understandings or agreements, or any other rights or encumbrances whatsoever ("**Encumbrances**") (except any Encumbrances arising under securities laws or arising hereunder); and (ii) does not beneficially own any securities of DMK other than the shares of DMK Common Stock and options and warrants or such other security which may be directly or indirectly convertible into or exercisable or exchangeable for shares of capital stock of DMK indicated on <u>Exhibit A</u> hereto.

(d) <u>Voting Power</u>. Securityholder has or will have sole voting power, sole power of disposition, sole power to issue instructions with respect to the matters set forth herein, and sole power to agree to all of the matters set forth in this Support Agreement and Proxy, in each case with respect to all of Securityholder's Shares, with no limitations, qualifications or restrictions on such rights, subject to applicable federal securities laws and the terms of this Support Agreement and, if executed by Securityholder, Proxy.

(e) <u>No Finder's Fees</u>. No broker, investment banker, financial advisor or other person is entitled to any broker's, finder's, financial adviser's or other similar fee or commission in connection with the transactions contemplated by this Support Agreement and, if executed by Securityholder, Proxy based upon arrangements made by or on behalf of Securityholder.

(f) <u>Reliance by Adamis</u>. Securityholder understands and acknowledges that Adamis is entering into the Merger Agreement in reliance upon Securityholder's execution and delivery of this Support Agreement and, if executed by Securityholder, Proxy.

10. <u>Certain Restrictions</u>. Prior to the termination of this Support Agreement, Securityholder agrees not to, directly or indirectly, take any other action that would make any representation or warranty of Securityholder contained herein untrue or incorrect.

11. <u>Disclosure</u>. Securityholder agrees to permit DMK and Adamis to publish and disclose in all documents and schedules filed with the Securities and Exchange Commission or any applicable state authority or agency, and any press release or other disclosure document that DMK or Adamis, in their sole discretion, determine to be necessary or desirable in connection with the Merger and any transactions related to the Merger, Securityholder's identity and ownership of Shares and the nature of Securityholder's commitments, arrangements and understandings under this Support Agreement and, if executed by Securityholder, the Proxy.

12. <u>Consents and Waivers</u>. Securityholder hereby gives any consents or waivers that are reasonably required for the consummation of the Merger under the terms of any agreements to which the Securityholder is a party or pursuant to any rights the Securityholder may have.

13. <u>Legending of Shares</u>. If so requested by Adamis, Securityholder agrees that the Shares shall bear a legend stating that they are subject to this Support Agreement and, if executed by Securityholder, the Proxy.

14. <u>Termination</u>. This Support Agreement and, if executed by Securityholder, the Proxy delivered in connection herewith shall terminate and shall have no further force or effect as of the Expiration Date. Nothing in this <u>Section 14</u> shall relieve or otherwise limit any party of liability for breach of this Support Agreement.

15. <u>Miscellaneous</u>.

(a) <u>Validity</u>. The invalidity or unenforceability of any provision of this Support Agreement will not affect the validity or enforceability of the other provisions of this Support Agreement, which will remain in full force and effect. In the event any governmental entity of competent jurisdiction holds any provision of this Support Agreement to be null, void or unenforceable, the parties hereto will negotiate in good faith and will execute and deliver an amendment to this Support Agreement in order, as nearly as possible, to effectuate, to the extent permitted by law, the intent of the parties hereto with respect to such provision.

(b) <u>Binding Effect and Assignment</u>. This Support Agreement and all of the provisions hereof shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns, but, neither this Support Agreement nor any of the rights, interests or obligations of the parties hereto may be assigned by either of the parties without prior written consent of the other.

(c) <u>Amendments; Waiver</u>. This Support Agreement may be amended by the parties hereto and the terms and conditions hereof may be waived only by an instrument in writing signed on behalf of each of the parties hereto, or, in the case of a waiver, by an instrument signed on behalf of the party waiving compliance.

(d) <u>Specific Performance; Injunctive Relief</u>. The parties hereto acknowledge that Adamis shall be irreparably harmed and that there shall be no adequate remedy at law for a violation of any of the covenants or agreements of Securityholder set forth herein or, if executed by Securityholder, in the Proxy. Securityholder agrees that, in the event of any breach or threatened breach by Securityholder of any covenant or agreement contained in this Agreement or in the Proxy, Adamis shall be entitled, in addition to any other remedies that may be available to Adamis upon any such breach or threatened breach, Adamis shall have the right to enforce such covenants and agreements by specific performance, injunctive relief or by any other means available to Adamis at law or in equity. Securityholder further agrees that Adamis shall not be required to obtain, furnish or post any bond or similar instrument in connection with or as a condition to obtaining any remedy referred to in this Section, and Securityholder irrevocably waives any right he, she or it may have to require the obtaining, furnishing or posting of any such bond or similar instrument.

(e) <u>Non-Exclusivity</u>. The rights and remedies of Adamis under this Support Agreement are not exclusive of or limited by any other rights or remedies which it may have, whether at law, in equity, by contract or otherwise, all of which shall be cumulative (and not alternative). Without limiting the generality of the foregoing, the rights and remedies of Adamis under this Agreement, and the obligations and liabilities of Securityholder under this Agreement, are in addition to their respective rights, remedies, obligations and liabilities under common law requirements and under all applicable statutes, rules and regulations.

(f) <u>Notices</u>. All notices and other communications pursuant to this Support Agreement shall be in writing and deemed to be sufficient if contained in a written instrument and shall be deemed given if delivered personally, sent by email or other electronic transmission, sent by nationally-recognized overnight courier or mailed by registered or certified mail (return receipt requested), postage prepaid, to the parties at the following address (or at such other address for a party as shall be specified by like notice):

If to Adamis: Adamis Pharmaceuticals Corporation 11682 El Camino Real, Suite 300 San Diego, CA 92130 Attention: Chief Executive Officer Telephone No.: (858) 997-2400

If to Securityholder:

At the address for Securityholder that is set forth in the books and records of DMK

(g) <u>No Waiver</u>. The failure or delay of any party to exercise any right, power or remedy provided under this Support Agreement or otherwise available in respect of this Support Agreement at law or in equity, or to insist upon compliance by any other party with its obligation under this Support Agreement, and any custom or practice of the parties at variance with the terms of this Support Agreement, will not constitute a waiver by such party of its right to exercise any such or other right, power or remedy or to demand such compliance. Neither Adamis or DMK shall be deemed to have waived any claim available to it arising out of this Support Agreement, or any power, right, privilege or remedy of Adamis or DMK under this Agreement, unless the waiver of such claim, power, right, privilege or remedy is expressly set forth in a written instrument duly executed and delivered on behalf of Adamis or DMK (as the case may be); and any such waiver shall not be applicable or have any effect except in the specific instance in which it is given.

(h) <u>No Third Party Beneficiaries</u>. This Support Agreement is not intended to confer upon any person other than the parties hereto any rights or remedies hereunder.

(i) <u>Governing Law</u>. This Support Agreement and the Proxy shall be governed by the laws of the State of Delaware, without reference to rules of conflicts of law.

(j) Submission to Jurisdiction. All actions and proceedings arising out of or relating to this Support Agreement or Proxy shall be heard and determined exclusively in any state or federal court located in the State of Delaware. The parties hereto hereby (a) submit to the exclusive jurisdiction of any state or federal court located in Delaware, for the purpose of any action arising out of or relating to this Support Agreement or Proxy brought by any party hereto, and (b) irrevocably waive, and agree not to assert by way of motion, defense, or otherwise, in any such action, any claim that it is not subject personally to the jurisdiction of the above-named courts, that its property is exempt or immune from attachment or execution, that the action is brought in an inconvenient forum, that the venue of the action is improper, or that this Support Agreement, the Proxy or the transactions contemplated hereby may not be enforced in or by any of the above-named courts. STOCKHOLDER IRREVOCABLY WAIVES THE RIGHT TO A JURY TRIAL IN CONNECTION WITH ANY LEGAL PROCEEDING RELATING TO THIS VOTING AGREEMENT OR THE PROXY OR THE ENFORCEMENT OF ANY PROVISION OF THIS VOTING AGREEMENT OR THE PROXY. The Parties agree that service of process in any such action may be made in the manner provided for in this Agreement for delivery of notices.

(k) <u>Rules of Construction</u>. The parties hereto agree that they have been represented by counsel during the negotiation and execution of this Support Agreement and Proxy and, therefore, waive the application of any law, regulation, holding or rule of construction providing that ambiguities in an agreement or other document will be construed against the party drafting such agreement or document.

(l) <u>Entire Agreement</u>. This Support Agreement and the Proxy contain the entire understanding of the parties in respect of the subject matter hereof, and supersede all prior negotiations, agreements and understandings, both written and oral, between the parties hereto with respect to the subject matter hereof.

(m) <u>Severability</u>. If any term or other provision of this Support Agreement is invalid, illegal or incapable of being enforced by any rule of law, or public policy, all other conditions and provisions of this Support Agreement shall nevertheless remain in full force and effect. If the final judgment of a court of competent jurisdiction declares that any term or provision hereof is invalid or unenforceable, the parties hereto agree that the court making such determination shall have the power to limit the term or provision, to delete specific words or phrases, or to replace any invalid or unenforceable term or provision with a term or provision that is valid and enforceable and that comes closes to expressing the intention of the invalid or unenforceable term or provision, and this Support Agreement shall be enforceable as so modified. In the event such court does not exercise the power granted to it in the prior sentence, the parties hereto shall negotiate in good faith to modify this Support Agreement so as to effect the original intent of the parties as closely as possible in a mutually acceptable manner.

(n) <u>Interpretation</u>.

(i) Whenever the words "include," "includes" or "including" are used in this Agreement they shall be deemed to be followed by the words "without limitation." As used in this Agreement, the term "affiliate" shall have the meaning set forth in Rule 12b-2 promulgated under the Exchange Act.

(ii) The article and section headings contained in this Agreement are solely for the purpose of reference, are not part of the agreement of the parties hereto and shall not in any way affect the meaning or interpretation of this Agreement.

(o) <u>Attorneys' Fees</u>. If any legal action or other legal proceeding relating to this Support Agreement or the enforcement of any provision of this Support Agreement is brought against Securityholder, the prevailing party shall be entitled to recover reasonable attorneys' fees, costs and disbursements (in addition to any other relief to which the prevailing party may be entitled).

(p) <u>Expenses</u>. All costs and expenses incurred in connection with this Support Agreement, the Proxy and the transactions contemplated hereby and thereby shall be paid by the party incurring such costs and expenses.

(q) <u>Further Assurances</u>. From time to time, at any other party's request and without further consideration, Securityholder shall (at Securityholder's sole expense) execute and deliver any additional documents and take any further lawful action as may be necessary or desirable, in the reasonable opinion of DMK, to consummate and make effective, in the most expeditious manner practicable, the transactions contemplated by this Support Agreement and to carry out the intent of this Support Agreement.

(r) <u>Counterparts</u>. This Support Agreement may be executed in several counterparts, each of which shall be an original, but all of which together shall constitute one and the same agreement.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, the parties have caused this Support Agreement to be duly executed on the day and year first above written.

ADAMIS PHARMACEUTICALS CORPORATION

By: <u>/s/ David J. Marguglio</u> Name:David J. Marguglio Title: Chief Executive Officer

AARDVARK MERGER SUB, INC.

By: /s/ David J. Marguglio

Name: David J. Marguglio Title: Chief Executive Officer

[Signature Page to Support Agreement]

DMK PHARMACEUTICALS CORPORATION

By: /s/ Ebrahim Versi Name: Eboo Versi, M.D., Ph.D. Title: Chief Executive Officer

SECURITYHOLDER

VERSI GROUP, LLC

By: /s/ Ebrahim Versi Name:Eboo Versi, M.D., Ph.D. Title: Manager

[Signature Page to Support Agreement]

EXHIBIT A

SHARES BENEFICIALLY OWNED

| Securityholder | Number of Shares of Common Stock |
|------------------|----------------------------------|
| Versi Group, LLC | 37,949 |

Options, Convertible Notes or Other Securities

Stock Options

| Option Holder | Number of Shares of Common Stock Underlying Stock Options |
|----------------------|--|
| Option noticer | Underlying Stock Options |
| Ebrahim "Eboo" Versi | 2,000 |

Convertible Notes

- 1. Convertible Promissory Note between DMK Pharmaceuticals, Inc. and Ebrahim Versi (Holder), dated January 1, 2020 with a principal amount of \$132,000.00, to be cancelled at closing.
- 2. Convertible Promissory Note between DMK Pharmaceuticals, Inc. and Ebrahim Versi (Holder), dated January 1, 2021 with a principal amount of \$680,000.00, to be cancelled at closing.
- 3. Convertible Promissory Note between DMK Pharmaceuticals, Inc. and Ebrahim Versi (Holder), dated July 1, 2021 with a principal amount of \$384,000.00, to be cancelled at closing.
- 4. Convertible Promissory Note between DMK Pharmaceuticals, Inc. and Ebrahim Versi (Holder), dated October 1, 2022 with a principal amount of \$615,000.00, to be cancelled at closing.
- 5. Convertible Promissory Note between DMK Pharmaceuticals, Inc. and Ebrahim Versi (Holder), dated December 31, 2021, with a principal amount of \$480,000.00, to be cancelled at closing.
- 6. Convertible Promissory Note between DMK Pharmaceuticals, Inc. and Eboo Versi (Holder), dated October 1, 2016 with a principal amount of \$20,000.00.
- 7. Convertible Promissory Note between DMK Pharmaceuticals, Inc. and Ebrahim Versi (Holder), dated March 1, 2018 with a principal amount of \$20,000.00.
- 8. Convertible Promissory Note between DMK Pharmaceuticals, Inc. and Ebrahim Versi (Holder), dated January 1, 2020 with a principal amount of \$6,796.25.
- 9. Convertible Promissory Note between DMK Pharmaceuticals, Inc. and Ebrahim Versi (Holder), dated October 1, 2020 with a principal amount of \$10,000.00.
- 10. Convertible Promissory Note between DMK Pharmaceuticals, Inc. and Ebrahim Versi (Holder), dated January 1, 2021 with a principal amount of \$1,264.12.
- 11. Convertible Promissory Note between Dina Pharma, Inc. and Ebrahim Versi (Holder), dated December 31, 2021 with a principal amount of \$12,279.40, assigned to DMK.
- 12. Convertible Promissory Note between Dina Pharma, Inc. and Ebrahim Versi (Holder), dated October 2, 2022 with a principal amount of \$9,317.76, assigned to DMK.

EXHIBIT B

IRREVOCABLE PROXY

The undersigned stockholder ("**Securityholder**") of DMK Pharmaceuticals Corporation, a New Jersey corporation ("**DMK**"), hereby irrevocably (to the fullest extent permitted by law) appoints David J. Marguglio and David C. Benedicto of Adamis Pharmaceuticals Corporation, a Delaware corporation ("**APC**"), and APC, and each of them, as the sole and exclusive attorneys and proxies of the undersigned, with full power of substitution and resubstitution, to vote and exercise all voting and related rights (to the full extent that the undersigned is entitled to do so) with respect to all of the shares of capital stock of DMK that now are or hereafter may be beneficially owned by the undersigned, and any and all other shares or securities of DMK issued or issuable in respect thereof on or after the date hereof (collectively, the "**Shares**"), in accordance with the terms of this Proxy until the Expiration Date (as defined below). Upon the undersigned agrees not to grant any subsequent proxies with respect to the Shares that are inconsistent with this Proxy until after the Expiration Date (as defined in that certain Support Agreement of even date herewith by and among DMK, APC, and the undersigned stockholder, among others (the "**Support Agreement**")).

This Proxy is irrevocable (to the fullest extent permitted by law), is coupled with an interest and is granted pursuant to the Support Agreement, and is granted in consideration of DMK and APC entering into that certain Agreement and Plan of Reorganization date as of February 24, 2023 (the "**Merger Agreement**"), by and among DMK, APC and Aardvark Merger Sub, Inc., a wholly-owned subsidiary of APC ("**Merger Sub**"). The Merger Agreement provides for the acquisition by APC of DMK by means of DMK with and into Merger Sub, with Merger Sub as the surviving corporation (the "**Merger**").

The attorneys and proxies named above, and each of them, are hereby authorized and empowered by the undersigned, at any time prior to the Expiration Date, to act as the undersigned's attorney and proxy to vote the Shares, and to exercise all voting, consent and similar rights of the undersigned with respect to the Shares (including, without limitation, the power to execute and deliver written consents) at every annual, special or adjourned meeting of stockholders of DMK and in every written consent in lieu of such meeting (i) in favor of approval of the Merger and the adoption and approval of the Merger Agreement (and any related plan of merger), and in favor of each of the other actions contemplated by the Merger; (iii) against approval of any proposal made in opposition to, or in competition or inconsistent with, the consummation of the Merger or the transactions contemplated by the Merger Agreement (including, without limitation, any action or agreement that would result in a breach of any representation, warranty, covenant or obligation of DMK in the Merger Agreement); and (iv) in favor of waiving any notice that may have been or may be required relating to any reorganization of DMK or any subsidiary of DMK with or into any other person.

1

The attorneys and proxies named above may not exercise this Proxy on any other matter. The undersigned stockholder may vote the Shares on all other matters.

Any obligation of the undersigned hereunder shall be binding upon the heirs, estate, executors, personal representatives, successors and assigns of the undersigned (including any transferee of the Shares).

If any term or other provision of this Proxy is invalid, illegal or incapable of being enforced by any rule of law, or public policy, all other conditions and provisions of this Proxy shall nevertheless remain in full force and effect. If the final judgment of a court of competent jurisdiction declares that any term or provision hereof is invalid or unenforceable, the parties hereto agree that the court making such determination shall have the power to limit the term or provision, to delete specific words or phrases, or to replace any invalid or unenforceable term or provision with a term or provision that is valid and enforceable and that comes closes to expressing the intention of the invalid or unenforceable term or provision, and this Proxy shall be enforceable as so modified. In the event such court does not exercise the power granted to it in the prior sentence, the parties hereto shall negotiate in good faith to modify this Proxy so as to effect the original intent of the parties as closely as possible in a mutually acceptable manner.

[Signature Page to Follow]

This Proxy is irrevocable (to the fullest extent permitted by law). This Proxy shall terminate, and be of no further force and effect, automatically upon the Expiration Date.

Dated: February 24, 2023

VERSI GROUP, LLC

By: /s/ Ebrahim Versi Name: Eboo Versi, M.D., Ph.D. Title: Manager

[Signature Page to Irrevocable Proxy]

Adamis Pharmaceuticals and DMK Pharmaceuticals Announce Agreement and Plan of Merger

- Combined company will have commercial products and a library of development candidates, two of which are clinical stage
- Will be under the leadership of DMK CEO, Dr. Ebrahim (Eboo) Versi MD, PhD
- Lead development program will be a clinical stage therapeutic under development for the treatment of opioid use disorder

SAN DIEGO and SOMERVILLE, NJ February 27, 2023 – <u>Adamis Pharmaceuticals Corporation</u> (NASDAQ: ADMP), a specialty biopharmaceutical company focused on developing and commercializing products in various therapeutic areas, including opioid overdose, allergy, respiratory and inflammatory disease, and DMK Pharmaceuticals, Corp., a private, clinical-stage biotechnology company at the forefront of endorphin-inspired drug design focused on developing novel treatments for opioid use disorder (OUD) and other neuro-based diseases, today announced that the companies have entered into an Agreement and Plan of Merger and Reorganization (the "Agreement"). Pursuant to the Agreement, Adamis will acquire DMK, including its library of approximately 750 small molecule neuropeptide analogues and on-going government funding for its development programs.

"Last fall, we initiated a process to explore strategic alternatives for the company with the goal of maximizing stockholder value," stated David J. Marguglio, CEO of Adamis. "After engaging in a thorough process of exploring potential alternatives and transactions, we believe a merger with DMK is the best path forward for Adamis and the strategy that has the potential to deliver significant value to Adamis' shareholders. I believe by combining Adamis' commercial products and development infrastructure with DMK's clinical-stage programs and library of small molecules, under Dr. Versi's leadership, the new company will have the potential to develop multiple groundbreaking treatments and ultimately grow shareholder value."

At the close of the merger, Eboo Versi, the current CEO of DMK, will assume the role of CEO and chairman of the combined company. Dr. Versi explained, "There are substantial synergies between Adamis and DMK. The combined company will have both the expertise and infrastructure to further the development of DMK's potentially life-changing products to address large unmet medical needs. I believe that each of our clinical-stage product candidates has blockbuster potential, and we only need one to succeed to significantly increase shareholder value. I believe the combined companies present a risk diversified portfolio which is especially important in a time of market uncertainty."

Please find a short video discussion with both CEOs on Adamis' website.

Transaction Details

On February 24, 2023, Adamis entered into the Agreement with DMK and Aardvark Merger Sub ("Merger Sub"), a newly created whollyowned subsidiary of Adamis, pursuant to which DMK will merge with and into Merger Sub (the "Merger"), with Merger Sub as the surviving corporation of the Merger and a wholly owned subsidiary of Adamis.

Subject to approval by the Adamis stockholders of proposals relating to the transaction and the satisfaction of other closing conditions, in connection with and before the effective time of the Merger (the "Effective Time"), a reverse stock split of Adamis Common Stock will be consummated, pursuant to which a number of outstanding shares of Adamis Common Stock (determined by the Reverse Stock Split Ratio) will be converted into one share of Adamis Common Stock at a ratio to be determined by the Adamis board of directors.

Please see Adamis Report on Form 8-K which will be filed with the Securities and Exchange Commission for additional detail on the proposed transaction.

About DMK Pharmaceuticals

DMK Pharmaceuticals, Corp. is a privately held, clinical stage neuro-biotechnology company focused on developing novel therapies for opioid use disorder (OUD) and other important neuro-based conditions where patients are currently underserved. The company's technology is at the forefront of endorphin-inspired drug design. DMK is developing mono, bi- and tri-functional small molecules that simultaneously modulate critical networks in the nervous system with the goal of creating treatments that are efficacious, safe, and tolerable. DMK has a library of high value, first-in-class compounds and a differentiated pipeline that could address several unmet medical needs by taking the novel approach to integrate with the body's own efforts to regain balance of disrupted physiology. By designing small molecule analogs of neuropeptides, one or multiple receptors can be targeted by a single molecule to support a transition back to a balanced neurophysiological state.

Since the company's inception, DMK's development programs have been largely financed by non-dilutive funding from the government. DMK's lead clinical stage product candidate, DPI-125, is being studied as a potential novel treatment for OUD. The company also plans to develop the compound for the treatment of moderate to severe pain, where it could potentially offer a superior safety profile with lower addiction risk than currently marketed opioids (pain killers) and hence help prevent opioid addiction. DMK's other product candidates include DPI-221 being developed for treating bladder control problems and DPI-289 being developed for treating severe end stage Parkinson's disease. For additional information about DMK Pharmaceuticals, please visit the company website

About Dr. Versi

Eboo Versi received a BA, MA and DPhil (Ph.D) from Oxford University before obtaining his MB BChir (MD) degree from Cambridge University in the United Kingdom. Following medical school, Dr. Versi completed a residency and fellowship at Kings College Hospital and served as a Senior Registrar at the Royal London Hospital. He was then a Senior Lecturer and Consultant (Attending) at St. Thomas' & Guys Hospitals before moving to the U.S. to accept a senior academic position at Harvard Medical School. There, he set up the first urogynecology program at the Brigham & Women's Hospital and served as Chief of Urogynecology. Since Harvard, Dr. Versi has spent the last 20+ years in the pharmaceutical and medical device industry, holding positions at large companies such as Pfizer, Lilly and Astellas, as well as smaller companies including Odyssey, Plethora, Auxilium and Mt. Cook. During his career in the industry, Dr. Versi has been the inventor of several patents and the recipient of several NIH grants, and has helped develop drugs and devices for a variety of indications. Dr. Versi is an internationally recognized opinion leader with more than 100 scientific publications.

About Adamis Pharmaceuticals

Adamis Pharmaceuticals Corporation is a specialty biopharmaceutical company focused on developing and commercializing products in various therapeutic areas, including opioid overdose, allergy, respiratory and inflammatory disease. Company products approved by the FDA include ZIMHI[®] (naloxone) Injection for the treatment of opioid overdose and SYMJEPI[®] (epinephrine) Injection for use in the emergency treatment of acute allergic reactions, including anaphylaxis. For additional information about Adamis Pharmaceuticals, please visit our website and follow us on Twitter and LinkedIn.

Forward Looking Statements

This press release contains forward-looking statements within the meaning of the Private Securities Litigation Reform Act of 1995. Forwardlooking statements are identified by terminology such as "may," "should," "expects," "plans," "anticipates," "could," "intends," "target," "projects," "contemplates," "believes," "estimates," "predicts," "potential" or "continue" or the negative of these terms or other similar words. Such forward-looking statements include those that express plans, anticipation, intent, contingencies, goals, targets or future development and/or otherwise are not statements of historical fact. These statements relate to future events or future results of operations, including, but not limited to statements concerning the following matters: (i) risks associated with Adamis' and DMK's ability to obtain the stockholder approvals required to consummate the proposed Merger and the timing of the closing of the proposed Merger; risks that one or more conditions to closing of the Merger may not be satisfied within the expected timeframe or at all or that the closing of the proposed Merger will not occur; (ii) the outcome of any current legal proceedings or future legal proceedings that may be instituted against the parties or others. including proceedings related to the Merger Agreement; (iii) the occurrence of any event, change or other circumstance or condition that could give rise to the termination of the Merger Agreement; (iv) unanticipated difficulties or expenditures relating to the proposed Merger; (v) potential difficulties in employee retention as a result of the announcement and pendency of the proposed Merger; (vi) whether the combined business of DMK and Adamis will be successful; (vii) whether any DMK product candidates will be successfully developed or commercialized; (viii) the Company's review and evaluation of potential strategic alternatives and their impact on stockholder value; (ix) the Company's ability to raise capital to continue as a going concern; and (x) those risks detailed in Adamis' most recent Annual Report on Form 10-K and subsequent reports filed with the Securities and Exchange Commission ("SEC"), as well as other documents that may be filed by Adamis from time to time with the SEC. These statements are only predictions and involve known and unknown risks, uncertainties, and other factors, which may cause Adamis' actual results to be materially different from the results anticipated by such forward-looking statements. Accordingly, you should not rely upon forward-looking statements as predictions of future events. Neither Adamis nor DMK can assure you that the events and circumstances reflected in the forward-looking statements will be achieved or occur, and actual results could differ materially from those projected in the forward-looking statements. Factors that could cause actual results to differ materially from management's current expectations include those risks and uncertainties relating to: our ability to raise capital: the results of our strategic review process; the risk of not obtaining stockholder approval for the proposals required to consummate the Merger; risks associated with development of DMK's drug product candidates; our cash flow, cash burn, expenses, obligations and liabilities; the outcomes of any litigation, regulatory proceedings, inquiries or investigations that we are or may become subject to; and other important factors discussed in the Company's filings with the Securities and Exchange Commission ("SEC"). If we do not obtain required additional equity or debt funding, our cash resources will be depleted and we could be required to materially reduce or suspend operations, which would likely have a material adverse effect on our business, stock price and our relationships with third parties with whom we have business relationships, at least until additional funding is obtained. If we do not have sufficient funds to continue operations or satisfy out liabilities, we could be required to seek bankruptcy protection or other alternatives to attempt to resolve our obligations and liabilities that could result in our stockholders losing some or all of their investment in us. You should not place undue reliance on any forward-looking statements. Further, any forward-looking statement speaks only as of the date on which it is made, and except as may be required by applicable law, we undertake no obligation to update or release publicly the results of any revisions to these forward-looking statements or to reflect events or circumstances arising after the date of this press release. Certain of these risks and additional risks, uncertainties, and other factors are described in greater detail in Adamis' filings from time to time with the SEC, including its annual report on Form 10-K for the year ended December 31, 2021, and subsequent filings with the SEC, which Adamis strongly urges you to read and consider, all of which are available free of charge on the SEC's website at http://www.sec.gov.

Additional Information about the Merger and Where to Find It

Adamis intends to file a proxy statement in connection with the proposed transaction. Investors and stockholders are urged to read this filing when it becomes available because it will contain important information about the transaction. This press release does not constitute an offer of any securities for sale or the solicitation of any proxy. BEFORE MAKING ANY VOTING OR INVESTMENT DECISION, ADAMIS' STOCKHOLDERS ARE URGED TO READ THE PROXY STATEMENT REGARDING THE PROPOSED TRANSACTION CAREFULLY AND IN ITS ENTIRETY WHEN IT BECOMES AVAILABLE BECAUSE IT WILL CONTAIN IMPORTANT INFORMATION ABOUT THE PROPOSED TRANSACTION AND RELATED MATTERS. Investors and stockholders may obtain free copies of the proxy statement and other relevant documents (when they become available) and other documents filed with the Securities and Exchange Commission at the Securities and Exchange Commission by Adamis by contacting David C. Benedicto, Adamis' chief financial officer, at (858) 997-2400.

Participants in the Solicitation

Adamis and DMK, and their respective directors and executive officers, may be deemed to be participants in the solicitation of proxies from the companies' stockholders in connection with the proposed transaction. Information regarding the interests of directors and executive officers in the transaction will be included in the proxy statement to be filed by Adamis. Investors and security holders are urged to read the Company's proxy statement and the other relevant materials when they become available before making any voting or investment decision with respect to the proposed transaction. Additional information regarding directors and executive officers of Adamis is also included in the Company's annual report on Form 10-K for the year ended December 31, 2021, and, when it becomes available, its annual report on Form 10-K for the year ended December 31, 2021, and Exchange Commission, which is available as described above.

No Offer or Solicitation

This press release is not intended to and shall not constitute an offer to sell or the solicitation of an offer to buy any securities or the solicitation of any vote in any jurisdiction pursuant to the proposed transaction or otherwise, nor shall there be any sale of securities in any jurisdiction in contravention of applicable law.

Contacts

Adamis Investor Relations Robert Uhl Managing Director ICR Westwicke 619.228.5886 robert.uhl@westwicke.com